THE ROLE OF THE COUNTY EMERGENCY MANAGER IN DISASTER MITIGATION

A Dissertation
Submitted to the Graduate Faculty
of the
North Dakota State University
of Agriculture and Applied Science

By

Amanda Miller Savitt

In Partial Fulfillment of the Requirements for the Degree of DOCTOR OF PHILOSOPHY

> Major Department: Emergency Management

> > November 2019

Fargo, North Dakota

North Dakota State University Graduate School

Title

The Role of the Count	ty Emergency Manager in Disaster Mitigation	
By		
	A 1.6 %	
	Amanda Savitt	
The Supervisory Committee cert	rifies that this <i>disquisition</i> complies with North Dakota State	
University's regulations and meets the accepted standards for the degree of		
DOCTOR OF PHILOSOPHY		
SUPERVISORY COMMITTEE:		
Dr. Sarah Kirkpatrick		
Chair		
Dr. Carol Cwiak		
Dr. Daniel Klenow		
Dr. Nicholas Bauroth		
Approved:		
8/10/2020	Dr. Jessica Jensen	
Date	Department Chair	

ABSTRACT

Scholarship on disasters in the United States would suggest that emergency managers should play a role in hazard mitigation. Yet, little empirical research has investigated precisely what role or roles emergency managers actually do play during this phase. This study explored the role of county-level emergency managers in hazard mitigation and the factors that might influence those roles.

Data for this study was collected through 42 in-depth, telephone interviews with county-level emergency managers in FEMA Regions III, V, and X, which includes the Mid-Atlantic, Midwest, and Pacific Northwest regions of the United States. Grounded theory was utilized in order to organize and analyze the data.

The data suggests that emergency managers play several roles within mitigation: a generic role, a support role, an administrative role, a promoter role, a public educator role, and a planning role. These roles are explained by a number of factors, including conceptual confusion, response and preparedness orientation, financial resource factors, planning factors, additional resource factors, competition between mitigation and development, resistance to mitigation, and engagement in mitigation. It is also important to note that emergency managers spend only a small amount of their time in mitigation.

The results of this study suggest that there is a gap between the theorized role and the actual role that emergency managers play within mitigation. Closing this gap will likely require additional resources for mitigation and county-level emergency management, as well as greater consistency in defining mitigation through policy and education.

ACKNOWLEDGEMENTS

So many people helped make this project possible, and I am deeply grateful to each and every one of you. Thank you first to all the emergency managers who agreed to be interviewed for this project – none of this would have been possible without you and the valuable information and insights you provided.

I am also deeply grateful to my dissertation advisor, Dr. Sarah Kirkpatrick. Thank you for believing in me and in this study, and of course for your guidance, graciousness and support throughout this process.

I would also like to thank my dissertation committee members, Dr. Carol Cwiak, Dr. Daniel Klenow, and Dr. Nicholas Bauroth. I sincerely appreciate the time you have taken to serve on my committee, as well as the wisdom and helpfulness of your recommendations for improvement.

I also need to thank everyone who has listened to me opine about mitigation for the past year, especially Samantha, without whom I sincerely doubt I would have completed this project. Mom and Dad, thank you for helping me not lose perspective when I was frustrated, and for the love and support you give me always. Stephen, thank you for sticking it out with me even though we started dating in the middle of my data collection, and for being my rock even while I was not much fun. Hannah and Niko, thank you for always believing in me even when I didn't believe in myself. My Killer Bees (and Joe), thank you all for the readily-available emotional support, as well as for the opportunity to punch things a few times a week.

Finally, thank you to all the baristas at the Starbucks on 52nd Ave, trying to imagine writing a dissertation without caffeine boggles the mind. I was going to dedicate this dissertation to you but I lost a bet and had to dedicate it to my parents' cat instead.

DEDICATION

This dissertation is dedicated to Jack Bauer, a very good boy.

TABLE OF CONTENTS

ABSTRACT	iii
ACKNOWLEDGEMENTS	iv
DEDICATION	V
LIST OF TABLES	ix
LIST OF ABBREVIATIONS	x
CHAPTER ONE: INTRODUCTION	1
Background	1
Significance	7
Conclusion	8
CHAPTER TWO: LITERATURE REVIEW	9
Mitigation Tasks and Activities	10
Factors that Influence Community Mitigation	13
Local Government and Leadership	15
Community	
External	24
Emergency Management Context	25
The Idealized Emergency Manager	26
Conclusion	29
CHAPTER THREE: RESEARCH METHODS	30
Methodological Approach	30
Population and Sampling	32
Data Collection	36
Data Analysis	38
Limitations	40

Conclusion	4
CHAPTER FOUR: FINDINGS	42
Allocation of Effort	42
Disaster Mitigation Role	43
Generic Terms to Describe Role	43
Support Role	45
Administrative Role	46
Promoter Role	49
Public Educator Role	50
Planning Role	52
Emergency Manager Characteristics and Role	55
Explanatory Factors	56
Conceptual Confusion	56
Response and Preparedness Orientation	59
Financial Resource Factors	60
Funding Requirements	63
Planning Factors	65
Additional Resource Factors	67
Competition Between Mitigation and Development	68
Resistance to Mitigation	69
Engagement in Mitigation	73
Conclusion	75
CHAPTER FIVE: DISCUSSION	77
Emergency Manager Role as Theorized and as Practiced	77
Findings and Analysis from a Similar Study	86

Understanding the Gap Between Theory and Practice	88
Adequacy of Mitigation Efforts	89
Why Do Theorized and Actual Emergency Managers Differ?	90
Why Are Emergency Managers Doing Too Little Mitigation?	92
Implications for Communities of Policy and Practice	95
Implications for Policy	96
Implications for Education.	100
Conclusion	101
CHAPTER SIX: CONCLUSION	102
Significance of Study for Emergency Management	102
Recommendations for Future Research	104
REFERENCES	106
APPENDIX A: EXCERPTS OF FEDERAL POLICY RELATED TO MITIGATION	124
APPENDIX B: INTERIVEW INVITATION LETTER	126
APPENDIX C: INFORMATION SHEET	127
APPENDIX D: INTERVIEW GUIDE	129
APPENDIX E: LIST OF POTENTIAL PROBING QUESTIONS	130
APPENDIX F: PARTICIPANT COUNTIES AND SELECT DEMOGRAPHIC INFORMATION	131
APPENDIX G: INSTITUTIONAL REVIEW BOARD APPROVAL	133
APPENDIX H: PARTICIPANTS' DEFINITIONS OF MITIGATION	134

LIST OF TABLES

<u>Table</u>		<u>Page</u>
1.	Categories and Examples of Mitigation Activities.	11
2.	Idealized Activities of Emergency Managers in Mitigation	28
3.	Summary of Participants Interviewed by Location	35
4.	Mitigation Roles Described by Participants	44
5.	Idealized Activities of Emergency Managers in Mitigation Compared to Activities of Study Participants in Mitigation	86

LIST OF ABBREVIATIONS

BLM	
CBRN	
CDBG-MIT	
CEMHS	Center for Emergency Management and Homeland Security
CRS	
DHS	Department of Homeland Security
DMA	
DRRA	
EMAP	Emergency Management Assessment Program
FEMA	Federal Emergency Management Agency
HUD	
IAEM	International Association of Emergency Managers
IPCC	
IRB	
NEMA	
NMF	

CHAPTER ONE: INTRODUCTION

For nearly four decades, Comprehensive Emergency Management has served as both a key organizing paradigm and recommended approach for emergency management research and practice in the United States (National Governor's Association, 1979). This approach employs four functional phases – response, preparedness, recovery, and mitigation – and underscores emergency management's involvement in activities across each phase (National Governor's Association, 1979). There is, however, some suggestion of an ongoing bias toward response and preparedness, with mitigation and recovery being comparatively neglected (Britton, 1999; Stehr, 2007). If the emergency management profession has indeed tasked itself with engaging in all phases of the comprehensive emergency management cycle (FEMA, 2011), it would seem to follow that emergency management practitioners would play an active role in implementing mitigation activities within their communities. Yet, little research has examined the extent to which emergency managers are involved in mitigation broadly, and implementation of mitigation activities specifically. This research will therefore seek to address the following question: What is the role of the county-level emergency manager in disaster mitigation?

Background

Hazard events may be destructive, disruptive, expensive, and pose a threat to lives, homes, and communities (Aherns & Rudolph, 2006; Bourque et al., 2007; Disaster Mitigation Act of 2000, 2000; Mileti, 1999; National Research Council, 1999; Zhang, Lindell & Prater, 2009). Hazard events can be broadly classified as emergencies, disasters and catastrophes (Auf der Heide, 1989; Quarantelli, 2006). Emergency management, despite having 'emergency' in its title, focuses primarily on disasters – namely, those hazard events for which jurisdictions' resources are overwhelmed, responding personnel and organizations must take on new tasks, the

structure of responding organizations changes or new organizations are formed, participants who are not typically involved in emergency management become mobilized, jurisdictional boundaries are crossed, and there is a need for coordination (Auf der Heide, 1989).

The way disasters have historically been managed in the United States, both in practice and in policy, has been highly reactive (McEntire, 2004; Rubin, 2012). Rather than focusing on proactive measures such as locating communities outside of hazardous areas or writing and enforcing building codes designed to protect people from disasters, U.S. society, as a whole, has instead tended to respond to disaster threats only once imminent or after they have occurred. Moreover, our exposure to disasters, as well as their costs to society, have only increased (Brouwer et al., 2011; Cutter & Emrich, 2005; IPCC, 2012; Munich Re, 2013); the number of hazard events has continued to rise year after year, with growing impacts on individuals and households, organizations and communities (Rubin, 2012). Exacerbating this issue, the disaster impacts we face are likely to grow even more across stakeholder groups due to climate change (O'Brien et al., 2006; Van Aalst, 2016), as hazards intensify in number and scope (IPCC, 2018), and more communities find themselves in the way of rising sea levels (Nicholls & Cazenave, 2010; Rahmstorf, 2007).

Instead of continuing to pursue reactionary responses to disaster threats, it is possible to seek out strategies that reduce our exposure to them. One of the principle ways in which we can reduce our disaster exposure is through mitigation (Highfield & Brody, 2017; Hill & Gaillard, 2013; McSweeney & Coomes, 2011). Mitigation refers to the phase of the emergency management cycle in which "sustained action [is taken] to reduce or eliminate risk to people and property from hazards and their effects" (FEMA, 2017, p. 1). Mitigation would ideally occur before a disaster has occurred (Godschalk, Brower & Beatley, 1989), but in many instances,

communities instead attempt to take advantage of "windows of opportunity" to implement mitigation in the wake of disasters (Asgary et al., 2007; Baker, 1977; Fois & Forino, 2014; Healy & Malhotra, 2009; Hill & Gaillard, 2013; Labossière & McGee, 2017; Le Masson, 2015; McSweeney & Coomes, 2011; Olson, Olson & Gawronski, 1998; Paul & Che, 2011; Solecki & Michaels, 1994; Tunstall, Johnson & Penning-Roswell, 2004).

Mitigation has been found to be a cost-effective strategy for reducing disaster losses, with researchers estimating that for every dollar spent on hazard mitigation, communities receive four dollars in future benefits (although this figure varies by hazard type) (FEMA, 2018; Rose et al., 2007). Research and policy alike also suggest that mitigation should, however, exist in a broader context of community development (Berke, Karetz & Wenger, 1993; DHS, 2016; Godschalk, 2003; Mileti, 1999; Paton, 2000; Pearce, 2003; Schneider, 2002). Mileti (1999) uses the term 'sustainable hazard mitigation' to link "wise management of natural resources with local economic resiliency, viewing hazard mitigation as an integral part of a much larger context" (p. 2). Other researchers have expanded on this idea, linking disaster mitigation to environmental conservation (e.g. Barbier, 2006; Costanza et al., 2008; Duxbury & Dickinson, 2007), sustainable livelihoods (e.g. Cannon, Twigg & Rowell, 2003; Pomeroy et al., 2006; Sanderson, 2000), and community economic growth (e.g. Berke, Karetz & Wenger, 1993), among others. For example, conserving wetlands may not only preserve biodiversity and natural beauty, but also reduce hurricane impacts to coastal communities (Berke, Song & Stevens, 2009; Godschalk, 2003; Prater & Lindell, 2000; Reddy, 2000). In spite of the benefits it confers, the trend of increasing disaster losses suggests that communities are not implementing enough mitigation.

Both literature and policy suggest that the local level bears the primary responsibility for emergency management functions, including mitigation (e.g., Cutter et al., 2013; Dynes,

Quarantelli & Kreps, 1972; National Mitigation Framework, 2016; Robert T. Stafford Act, 2000). It is axiomatic among disaster scholars, for example, that "all disasters are local" (e.g. Cutter et al., 2013; Dynes, Quarantelli & Kreps, 1972; Perry, 2003; Somers & Svara, 2009). This emphasis on local primacy is due to the fact that the local level is closest to the impacts of disasters (Somers & Svara, 2009; Waugh, 1994), is able to respond to disasters most quickly (Schneider, 1995; Waugh, 1996), and has the best understanding of local priorities, needs, and emergent needs (Schneider, 1995; Somers & Svara, 2009). From a mitigation standpoint, local primacy is reasonable, because local governments control many of the policies often associated with mitigation, including building codes, zoning decisions, and other land use policies (Burby, 2006; Col, 2007; Godschalk, 2003; Henstra, 2010; Reddy, 2000).

This notion of local primacy is echoed by legislation. Federal policy and legal frameworks suggest that emergency management activities broadly, and mitigation activities specifically, are a responsibility of local government. For example, the National Mitigation Framework (NMF), a nonbinding document produced by FEMA that "describes the benefits of being prepared by understanding risks and what actions can help address those risks" (FEMA, 2016, p. 1), identifies thirteen categories of mitigation roles and responsibilities that should be undertaken by local governments (National Mitigation Framework, 2016). The Disaster Mitigation Act (DMA) requires that local governments develop and submit mitigation plans in order to receive federal funding for mitigation (Disaster Mitigation Act, 2000). It further emphasizes the value of encouraging mitigation at the local level, including forming effective community partnerships, implementing hazard mitigation measures, and committing to long-term hazard mitigation (Disaster Mitigation Act, 2000). Although it is clear that actors at the

local level should be responsible for mitigation, it is not clear who within local jurisdictions should take on particular tasks during a mitigation process.

Although local government is responsible for mitigation, there is nothing that would suggest that mitigation is the sole responsibility of any singular person or agency within local government. Emergency management is a "distributed function" (Canton, 2007). That is, the responsibility for the broad array of tasks and activities associated with emergency management, including mitigation, is shared across departments and organizations throughout government, and society more broadly (Canton, 2007; Drabek & Hoetmer, 1991; McEntire, 2007). In the case of mitigation, this distributed function could include planners (e.g. Sipe & Vella, 2014; Solecki & Michaels, 1994; Stevens, Berke & Song, 2008; Wu & Lindell, 2004 e.g. Sipe & Vella, 2014; Solecki & Michaels, 1994; Stevens, Berke & Song, 2008; Wu & Lindell, 2004), engineers (e.g. Posner & Georgakakos, 2017), local elected government officials (e.g. Birkland, 1996; Birkland, 1998), members of the public (e.g. Evans-Cowley & Gough, 2008; Godschalk, Brody & Burby, 2003), other local government agencies, and potentially many others. For example, planners may be responsible for spearheading a mitigation planning process, while engineers may be responsible for recommending specific mitigation measures for a given hazard and community, and local elected officials are responsible for passing legislation related to zoning and land use.

It would also seem that emergency management professionals would have a role to play in mitigation. Emergency managers, defined as the professionals who "coordinat[e] and integrat[e] all activities necessary to build, sustain, and improve the capability to mitigate against, prepare for, respond to, and recovery from threatened or actual natural disasters, acts of terrorism, or other man-made disasters" (FEMA, 2007, p. 4), have tasked themselves with

engaging in all phases of the comprehensive emergency management cycle (FEMA, 2011). It would follow, therefore, that emergency managers would be engaged in mitigation activities.

Moreover, the suggestion that emergency managers should move beyond preparedness and response has been made in the literature. Although he was describing the evolution of the role of the emergency manager more broadly, Britton (1999) identified a number of areas that emergency managers should be responsible for related to mitigation. These include: assisting in the creation and management of community development and growth; assisting in the "process of enhancing the long-term equilibrium between human and natural environmental interactions" (p. 232); helping "ensure that appropriate emergency management mechanisms are in place, are operable, and are capable of responding to the overall risk environment" (p. 232); and linking emergency management with wider community management (pp. 232-233). This framing of the role of the emergency manager would strongly suggest that mitigation, especially sustainable hazard mitigation (e.g. Mileti, 1999), is within the scope of the local emergency manager's responsibilities.

Mitigation represents a critical strategy for managing the growing disaster threat faced by our communities (Schneider, 2002). Mitigation can reduce both the suffering caused by disasters, as well as the costs communities incur due to disasters (FEMA, 2018; Highfield & Brody, 2017; Rose et al., 2007). Although much of the work of mitigation is appropriately done by members of emergency management's distributed function, the comprehensive emergency management framework suggests that emergency management professionals have a role to play as well (Britton, 1999; FEMA, 2011; National Governor's Association, 1979; Stehr, 2007). Yet, it is not clear from the research what, if anything, emergency managers are doing related to disaster mitigation.

It is important to note here that local government may refer to county government, municipal government, or to any one of a variety of other jurisdictional structures, and that there is significant variation within local government structure (National Academy of Public Administration, 1980; National League of Cities, 2017; Silvestri & Nelson, 1999). Counties may be an especially important unit of local government to investigate, because they are located close to environmental problems or hazards (Benton, 2002; Benton & Menzel, 1993; Hoene, Baldassare & Shires, 2002; Park, 1996; Steel & Lovrich, 2000; Waugh, 1994), are likely to have greater resources than municipalities (Benton, 1996; Johnston, Pagano & Russo, 2000; Pagano & Johnston, 2000; Waugh, 1994), are structured in a way that may encourage inter- and intragovernmental cooperation (DeSantis & Renner, 1993; Menzel et al., 1992; Waugh, 1994), have a close relationship with state governments (Martin, 1993; Salant, 1989; Salant, 1993; Waugh, 1994), create opportunities for local-local cooperation (Cigler, 1993; Cigler, 1995; Massey & Smith, 1994; Sekwat, 1996; Waugh, 1994), and "serve as general purpose local governments representing local interests and having strong local identification" (Waugh, 1994, p. 253). As such, the focus of this research will be the county level. This does not preclude the importance of other levels of local government, but an investigation of municipal and special forms of local government has been left to future researchers.

Significance

Emergency management, both in scholarship and practice, has historically neglected mitigation, especially compared to preparedness and response (Britton, 1999; Stehr, 2007). In fact, very little research has been done using an emergency management lens at all; most empirical investigations of mitigation have been conducted through the lens of other disciplines and areas of practice, including planning and public policy (e.g. Burby et al., 1997; Berke,

Karetz & Wenger, 1993; May et al., 2013; Stevens, Berke & Song, 2008). Yet, mitigation not only has the potential to reduce or eliminate the effects of disasters on communities, but may also make communities better in a variety of ways, from economic and financial well-being, to improved environmental health, making it a crucial subject area for emergency management professionals and researchers alike.

Comprehensive emergency management includes mitigation, which suggests that emergency managers have some responsibility to help their communities mitigate. Yet it is not currently known what, if anything, local or county-level emergency managers are doing in regard to mitigation. By investigating the role of emergency managers in mitigation, this study will provide important foundational knowledge for scholars and practitioners. Understanding what (and how much) emergency managers do within mitigation will shed light on the value of the comprehensive emergency management framework to practice, and may suggest recommendations about how the role of the emergency manager should be modified to fit the realities of practice.

Conclusion

This chapter has introduced hazard mitigation and its importance to communities, and its place in comprehensive emergency management. It has also discussed ideal features of community hazard mitigation, discussed the influence of state and federal policies and programs on mitigation outcomes, and provided an overview of the history of the academic literature on mitigation. Chapter Two reviews the academic literature in more depth. Chapter Three discusses the research methods for this project. Chapter Four describes the findings of this research. Chapter Five discusses and contextualizes these findings, and considers their implications for education, theory, practice, and policy.

CHAPTER TWO: LITERATURE REVIEW

Chapter Two reviews the literature foundation for this study. No empirical research had specifically examined the county-level emergency manager's role in disaster mitigation at the onset of this project. As such, the purpose of this literature review was not to establish how the study fit into what is known already on the topic (Maxwell, 2005). Rather, the intent was to offer theoretical sensitivity (Glaser, 1978), allowing the researcher to initiate the project with an understanding of the nuances that may be associated with collected data. Specifically, in this chapter, the researcher synthesizes the work related to disaster mitigation that had been conducted across disciplines—including public administration, planning, sociology, and political science—to identify the contextual and background elements that provide the meaning and insight necessary for determining the relevance of data during collection and analysis (Strauss & Corbin, 1998). In addition, the researcher outlines how these contextual and background factors indicate a role that county-level emergency managers could be playing in disaster mitigation, which offered the source for comparison between the current role of county-level emergency managers and the role inferred by the relevant literature.

It must be noted here that Samuel and Siebeneck (2019) published an exploratory study while data collection and analysis for this project were ongoing that examined a similar research question, namely the role of local emergency managers in hazard mitigation planning and implementation. Given the express intent of the literature review and the grounded theory approach to this research, that study was not reviewed until data analysis was complete. While not considered here, that study and its related findings are integrated into the discussion in Chapter 5.

The first section of this chapter briefly discusses mitigation tasks and activities. The second section reviews the factors that have been found to influence mitigation, which include municipal, community, and external factors, and discusses the potential relationship between these factors and the role of the emergency manager. The third section considers the context in which local emergency managers operate. The fourth and final section describes an idealized emergency manager, based on the factors that influence mitigation.

Mitigation Tasks and Activities

Literature has identified a variety of mitigation tasks and activities than can be implemented at the community level. These activities may be classified into five main categories developed by Lindell et al. (2006): hazard source control, community protection works, land use practices, building construction practices, and building contents protection. According to Lindell et al. (2006),

Hazard source control involves intervention at the point of hazard generation..., community protection works attenuate disaster impact by altering the hazard transmission process... Land use practices limit hazard exposure by minimizing development in areas where the likelihood of hazard impact is high. By contrast, building construction practices limit physical vulnerability by building structures whose resistance to hazard impacts is high. Finally, building contents protection prevents furniture, equipment..., and other building contents from being damaged or destroyed (pg. 195).

Such a classification captures more mitigation activities than a simplistic structural versus nonstructural classification system (Lindell et al., 2006), although some activities, especially those related to education and awareness, could be added (e.g. Berke, Smith & Lyles, 2012; Kang, Peacock & Husein, 2010). Table 1 provides examples of activities that fall into each of these categories.

Table 1
Categories and Examples of Mitigation Activities

control reducing chemical quantities, low maintaining equipment (Lindell e include: slope terracing, slope dra	ing fuels, substituting nontoxic chemicals, ering operating temperatures and pressures,
maintaining equipment (Lindell einclude: slope terracing, slope dra	
include: slope terracing, slope dra	
1 0, 1	
	O ,
include: sanding, reforestation, so	ers, dredging reservoirs; replenishment, to il backfill; restoration: wetland, sand dune
`	ream channelization, dams, levees, and
	cluding reducing shear stress, increasing
), industrial storage failure controls (Lindell
	seawalls, natural or synthetic wind and
	ction systems, to include: avalanche bridges,
	ion trenches, channels, canals, spillways;
· · · · · · · · · · · · · · · · · · ·	m water management, dams, levees and
	valls, slope stabilization covers; detection
, ,	lites, CBRN systems, ground movement
	, weather stations, undersea and buoy
	rmation systems; treatment systems, to
include: water treatment systems,	
	ous materials decontamination systems;
Land use build/designate community shelte Acquisition of land and developm	
	ent rights, capital improvement programs, osts of private development on hazardous
1 1 1	ing, zoning, and subdivision regulations
	denial of services, density control, open
space preservation and/or protecti	
	protection from hydrological hazards,
	e hazard, structural protection from wind
	seismic hazard, structural protection from
· · · · · · · · · · · · · · · · · · ·	les (Lindell et al., 2006); safety standards,
building codes, resistant construct	
<u> </u>	rotection-in-place, and contents stabilization
	niture and equipment (e.g. furnaces, air
•	I other building contents from being
damaged or destroyed	omer durang contents from being

It should be noted that not every activity listed above is appropriate for every community or hazard. For example, Lindell et al. (2006) note that most of the hazard source control activities are appropriate primarily for technological hazards, which not every community must contend with. At the same time, all communities face risk, and, as discussed in the introduction, local

governments have a responsibility for mitigation. This would suggest that emergency managers should be engaging in at least some of these activities. Emergency managers themselves are not directly responsible for implementation of mitigation in most cases – for example, they would not be responsible for constructing a levee or enforcing building codes. However, they could, at minimum, work with those who are responsible for implementation to ensure that all parties have the resources and information they need.

In addition to the mitigation activities discussed above, planning is an additional process related to mitigation that has been studied extensively, and is incentivized by policy (e.g. Robert T. Stafford Act, 2000; Stehr, 2007). Planning has been found to improve mitigation outcomes (Berke, 1996), and research has identified several factors that improve planning outcomes, including the existence and quality of planning mandates (Berke, 1996; Brody et al., 2009), increased hazard salience (Deyle & Smith, 1998), and high levels of participation from community members (Brody, 2003). The literature has not addressed the extent to which emergency managers are engaged in mitigation planning processes, or should engaged in planning processes, but research would suggest that, at minimum, they could make mitigation planning more successful by bringing varied stakeholders together and educating the public about hazard risk.

Each category of mitigation activities, as well as mitigation planning, requires engagement and participation by many parties within a community, rather than the emergency manager alone. Many of these activities, however, fit within Britton's (1999) description of the role of the local emergency manager, and it is easy to see where others would represent emergency manager responsibilities. For example, emergency managers might take on the role of

negotiating the construction of a community protection work by working across government departments, or by helping to develop community-based coalitions.

Factors that Influence Community Mitigation

In addition to tasks and activities, the disaster literature has identified a number of factors that may facilitate or inhibit disaster mitigation at the local or community level. Some of the factors that have been found to influence community mitigation are entirely, or nearly entirely, outside the control of an emergency manager. These factors include the characteristics of policies and the legal framework or mandates that govern mitigation (e.g., Berke, 1996; Brody, 2009), and the socioeconomic or demographic characteristics of a community (e.g., Burby & Dalton, 1994; Landry & Li, 2011). With respect to policies, research has found that incentive programs, like the Community Rating System (CRS), have been somewhat successful at increasing mitigation (Rivera & Wamsler, 2014; Zahran et al., 2010), essentially acting as an additional financial resource for communities. The role of mandates has also been investigated extensively. Mandates have been found to influence mitigation in a number of ways, including: making the existence of plans more likely (Berke et al., 1996), making plans better (Berke et al., 1996), substituting for other community characteristics that make having a plan more likely (Berke et al., 1996), keeping post-disaster 'windows of opportunity' open longer (Solecki & Michaels, 1994), and increasing the amount of mitigation that is undertaken (Brody et al., 2009; Burby & Dalton, 1994; Reddy, 2000). Some research has also found, however, that even state-approved mandated plans do not meaningfully reduce hazard risk (Deyle, Chapin & Baker, 2008). Although emergency managers cannot control the content of federal or state mitigation policy, they can help ensure that their communities are in compliance with these policies, so that they are eligible for all associated benefits and incentives.

Regarding the relationship between a community's population profile and mitigation outcomes, findings are surprisingly scarce. Where researchers have considered population characteristics, there has been a great deal of variation in the specific factors they have evaluated. These have included communities' economic conditions (Landry & Li, 2011) and their size and density (Burby & Dalton, 1994; Landry & Li, 2011). Landry and Li (2011), for example, found that denser and wealthier communities were more likely to implement mitigation, and Burby and Dalton (1994) found that larger populations were associated with more mitigation. In the only study considered for this research that examined demographic factors, the researchers found that these factors did not influence mitigation implementation (Landry & Li, 2011). Although emergency managers may not be able to affect their community's geographic or demographic profile, or the regulations, policies and mandates under which they must operate, simply understanding these factors and the dynamics within their communities may make other mitigation-related activities they undertake more successful.

Although emergency managers may not be able to affect the above factors, there are other factors over which they do exercise some control. Among these factors, some are specific to the community. For example, characteristics of the government itself and local leaders have been found to affect mitigation outcomes (e.g., Birkland, 1996; Solecki & Michaels, 1994; Stevens, Berke & Song, 2008). Other factors are associated with the community. Among these, the support and engagement of the community (e.g. Evans-Cowley & Gough, 2008; Godschalk, Brody & Burby, 2003), and the community's hazard profile (e.g., McGee, 2011; Mockrin et al., 2015; Neuvel & Van den Brink, 2009) are particularly important. Finally, factors associated with actors external to the community, especially the availability of resources (e.g. Becker & Reusser, 2016; Brody et al., 2009; Fois & Forino, 2014; Hakaloba et al., 2016) have also been found to

affect mitigation. The research on these factors rarely offers findings about emergency managers specifically. However, the fact that emergency managers are generally understood to have a responsibility for coordination and integration (FEMA, 2007; Waugh & Streib, 2006) allows an appropriate role for emergency managers in mitigation to be induced from the literature. The following subsections include an in-depth discussion of these additional factors that may affect mitigation outcomes, and how emergency managers may influence these factors.

Local Government and Leadership

There are a number of leadership characteristics that researchers have found to influence mitigation, all of which emergency managers themselves may exhibit. One particularly important community leadership role is that of a policy entrepreneur (Birkland, 1996; Labossière & McGee, 2017; Olshanksy et al., 2008; Olson & Olson, 1993; Solecki & Michaels, 1994). The policy entrepreneur is an individual (or organization) committed to implementing mitigation, often after a disaster, and who has the resources to influence the mitigation process (Solecki & Michaels, 1994). The presence of a policy entrepreneur can give a mitigation process the direction, resources, and enthusiasm necessary for its success. Although in some cases, community planners have been identified as mitigation policy entrepreneurs (Birkland, 1996; Solecki & Michaels, 1994), emergency managers, as the profession tasked with engaging in mitigation as part of the comprehensive emergency management framework, should also be well-positioned to serve in this capacity. Yet, the extent to which they act as policy entrepreneurs remains unknown.

Researchers have also investigated the relationship between successful mitigation outcomes and the characteristics of the people responsible for driving mitigation processes. Like the research on policy entrepreneurs, most of this scholarship has investigated the characteristics

of planners, but it is likely that these characteristics would inform the success of emergency managers as well. Birkland (1996), for example, found that the presence and cohesiveness of professional policy communities can encourage mitigation. For example, in contrasting hurricane and earthquake policy, Birkland notes that there is a committed community of advocates for improved earthquake protections who can advocate for specific policies, while there is an absence of such a community for hurricanes, resulting in better earthquake policy and worse hurricane policy. Emergency managers could leverage this information by developing policy communities with others whose jurisdictions face similar hazards, and mobilizing these communities to advocate for better policy. This way, in the wake of a disaster, these policy communities of emergency managers would be able to propose policies and legislation that support community mitigation.

Stevens, Berke and Song (2008) investigated planners' role orientation (i.e. planners' beliefs about the scope of their professional role), and found that technical and mobilizer role orientations had a positive influence on mitigation outcomes, whereas advisor and participant role orientations had a negative influence on mitigation outcomes. That is, communities with planners whose beliefs reflected the idea that "planning is a 'rational' process in which the planner is a value-neutral technical adviser who provides technical information to decision makers and the public" (Stevens, Berke & Song, 2008, p. 741) had better mitigation outcomes than communities with planners who are "active participant[s] in decision making and ... advocate[s] for particular policies and ideas" (Stevens, Berke & Song, 2008, p. 741).

Extrapolating to emergency managers, this finding would suggest that emergency managers who are able to provide technical information and to mobilize their communities may encourage more mitigation than emergency managers who take a more direct role as an advocate. Although this is

seemingly at odds with findings about policy entrepreneurship, it can be resolved, as Stevens, Berke and Song (2008) suggest, by considering an advocate's values. That is, regardless of role orientation, if community leaders value and prioritize mitigation, the communities they serve may be more likely to incorporate mitigation measures.

Additional leadership characteristics scholars have found increase mitigation include political astuteness and awareness (Rubin & Barbee, 1985), discretion and decision-making authority over the mitigation process (Stevens, 2010), and independence from outside influences and resources (Norris-Raynbird, 2005). Political astuteness and awareness refers to the ability to make strategic decisions for the community that enhance long-term future security, requiring political and administrative leadership (Rubin & Barbee, 1985, p. 62), including clear communication with constituents and willingness to put the community's preferences into action (Rubin, Saperstein & Barbee, 1985, p. 42). For emergency managers, this may mean that engaging with their community and seeking buy-in for mitigation is particularly important. With respect to discretion, Stevens (2010) found that planners who had more authority over their mitigation planning process were able to achieve better mitigation outcomes, compared to those who shared control with developers, elected officials and others. Stevens (2010) notes, however, that commitment to mitigation should be combined with discretion to achieve the best outcomes. Emergency managers may also seek to increase the autonomy they have over the mitigation process to implement mitigation more effectively. Finally, independence here refers specifically to the ability of a community to control its own mitigation process, without outside assistance (Norris-Raynbird, 2005). Emergency managers may be able to influence independence by helping to develop community capacity, including working to educate and engage the public, and by seeking out financial and technical resources.

Although none of these findings are specific to emergency managers, by taking them together, it is possible to describe characteristics of an emergency manager who is able to successfully mitigate in their community. Such an emergency manager would make efforts to be engaged with their community on mitigation topics (Brody, Kang & Bernhardt, 2010; Norris-Raynbird, 2005; Rubin, Saperstein & Barbee, 1985; Stevens, Berke & Song, 2008). They would commit to mitigation (Birkland, 1996; Brody, Kang & Bernhardt, 2010; Labossière & McGee, 2017; Olshansky et al., 2008; Olson & Olson, 1993; Reddy, 2000; Stevens, Berke & Song, 2008), but take a hands-off, technical approach to their role (Stevens, Berke & Song, 2008), in which they would provide information to the community and allow others to set priorities for mitigation. They would also seek to develop their community's capacity to operate independently of outside forces (Norris-Raynbird, 2005).

In addition to leadership characteristics, characteristics of jurisdictional governments also influence the mitigation process. Brody, Kang and Bernhardt (2010) identify organizational capacity as a particularly important jurisdictional factor. Organizational capacity is defined as "the ability to anticipate [hazards], make informed decisions about mitigation, and implement effective policies" (Brody, Kang & Bernhardt, 2010, p. 171). Characteristics that explain organizational capacity include access to financial, staffing and technical resources, ability to communicate and share information, strong leadership, and a demonstrated commitment to hazard mitigation (Brody, Kang & Bernhardt, 2010). Emergency managers who work to develop staffing and other resources (Brody, Kang & Bernhardt, 2010; Manyena, 2013; Nelson, 2014; Olshansky et al., 2008), take advantage of education opportunities to become more technically adept (Brody, Kang & Bernhardt, 2010), and have become knowledgeable or experienced with

respect to their community's hazard risk and appropriate mitigation strategies (Brody, Kang & Bernardt, 2010; Rubin & Barbee, 1985; Stevens, 2010) may have better mitigation outcomes.

Related especially to information sharing, Raju and Van Niekerk (2013) address the negative influence of departmental siloing in government on mitigation outcomes. Government departments operate in silos when there is a lack of communication and integration across departments (Raju & Van Niekerk, 2013). Emergency managers can play an especially important role here, as some of their principle responsibilities are integration and coordination (Waugh, 1991; Waugh & Streib, 2006), which they can leverage by working across departments to increase communication and information sharing. A final dimension of jurisdictions that may influence mitigation outcomes is corruption (Le Masson, 2015). Emergency managers cannot control overall corruption within their jurisdictions, but they can operate ethically themselves and encourage ethical conduct across their community.

Community

Researchers agree that the extent of community engagement influences mitigation outcomes. Engagement here may refer to how much influence citizens have over the mitigation process (e.g. Evans-Cowley & Gough, 2008; Godschalk, Brody & Burby, 2003), the extent of citizen education efforts (e.g. Godschalk, Brody & Burby 2003; Hakaloba et al., 2016), the extent of community interest in engaging in the mitigation process (e.g. Godschalk, Brody & Burby, 2003; Hakaloba et al., 2016; Morris-Oswald & Sinclair, 2005), the extent to which the community is consulted during the mitigation process (e.g. Andersson-Sköld & Nyberg, 2016; Morris-Oswald & Sinclair, 2005), and how many citizens attended meetings related to mitigation (Morris-Oswald & Sinclair, 2005). This list would suggest that engagement includes both

autonomous community interest and involvement and efforts to engage the community on the part of emergency managers and other local officials.

In most cases, researchers agree that there is not extensive community engagement in mitigation (Evans-Cowley & Gough, 2008; Godschalk, Brody & Burby, 2003; Kita, 2017; Maly, 2017; Morris-Oswald & Sinclair, 2005; Stevens, Berke & Song, 2010). However, most researchers have found that community engagement positively affects mitigation outcomes (or that the absence of community engagement negatively affects mitigation outcomes) (Andersson-Sköld & Nyberg, 2016; Evans-Cowley & Gough, 2008; Hakaloba et al., 2016; Le Masson, 2015; Manyena, 2013; McNamara & Des Combes, 2015; Norris-Raynbird, 2005; Olshansky et al., 2008; Oulahen & Doberstein, 2012; Passerini, 2001; Sipe & Vella, 2014; Stevens, Berke & Song, 2010). Based on these findings, emergency managers should seek to increase engagement, including by working to consult stakeholders in the mitigation process, educating the community about mitigation projects, and allowing citizens to have a greater voice in decision-making.

Others, however, have noted that community engagement can generate conflict or unrealistic expectations (Asgary et al., 2007; Kennedy et al., 2008; MacAskill & Guthrie, 2016), which may worsen mitigation outcomes, and at least one study has found that engagement may result in worse outcomes (Stevens, Berke & Song, 2008). To some extent, this conflict can be resolved by considering how engagement takes place. For example, some authors have found that, where communities can effectively coordinate and develop coalitions, mitigation outcomes improve (Birkland, 1998; Goldstein, 2008; Labossière & McGee, 2017; Lane, 2000; Lu et al., 2017; Mannakkara, Wilkinson & Potangaroa, 2014; Meo, Ziebro & Patton, 2004; Nelson, 2014; Olson, Olson & Gawronski, 1998; Raju & van Niekerk, 2013; Sipe & Vella, 2014; Weischelgartner & Pigeon, 2015). This suggests that emergency managers who seek to

effectively manage community engagement could help the community to develop coalitions by working across stakeholder groups. In order to prevent unrealistic expectations from hampering outcomes, emergency managers could also focus on educating their communities and ensuring that they have the most accurate information for priority-setting and decision-making.

Media attention can also increase the implementation of mitigation (Olson & Olson, 1993), specifically by facilitating communication with the community (Sipe & Vella, 2014) and sharing data and information (Weischelgartner & Pigeon, 2016). Effective engagement of the media, which requires a consistent message be delivered across varying types of media, has also been found to increase mitigation (Sipe & Vella, 2014). It is also important for communities to have an effective and well-planned communication strategy (Baudoin & Wolde-Georgis, 2015; Fois & Forino, 2014), which includes how mitigation is framed to the community (Evans-Cowley & Gough, 2008), and the number of techniques used to transmit information (Stevens, Berke & Song, 2010). Evans-Cowley and Gough (2008), for example, found that citizens were more supportive of mitigation projects when they were framed as a reflection of community values than when they were framed as protection from hazards. Fois and Forino (2014) found that a well-developed communication strategy, including different messaging and channels for different audiences, was important to increase support both within and outside the community. It would therefore follow that successful mitigation outcomes would be associated with emergency managers who have developed strong communication strategies, made efforts to actively engage with the media, and used a large number of techniques to transmit mitigation information to the community.

Community priorities and values have also been found to have an influence on mitigation outcomes. Priorities and values are, essentially, what the community wants, either with respect to

disasters, or generally, and also include the community's attitudes toward government and governance (Diggins, Wright & Rossi, 1979; Greenberg et al., 2014; Harris, McGee & McFarlane, 2011; Maly, 2017; Morris-Oswald & Sinclair, 2005; Moser & Ekstrom 2010, p. 22029; Nelson, 2014; Powell, 2011; Townshend et al., 2015; Tunstall, Johnson & Penning-Roswell, 2004). For example, how much does the community value safety from disasters? Does it value this safety more than unrestricted economic development? Some researchers have attempted to monetize community values using willingness to pay for mitigation, and have found both cases in which people are willing to pay for mitigation (McNamara & Des Combes, 2015) and cases in which they are not (Greenberg et al., 2014).

Understanding community values and priorities can give emergency managers insight into whether to pursue mitigation projects, and which projects will be most likely to be supported and successfully implemented. Particularly effective emergency managers may also be able to influence priorities and values through education and well-developed communication strategies. As Rubin, Saperstein and Barbee (1985) found, using community values and priorities to determine mitigation strategies is a hallmark of political savvy managers, and leads to more successful outcomes.

In addition to these 'intangible' community characteristics, a community's physical characteristics, including its hazard profile, have also been found to influence mitigation outcomes. There are two primary elements of a community's hazard profile: the risk they face (or believe they face) (Burby & Dalton, 1994; Landry & Li, 2011) – associated with the location of the community relative to hazard risks (Burby & Dalton, 1994), quality of community infrastructure (Hunt & Watkiss, 2011, p. 38), and risk perception (Ford, Berrang-Ford & Paterson, 2011; Harris, McGee & McFarlane, 2011; Neuvel & van den Brink, 2009) – and their

hazard experience, both historical (Lu et al., 2017; Rubin & Barbee, 1985), and recent (McGee, 2011; Mockrin et al., 2015; Neuvel & Van den Brink, 2009). When communities have recently experienced a disaster, some scholars have investigated whether a "window of opportunity" has followed, in which policy makers and the public are more likely to be willing to undertake mitigation projects (Asgary et al., 2007; Baker, 1977; Fois & Forino, 2014; Healy & Malhotra, 2009; Hill & Gaillard, 2013; Labossière & McGee, 2017; Le Masson, 2015; McSweeney & Coomes, 2011; Olson, Olson & Gawronski, 1998; Paul & Che, 2011; Solecki & Michaels, 1994; Tunstall, Johnson & Penning-Roswell, 2004).

Findings about hazard experience and windows of opportunity are, however, not straightforward. Some researchers have found that windows of opportunity existed after disasters (Asgary et al., 2007; Baker, 1977; Fois & Forino, 2014; Healy & Malhotra, 2009; Hill & Gaillard, 2013; Labossière & McGee, 2017; Le Masson, 2015; McSweeney & Coomes, 2011; Olson, Olson & Gawronski, 1998; Paul & Che, 2011; Tunstall, Johnson & Penning-Roswell, 2004), some found that they only existed under specific conditions (Birkland, 1996; Birkland, 1998; Ozcevik et al., 2009; Solecki & Michaels, 1994; Tunstall, Johnson & Penning-Roswell, 2004; Wu & Lindell, 2004), some found they existed but were largely ineffective (Bolton & Orians, 1998), and others did not observe them at all (Allan & Bryant, 2011; Evans-Cowley & Gough, 2007). Similarly, some scholars have found that disaster experience increases mitigation implementation (Baker, 1977; Deyle & Smith, 1998; Evans-Cowley & Gough, 2008; Fois & Forino, 2014; Harris, McGee & McFarlane, 2011; Labossière & McGee, 2017; Landry & Li, 2011; McGee, 2011; Mockrin et al., 2015; Neuvel & van den Brink, 2009; Sipe & Vella, 2014), while others have found that it decreases mitigation implementation (Lin, Shaw & Ho, 2008; Mannakkara, Wilkinson & Potangaroa, 2014). Other researchers have simply found that disaster

experience tends to reinforce pre-existing ideas about mitigation (Morris-Oswald & Sinclair, 2005).

Even if windows of opportunity are rare or only modestly helpful, emergency managers should attempt to take advantage of the periods after disasters to encourage mitigation adoption. In communities that have experienced a disaster, those with emergency managers who are aware of the potential window of opportunity may have more successful outcomes, especially when combined with an effective strategy for engagement and pre-disaster planning initiatives. More generally, without implementing mitigation, emergency managers cannot affect the risk their community faces, but they can influence how the residents they work for understand and perceive their risk through risk assessments and education initiatives. Where residents have a more accurate understanding of the hazard risk they face, they may be more likely to support mitigation initiatives (Rubin, Saperstein & Barbee, 1985).

External

In terms of the existing research, the principal external category of factors over which emergency managers may nevertheless exercise some control is resource availability. For the most part, the findings about resource availability related to mitigation are straightforward. When projects are expensive, or when communities lack resources, less mitigation occurs. This is true for both financial (Baudoin & Wolde-Georgis, 2015; Becker & Reusser, 2016; Brody et al., 2009; Fois & Forino, 2014; Hakaloba et al., 2016; Hill & Gaillard, 2013; Labossière & McGee, 2017; Manyena, 2013; Nelson, 2014; Raju & van Niekerk, 2013; Sipe & Vella, 2014; Wiek et al., 2010; Yoon, Youngs & Abe, 2012) and non-financial resources, including technological (Meo, Ziebro & Patton, 2004; Posner & Georgakakos, 2017; Stevens, Berke & Song, 2010), human (Hunt & Watkiss, 2011), and information resources (Moser & Ekstrom, 2010). The

availability of external resources, including technical and financial resources (Asgary et al., 2007; Maly, 2017; Raju & van Niekerk, 2013) among others (Olshansky et al., 2008; Rubin & Barbee, 1985), has also been found to increase mitigation, although dependence on external resources – especially leadership resources – has been found to limit mitigation in at least one case (Norris-Raynbird. 2005).

Although emergency managers cannot control their community's resources, they can influence resource availability at least two ways: seeking out grants and other sources of funding, and ensuring that they have met requirements to receive federal resources. For example, the Robert T. Stafford Act requires that communities have mitigation plans in order to receive federal funding (Robert T. Stafford Act, 1988; Stehr, 2007). We would likely expect communities who have successfully undertaken mitigation to have emergency managers who actively work to increase resources available for mitigation. Emergency managers could also coordinate with neighboring jurisdictions to access technical or human resources that their own communities lack.

Emergency Management Context

As the above section explains, county-level emergency managers may be able to exert their influence over elements of the mitigation process, however it is possible that their role may be constrained by contextual factors. Some of these contextual factors include features of their county government, including whether they operate their own department or are part of another department (Labadie, 1984), how much statutory authority they have (Labadie, 1984), how informed and interested the public is in their role (Labadie, 1984; McEntire & Dawson, 2007), access to resources from the local government (Petak, 1985; Waugh, 1993), and how emergency management departments are perceived in terms of cost-benefit determinations (McEntire &

Dawson, 2007; Somers & Svara, 2009). There may be significant variance across these factors: for example, some county-level emergency managers may have a great deal of authority and autonomy whereas others do not; and some counties may have residents who are actively engaged in conversations about emergency management and mitigation whereas others may have residents who are unaware of or disinterested in emergency management issues.

With respect to mitigation, county-level emergency managers also operate within a larger federal and state policy context which may affect how and whether they may influence mitigation outcomes. One of these policies is the National Mitigation Framework (NMF), part of the National Preparedness System. The NMF does not legally require local governments to take any specific actions, but it does identify some of the mitigation activities they should take responsibility for. The NMF identifies a long list of activities as being roles or responsibilities of local governments, which can be found in Appendix A (National Mitigation Framework, 2016). In contrast, the Robert T. Stafford Act requires that municipalities have mitigation plans, although it does not require that they actually implement mitigation (Robert T. Stafford Act, 1988; Stehr, 2007). Mitigation planning is also mandated by a number of states (Berke et al., 1996; Burby & Dalton, 1994; Deyle, Chapin & Baker, 2008; Deyle & Smith, 1998; Solecki & Michaels, 1994). Although community and policy context and their importance for emergency managers have been considered and discussed in the disaster scholarship, it is not clear how these contextual factors influence mitigation outcomes.

The Idealized Emergency Manager

While the literature does not explicitly describe how county-level emergency managers are involved in disaster mitigation, it is possible to use the above factors to extrapolate a profile of the role an emergency manager could play in the successful implementation of mitigation in

the community. Such an emergency manager has sought to maximize their community's financial resources for mitigation through compliance with federal programs and by pursuing mitigation grants (e.g. Baudoin & Wolde-Georgis, 2015; Brody et al., 2009; Zahran et al., 2010). They have looked for opportunities to increase resources to improve organizational capacity (e.g. Brody, Kang & Bernhardt, 2010; Nelson, 2014; Olshanksy et al., 2008) and have taken advantage of educational opportunities to become more technically adept and knowledgeable (e.g. Brody, Kang & Bernhardt, 2008; Rubin & Barbee, 1985; Stevens, 2010). They may also have tried to make their governments more effective by working to reduce siloing by working across government departments (e.g. Brody, Kang & Bernhardt, 2010; Raju & Van Niekerk, 2013), and by behaving ethically (Le Masson, 2015). By developing greater organizational capacity, this emergency manager has also increased their community's independence and ability to run their own mitigation process without outside assistance (Norris-Raynbird, 2005).

When their community has experienced a disaster, they have recognized the potential for a window of opportunity for mitigation, and have tried to take advantage of it (e.g. Baker, 1977; Hill & Gaillard, 2013; Olson, Olson & Gawronski, 1998). They may act as a policy entrepreneur, actively advocating for mitigation for their community (e.g. Birkland, 1996; Olshansky et al., 2008; Solecki & Michaels, 1994). This commitment to mitigation could result in part from their membership in a robust and cohesive policy community that also advocates for mitigation (Birkland, 1996).

Even if they have not actively advocated for mitigation, however, the emergency manager can still facilitate its implementation. For example, they can take on a technical or mobilizing role, providing members of their community with the resources and opportunities to advocate for mitigation themselves (Stevens, Berke & Song, 2008). As a mobilizer, the

emergency manager would have centered community needs and preferences (Rubin, Sapertstein & Barbee, 1985), and worked to engage their community in mitigation decision-making (e.g. Evans-Cowley & Gough, 2008; Passerini, 2001; Stevens, Berke & Song, 2010). To further facilitate community engagement, the emergency manager has helped members of the community develop coalitions (e.g. Birkland, 1998; Lane, 2000; Weischelgartner & Pigeon, 2015), and created interest in the process through an effective media and communications strategy (e.g. Baudoin & Wolde-Georgis, 2015; Sipe & Vella, 2014; Stevens, Berke & Song, 2010). The activities of the idealized emergency manager in mitigation are summarized below in Table 2.

Table 2
Idealized Activities of Emergency Managers in Mitigation

Idealized Activities

Increase availability of financial resources

- Maintain compliance with federal, state and local requirements
- Apply for and manage mitigation grants
- Increase organizational capacity for mitigation and emergency management generally Engage in mitigation advocacy
 - Act as a mitigation policy entrepreneur
 - Mobilize other stakeholders
 - Develop a mitigation policy community
 - Take advantage of 'windows of opportunity'

Work with their community

- Center community needs and preferences in decision making
- Engage community in mitigation decision making
- Develop community coalitions
- Develop effective media and communications strategies

Improve professionalization

- Operate ethically
- Seek educational opportunities in mitigation
- Develop knowledge or experience in mitigation
- De-silo local government

The emergency manager can take actions to increase the likelihood that their community will successfully implement mitigation, but it is important to recognize the importance of

context. The most successful emergency managers likely operate in environments where they have a great deal of statutory authority and autonomy (Labadie, 1985) and in communities that support emergency management initiatives (e.g. Labadie, 1984; Petak, 1985; Somers & Svara, 2009).

Conclusion

This chapter has explored the factors that have been found to influence community-level mitigation, and discussed how a county-level emergency manager could leverage their roles to improve mitigation outcomes across these categories of factors. Yet, at the time this study was conducted, no study had investigated the extent to which county-level emergency managers are engaging in mitigation activities, or what mitigation activities they are engaged in. This study attempted to fill a gap in the research by addressing these questions.

CHAPTER THREE: RESEARCH METHODS

Chapter Three is organized into five sections. The first section describes the methodological approach to this study. The second section discusses the intended population and sampling process for the study. The third section details the data collection procedures that were used. The fourth section explains the data analysis process that was utilized in this research. The fifth section discusses the study's limitations.

Methodological Approach

This study investigated the role of the county-level emergency manager in disaster mitigation using a qualitative research approach. A qualitative methodological approach allows the researcher to study "things in their natural settings, attempting to make sense of, or to interpret phenomena in terms of the meaning people bring to them" (Denzin & Lincoln, 2000, p. 3). A qualitative approach also allows the researcher to investigate how people construct their social reality (Taylor & Bogdan, 1998), which makes it an appropriate method for a study investigating how people perceive their own roles.

A qualitative methodological approach is also "particularly well suited to exploring issues that hold some complexity and to studying processes that occur over time" (Ritchie & Lewis, 2003, p. 5). Disaster mitigation meets both those criteria: it is a complex process involving many factors, which often occurs over a long period of time. It therefore made sense to investigate the role of county-level emergency managers in mitigation, which may also be complex and occur over time, using qualitative methods. Qualitative methods are also appropriate as a tool to "explore substantive areas about which little is known" (Strauss & Corbin, 1990, p. 19). Little to no empirical work on the role of emergency managers in

mitigation exists, which again suggests qualitative methods were an appropriate approach for investigating this study's research questions.

The specific methodological approach to this study was grounded theory. Grounded theory has been refined and updated over time since it was developed by Glaser and Strauss (1967) (see for example: Bryant, 2002; Charmaz, 2006; Glaser, 1978, 1998; Strauss, 1987; Strauss & Corbin, 1998). However, there are several components comprising an agreed-upon foundation for grounded theory. According to Charmaz (2006), a grounded theory approach should include:

- Data collection and analysis which occur at the same time
- The construction of categories and analytic codes from the data rather than from predetermined logically deduced hypotheses
- Use of the constant comparative method, which requires the researcher to make comparisons during every analytic stage
- The advancement of theoretic development at every phase of data collection and analysis
- The use of memo-writing in order to develop and specify properties of categories, explore the relationships between categories, and recognize gaps between categories
- A sampling methodology aimed at the construction of theory, rather than for representativeness (pp. 5-6)

Instead of attempting to impose existing theory or hypotheses on data, grounded theory research is designed to derive theory from the data collected (Strauss & Corbin, 1998). Doing so requires careful and consistent evaluation of the data by the researcher, ideally resulting in reflection, questioning and conceptual refinement (Strauss & Corbin, 1998). This interaction with the data allows the research design to be flexible, because researchers can investigate new

questions as they emerge and evaluate new ideas through theoretical sampling and collection of additional data (Charmaz, 2006). This approach is justified by grounded theorists because the close interaction with the data that is demanded, as well as the collection of additional data as needed or appropriate, makes it more likely that new theory will be consistent with reality (Strauss & Corbin, 1998).

Another benefit of grounded theory is that it allows for the creation of theory that is complex, detailed, and explanatory, rather simplified descriptions of phenomena (Charmaz, 2006). Due to the absence of existing empirical research on the role of county-level emergency managers in mitigation, and the complex and dynamic nature of the mitigation process, it was appropriate to begin an investigation on this subject using a qualitative, grounded theory approach with rich data from county-level emergency managers.

Population and Sampling

The population for this study included all county-level emergency managers in FEMA Regions III, V, and X. Region III comprises Delaware, Maryland, Pennsylvania, Virginia, West Virginia, and Washington DC. Region V comprises Ohio, Michigan, Indiana, Illinois, Wisconsin and Minnesota. Region X comprises Alaska, Washington, Oregon, and Idaho. It was determined that selecting FEMA regions, instead of the entire United States, was appropriate both for the purpose of expediency in selecting a sample and to control for factors like regional governance that could affect the data and analysis. These regions face diverse risks, including earthquakes (Fisher et al., 1999), tsunamis (Wood, Burton & Cutter, 1999), wildfire (Steele, Arno & Geier-Hayes, 1986), flooding (Ruggiero, 2012), volcanoes (Hoblitt, Miller & Scott, 1987), and others. These regions have also faced fewer disasters in recent months than others, including hurricanes in Region IV and wildfires in Region IX, and the researcher anticipated that emergency

managers would not be as time-limited as emergency managers managing responses or recoveries.

All counties in FEMA Regions III, V and X were included in the population, because all counties are threatened by hazards (CEMHS, 2018), and all counties have a responsibility to protect the populations they serve from these hazards (FEMA, 2018). Counties were selected as the unit of analysis, because county governments tend to have greater capacity than municipal governments (Waugh, 1994), they may be local agents of state administrations and have close administrative ties to state agencies (Waugh, 1994), but they also "serve as general-purpose local governments representing local interests and having strong local identification" (Waugh, 1994, p. 253).

This study used a combination systematic random sampling and purposive sampling to build the sample. Purposive sampling units are selected on the basis of "particular features or characteristics which will enable detailed exploration and understanding of the central themes and puzzles which the researcher wishes to study" (Ritchie & Lewis, 2003, p. 78). Because this study was interested in the role of county-level emergency managers in mitigation, county-level emergency managers were the most appropriate individuals to provide data regarding that role. County-level emergency managers can be thought of as a "homogenous sample," or people "chosen to give a detailed picture of a particular phenomenon... this allows for detailed investigation of social processes in a specified context" (Ritchie & Lewis, 2003, p. 79). In order to avoid a sample comprising mainly or exclusively small, poorly resourced counties, the list of counties was sorted using population. Systematic random sampling was conducted on the population-sorted list of all the counties in FEMA Regions III, V, and X. Using this list, individuals were selected to participate in interviews randomly, and were invited by email.

Initially, the researcher estimated that 50 interviews would be sufficient to examine the experiences of county-level emergency managers in FEMA Regions III, V and X, provide at least some degree of theoretical saturation, and reflect "the practical exigencies of time, money and other resources" (Seidman, 1991, p. 45). In consultation with the researcher's dissertation advisor, it was determined that a lower total number of interviews was sufficient to achieve theoretical saturation. Emergency managers were contacted in waves, with approximately 50 offices in each wave. Ultimately, 520 county-level emergency management offices were contacted, resulting in 42 completed interviews, representing a response rate of approximately 8.1%. Several emergency managers expressly declined to participate in this project (n=11) either because they were too busy (n=6) or because they did not play a significant role in mitigation (n=5). More often, emergency managers did not respond to the invitation at all. A breakdown of respondents by FEMA Region and state is provided in Table 3 below.

In order to properly contextualize participants' responses, it is also helpful to understand the communities in which they work. Appendix F provides relevant county information (US Department of Health and Human Services, 2013; US Census, 2018), including county government type, population, population density, jurisdiction type, 2019 county budget, poverty level, and location of the emergency management office, and is provided to help supply this context. In order to preserve anonymity, population has been rounded to the nearest thousand, density to the nearest integer, county budget to the nearest million dollars, and poverty level to the nearest integer percentage.

Table 3
Summary of Participants Interviewed by Location

Attribute		Number of Interviewees
FEMA Region		
 Region III 		12
 Region V 		17
• Region X		13
State		
 Washington 		5
 Alaska 		3
 Pennsylvania 		3
 Wisconsin 		4
 Virginia 		6
 Minnesota 		5
 Idaho 		3
 Ohio 		3
 Illinois 		2
 West Virginia 		2
• Oregon		2
 Indiana 		2
 Delaware 		1
 Michigan 		1
	Total Interviews	42

The ultimate goal of grounded theory research is achieving theoretical saturation (Charmaz, 2006; Glaser, 1998; Glaser & Strauss, 1967; Strauss & Corbin, 1998). Theoretical saturation may be defined as a point in data collection at which "no new properties, dimensions, or relationships emerge during analysis" (Strauss & Corbin, 1998, p. 143). Achieving theoretical saturation may be difficult, however, because "new" properties, dimensions and relationships can continue to emerge if the researcher commits to finding them (Strauss & Corbin, 1998). Due to this difficulty, Strauss and Corbin (1998) write that "saturation is more a matter of reaching the point where collecting additional data seems counterproductive; the 'new' that is uncovered does not add that much more to explanation" (p. 136). In addition to saturation, Seidman (1991) notes that the idea of sufficiency can be used to evaluate when enough data has been collected. The

hallmark of sufficiency is that there are "sufficient numbers to reflect the range of participants and sites that make up the population so that others outside the sample might have a chance to connect to the experience of those in it" (Seidman, 1991, p. 45).

Data Collection

Data collection and analysis for this study followed Charmaz's (2006) grounded theory model. In order to create grounded theory, the researcher must collect 'rich data,' or data that is "detailed, focused, and full" (Charmaz, 2006, p. 14). As a result, researchers using a grounded theory approach must use strategies for collecting data which illuminate "participants' views, feelings, intentions and actions as well as the contexts and structures of their lives" (Charmaz, 2006, p. 14). One strategy for collecting data that meets these criteria is intensive interviewing. This strategy is especially useful for collecting detailed data from people with relevant experience, as it allows "participants to describe and reflect upon his or her experiences in ways that seldom occur in everyday life" (Charmaz, 2006, p. 25). This study used intensive interviewing in order to allow county-level emergency managers to reflect on and discuss their experiences in mitigation in directed conversation (Charmaz, 2006), which gave them the opportunity to contribute valuable information about the role they have played during the mitigation process.

This study began in January 2019 after the research proposal had been approved by the dissertation committee, and the Institutional Review Board (IRB) had approved the IRB protocol, in January of 2019. The researcher collected data using telephone interviews.

Participants were contacted to request their involvement in the study via email. They were sent an information sheet explaining the specifics of the project as an attachment to the invitation email. See Appendix B for a copy of the invitation letter and Appendix C for a copy of the

information sheet. Once they consented to be interviewed, appointments for convenient dates and times were scheduled with the participants.

To facilitate the interviewing process, the researcher used an interview guide. The main questions were developed to be broad and open-ended, so that county-level emergency managers could discuss their understandings of their experiences in hazard mitigation in an in-depth fashion (Charmaz, 2006). The interview guide covered emergency managers' background and experience, a description of the county in which they worked, their definition of mitigation, what their county had done to mitigate hazard risk, and their role in mitigation efforts. In keeping with a grounded theory approach, the open-ended and flexible nature of the questions in the interview guide allowed "ideas and issues to emerge during the interview and interviewers can immediately pursue these leads" (Charmaz, 2006, p. 29). Appendix D includes the exact questions in the interview guide. A list of follow-up and probing questions were also used to explore leads that emerged during interviewing. Appendix E contains a list of potential probes that was used as a guide during the interview process.

Interviews were recorded using a digital recorder. The researcher uploaded recordings to her personal computer. Interview recordings were transcribed soon after they were uploaded, and identifying personal characteristics of interviewees were removed using code substitutions. No one but the researcher and dissertation advisor was able to access the audio files, transcriptions, and codes. Recordings were deleted as soon as interviews have been transcribed, and transcriptions and codes were destroyed when they were no longer relevant. The final dissertation contains neither participants' identifying characteristics nor codes. Where quotations are included, the researcher has protected participants' identities to the greatest extent possible.

Data Analysis

In keeping with grounded theory methodology, the researcher collected and analyzed data simultaneously. During data collection and analysis, the researcher followed analytical steps offering "a balance between science and creativity" (Strauss & Corbin, 1998, p. 13). Specifically, the procedures used in analysis have provided the research process with structure and rigor, while also giving the researcher the ability to "extract an innovative, integrated, realistic scheme from masses of unorganized raw data" (Strauss & Corbin, 1998, p. 13).

The coding process was integral to the organization of data collected into a meaningful structure (Charmaz, 2006). Specifically, coding helped the researcher "define what is happening in the data and begin to grapple with what it means" (Charmaz, 2006, p. 46). Coding occurred in two phases: an initial phase and a focused phase.

The initial coding phase required close reading of the early interview data to identify "analytic ideas to pursue in further data collection and analysis" (Charmaz, 2006, p. 46). In this initial phase of coding, each line of transcription was examined and coded individually. Despite the seeming arbitrariness of a line-by-line coding protocol (as not all lines will contain information relevant to the research question), such an approach encouraged the researcher to "remain open to the data and see nuances in it" (Charmaz, 2006, p. 50). This line-by-line approach to coding also provided insights that assisted with further data collection, including refinements to probing questions during the interview process (Charmaz, 2006).

After initial codes were developed during line-by-line coding, they were tested using constant comparative analysis, in which data was compared in and across interviews to locate similarities and differences (Charmaz, 2006). Through a constant comparative analysis, the researcher developed a set of codes that were significant, and which repeat across the data. The

codes created through this process were further evaluated during the second, focused coding phase. This phase included comparisons of early and later data in order to determine the appropriateness of the initial codes (Charmaz, 2006).

This process resulted in focused codes, which were then used to filter and integrate information from the larger data set (Charmaz, 2006). Analytical categories were refined through the process of comparing data to the focused codes, which improved the accuracy of the codes with respect to the data (Charmaz, 2006). The researcher experienced non-linearity during the coding process (Charmaz, 2006). That is, when this process revealed new insights into the data, earlier data was re-examined to investigate issues that were initially overlooked (Charmaz, 2006).

During coding, analytic memos detailing emerging categories were developed and used in analysis. These memos included categories' scope, applicability, definition, and linkages to other categories (Charmaz, 2006). From the analytical categories developed through this process, memos were used for theoretical sorting, in which theoretical links were sought between categories based on an examination of their relationships (Charmaz, 2006). The researcher also used diagramming to develop theoretical links, which offered a helpful visual representation of the relationships between categories (Charmaz, 2006). During this theoretical sorting and memo writing process, quotations representing various emerging themes were selected and listed with other quotations representing that theme, and labeled with information identifying the interview the quotation came from using the interview number and position data. By listing quotations under their corresponding theme, the researcher could ensure that the information from the interviews was accurately represented and that the themes were found consistently across the

data. After this process was completed, similar response data were reduced, and all labels removed before findings were reported.

Ultimately, the goal of this study was to produce meaningful research findings. Grounded theorists have advanced a number of criteria to evaluate grounded theory research. These criteria include: credibility of both the research and the researcher (Charmaz, 2006); research originality and its resonance with and beyond the population studied (Charmaz, 2006); the usefulness of research findings and researcher interpretations to practice and academia (Charmaz, 2006); fit, work, relevance and modifiability with respect to the relationship between theory and data (Glaser, 1978); presence of dense and well-developed categories (Strauss & Corbin, 1998); variation that is either built in or explained (Strauss & Corbin, 1998); and information or guidelines for action that either explain phenomena, suggest directions for future research, or guide programs (Strauss & Corbin, 1998). These criteria guided the researcher's analysis during this study, and every effort was made to meet the criteria for a high quality study identified above.

Limitations

This study has some limitations. It is possible that the researcher's bias and effects on participants may have influenced this study (Maxwell, 2006), in part because researchers "construct grounded theories through past and present involvement and interactions with people, perspectives and researcher practices" (Charmaz, 2006, p. 10). At the same time, the researcher relied on "checks" described by Charmaz (2006) as part of grounded theory, which are designed to reduce the likelihood of misinterpreting or misrepresenting participants' perspectives. The researcher also solicited feedback from the dissertation advisor while the processes of analysis

and writing are ongoing, in order to identify any unrecognized assumptions or biases on the part of the researcher, as well as any logical fallacies (Maxwell, 2006).

Due to the low response rate, this study may also have been limited by bias (Chambliss & Schutt, 2006). A particular concern is that only emergency managers who are actively engaged in mitigation were willing to participate, resulting in data that is not representative of emergency managers as a population. The study was also likely limited in its generalizability because the proposed sample represents only three of eleven FEMA regions, and only fourteen states out of fifty. The researcher hopes that this study will provide an important foundation for future research on this topic, which may include other FEMA regions.

Conclusion

This chapter reviewed the qualitative methods used for this study. Following Charmaz's (2006) grounded theory methodology, an interview guide was used to conduct telephone interviews with 42 county-level emergency managers about their role in hazard mitigation.

Grounded theory was also used as the methodology during coding and analysis of the collected data.

CHAPTER FOUR: FINDINGS

This chapter describes the main findings of this research from the 42 interviews conducted for this project. This chapter is organized into two sections. The first section details the key findings around the various roles county-level emergency managers played in disaster mitigation. The second section discusses possible explanatory factors around their described roles.

Allocation of Effort

By their own estimates, county-level emergency managers do not spend a great deal of time on mitigation. A few of the emergency managers interviewed for this study were mitigation specialists (n=3) and understandably spent the vast majority of their time engaged in the mitigation realm. However, for the generalist emergency managers (that is, emergency managers who were responsible for all phases of emergency management), the average percentage of time spent on mitigation was 26 percent, but the median was 13.75 percent and the mode was 5 percent. Although the average suggests that county-level emergency managers spend approximately a quarter of their time on mitigation—a percentage that would make sense given emergency management's four phases—the other measures of central tendency indicate that the average was affected by outliers in reported percentages. Indeed, at least two of these generalists considered everything they did to be mitigation, albeit their conceptualization of mitigation differed from the academic definition. If the outliers (i.e., mitigation specialists and generalist emergency managers who reported spending more than 75 percent of their time on mitigation) are excluded, the average percent of time spent on mitigation drops to 15.9 percent. These results might also be skewed by time allocation reports during years when the hazard mitigation plan was being written. A number of emergency managers (n=10) noted that the time they spent on

mitigation during the planning process was dramatically higher than the time they spent on mitigation otherwise. This suggests that county-level emergency managers are indeed investing at least some of their energies into mitigation—it is not being ignored—although they are not universally spending as much time in mitigation as the comprehensive emergency management framework might suggest they should.

Disaster Mitigation Role

Within that dedicated time commitment, participants in this study identified a number of role they may play related to disaster mitigation. These roles include a support role, an administrative role, a promoter role, a public education role, and a planning role. These roles are summarized in Table 4 below. In many cases, these roles were described generically, using language associated with the emergency management profession more generally. This section explores emergency managers' descriptions of their roles in disaster mitigation.

Generic Terms to Describe Role

County-level emergency managers did not necessarily view their role in mitigation specifically as distinct from their role as emergency management professionals more broadly—over a quarter (26%) relied on generic terms often associated with the description of the profession, e.g. coordination, facilitation, and relationship-building, to describe what they do related to the mitigation phase. An analysis of participants' use of these terms revealed that emergency managers seemed to have only a nebulous understanding of exactly how, and to what extent, the terms would be applied to disaster mitigation. Even when probed, many emergency managers struggled to clearly articulate what exactly they were doing to coordinate, facilitate, or build relationships and specifically with whom. For example, one county-level emergency manager said, "[T]hat coordination that comes with the role of emergency management, where

we just kind of coordinate things, and I think with mitigation it is just a good fit, because we do all that coordination already" (Participant 5). Another emergency manager, speaking about relationships, noted, "One of the things that is, again, building that network and having good relationships with your partners and your stakeholders" (Participant 29). And a third stated, "Our mission set is focused, centers around relationships, and being able to shake hands before a disaster happens" (Participant 7). It was clear that these emergency managers felt that in mitigation they should be, as in the other phases, taking some sort of actions related to the coordination and facilitation of activities, as well as the building of relationships—what was not as obvious, even after probing, was exactly how these generic functions were to be specifically employed in the mitigation context.

Table 4

Mitigation Roles Described by Participants

Role and Role Description	Number of Participants
Generic	N = 11 (26%)
 Includes activities typically associated with the emergency manager role, like coordinating between groups and facilitating collaboration 	
Administrative	N = 30 (71%)
• Includes paperwork, applying for and maintaining grants, locating and accessing resources	
Support	N = 10 (23%)
• Includes assisting partners, participation in mitigation processes (rather than running/being responsible for mitigation processes)	N = 14 (33%)
Promoter	11 (3370)
 Includes incorporating mitigation into other projects, reminding partners to consider mitigation, advocating for mitigation Public Education 	N = 28 (67%)
 Includes hosting public meetings, attending community events to discuss mitigation, meeting with community groups about mitigation 	
Planning	N = 37 (88%)
 Includes running a planning process, participating in a planning process, maintaining plan, following up on plan goals and targets 	

Support Role

Respondents did identify a variety of mitigation tasks and activities that were happening in their communities (e.g. buyouts, property elevations, facility retrofits, building code implementation and enforcement, levee/floodwall construction, planning). However, for the vast majority of these tasks and activities, many county-level emergency managers did not see themselves as driving these projects—they were not the primary "doers" of these mitigation tasks. Rather, almost a quarter (23%) of participants suggested that responsibilities for the completion of mitigation tasks and activities were distributed across a variety of departments and agencies within the local government, as well as other stakeholder groups, including private firms.

Emergency managers identified at least 38 different agencies or organizations involved in the completion of mitigation tasks and activities, specifically emphasizing the involvement of private businesses (n=9), the county floodplain manager (n=6), the state highway or transportation department (n=6), planning and zoning officials (n=6), nonprofits (n=5), and local hospitals (n=4), among others. Several county-level emergency managers, moreover, identified their role primarily as a support function (n=10). As summarized by one emergency manager, "We're a support function, so of course our engineers are involved, stuff like that, so they would probably be more of the lead on any physical projects, we would more be a support" (Participant 14). Another emergency manager helped paint a broader picture of what this support function looks like, saying,

This office helped find the funds through the grant programs to help finance the projects, kind of provide project management, oversight, reporting as required with the grant, work through with the vendors to make sure that appropriate request processes were filed, help the communities, kind of a leader and guide to make sure that they were doing what they needed to do, because they don't deal with this kind of project very often. Help work with the property owners and the

engineers on things such as the design of the structures, the maintenance and operations plans, finding the local matches for the grants to help offset the cost (Participant 36).

It was clear from the data that emergency managers considered themselves unable to complete the majority of the tasks and activities associated with mitigation, but rather their role was to support the efforts spearheaded by other offices or partners.

Administrative Role

Almost 75% of emergency managers indicated that they offered administrative support to county mitigation efforts (n=30). Almost half of participants identified the acquisition and maintenance of grants (n=18) as a way in which they contribute to the administration of mitigation activities. However, the extent to which respondents engaged in the grant process varied across jurisdictions with some playing a much larger role than others in the acquisition and maintenance of grants. It should also be noted that, in many instances, this funding acquisition was connected to preparedness activities, rather than mitigation activities. Within the grant activities of these 18 participants, for example, at least three specifically discussed applying for grants to acquire generators, two purchased communication equipment for their first responders, and at least one applied for a grant to install new tornado sirens, all of which are preparedness rather than mitigation activities.

For a few emergency managers (n=3), their role in this grant process was quite extensive, including identifying grants, completing grant applications, and being involved in grant administration. As one emergency manager stated,

We help to facilitate the applications for the hazard mitigation funding, and then we will get involved in the actual project development, just helping to sort of look at strategically if the places where we're looking at doing these things are the places we're actually trying to target (Participant 37).

In other jurisdictions, emergency managers only engaged in parts of this process. There were some emergency managers (n=7) who simply advised their jurisdictions on grant availability and potential applications, but did not themselves complete or assist in that applications process. One emergency manager noted, "We can provide them with advice based on our past experience and our knowledge of these types of projects and programs" (Participant 36). Still other emergency managers assisted different municipal agencies in completing the applications for grants identified in those other agencies (n=3). For example, one said, "I'm here to help you do the application or help [you with] the application and all that other stuff as well as part of my job" (Participant 34). And, a few others went beyond assisting to apply directly for those grants (n=2). One participant explained, "For a lot of the grant funding, at the county, the county becomes kind of the pass through for many grants, so we have to apply for grants on behalf of the municipalities" (Participant 5). Certain emergency managers were also engaged in grant maintenance (n=4). One said, for example, "EM financed that to make sure that the grant was being tracked appropriately and that everything was staying on the timeline" (Participant 11). Regardless of the extent of their involvement in grant management, the importance of securing and maintaining grants was noted by many emergency managers.

Emergency managers noted a variety of types of grants, many of which were FEMA grants (usually Hazard Mitigation Grant Program grants (n=12) or Pre-Disaster Mitigation grants (n=4), whereas Flood Mitigation Assistance was only identified by one participant). Others also identified state grants (n=2) and other, more specific grants (n=7), like forestry grants that could be used for mitigation. Two participants also identified homeland security grants, and two identified Project Impact grants as sources of mitigation funding. One participant noted they had received a Severe Repetitive Loss grant, and at least two participants knew that they had received

mitigation grants, but did not know which program they had come from. In at least one case, an emergency manager reported there being too many grants and grant programs to keep track of, and trusting the municipal offices on whose behalf they would apply to locate grant funding.

About twenty percent of emergency managers also reported doing paperwork and maintaining documentation (n=8) as part of their mitigation work. The paperwork emergency managers reported doing was typically connected to an existing mitigation project in the jurisdiction. In several cases, the required paperwork pertained specifically to financial reporting (n=3), but in the others, paperwork was described more generically. For example, one emergency manager explained,

I go, well, you need to get these people, you need this number of homes, and then I'll help them do the paperwork on that. So they do the legwork, but I'll do the paperwork, and that's really how it seems to be with a lot of this stuff, they want to do this stuff, they do the legwork and I end up having to do the paperwork (Participant 32).

In addition to filling out particular forms, these participants also suggested that gathering information and maintaining documentation related to potential projects was important. This was especially true for documentation connected to the receiving of grants or associated with ensuring that projects were eligible for funding through hazard mitigation plans. Documentation included taking photos and compiling relevant information. As described by one emergency manager,

I had previously documented, extensively documented, damage that had occurred during floods in the city of XXX. So, when the time came to show the need for the project, my office had pictures and data and damage assessment information to couple with the city to basically make the narrative and make the case for the project (Participant 7).

In all cases, the completion of projects was dependent on emergency managers completing administrative tasks related to documentation and paperwork.

A final administrative task related to mitigation in which a few emergency managers engaged was the hiring of contractors to help complete mitigation projects (n=3). In one case, the contractor was overall responsible for managing one of the county's fire mitigation programs, while in the other cases the contractors were hired to fulfill a specific function in an existing project or program. For example, one emergency manager noted in relation to getting a fire station built, "I've been working with an architect from out of state, but he's built fire stations, so he knows how to work the paperwork to get it through" (Participant 6). The presence of these contractors further reinforces the supporting role played by the majority of emergency managers in disaster mitigation.

Promoter Role

Although respondents did emphasize their support role and did not generally report driving mitigation projects, a third of the emergency managers (n=14) noted they played a role in promoting either mitigation activities or the integration of mitigation into other projects. One emergency manager explained,

[Mitigation] is not always heeded with as much gravity as we feel it should be, and so one of the things we try to do is work with our partners to make sure that they are thinking about hazards every time they make decisions about populations, development (Participant 12).

Another, discussing the importance of using partners to ensure mitigation is incorporated into other projects, said:

An emergency manager's actual role is kind of the end-to-end engagement, the ability to walk into a room of land use planners, of climate change people, of environmental people, of public works people, and have a good enough and broad enough knowledge of hazards and the relationships between projects, and public projects, and regulations, and hazards that they can help push things toward that better track and help people do things more safely and securely (Participant 30).

Some emergency managers (n=6) specifically discussed the importance of providing a forward-looking perspective to decision-making related to mitigation. One emergency manager suggested the example, "Hey, it's quick to build the school, but can we build the school in a way that provides a community resource after a major disaster?" (Participant 12). Other emergency managers also emphasized the importance of making sure this perspective is present when working with partners doing mitigation-related work. For example, one explained,

Mitigation is something that should happen in every phase, it should be something that is part of what we do in everything, and too often it ends up being opportunities for funding post-event, if you get the declaration. And you do the plan, all the municipalities do the plan, not to be forward thinking, and not to look for opportunities for growth, and the vision tends to be superficial (Participant 17).

Thus, while these emergency managers did not necessarily spearhead specific mitigation activities in their jurisdictions, they perceived that they had a role in championing the general cause of mitigation by threading it across community discussions.

Public Educator Role

Two thirds of the emergency managers interviewed (n=28) suggested that they had a role in educating the public about mitigation, including the need for mitigation and targeted mitigation strategies at both the household and community level. One emergency manager was quite blunt in their approach to education and outreach to the public: "As I put it, my job is to scare everybody" (Participant 24). Others described this role quite generally. One explained, "I think, primarily, we do education, we do a lot of outreach to people" (Participant 37). Others discussed the value of a public educated on mitigation, including one who said, "If you can use citizens that are concerned and educated and trained as a catalyst, then things happen" (Participant 31). Regardless, many of these emergency managers failed to robustly describe the

specific ways in which they were reaching members of the public in need of mitigation information, relying instead on broad and vague accounts of general public outreach efforts.

However, one tactic a few of the emergency managers described to educate the broader public was attending and participating in general community events (n=4). One emergency manager described a typical event, saying,

With the public, we have a very well-attended county fair, and our office is out there every single day for, I don't know, six, eight, ten hours, I don't know exactly how long we're out there. And we engage the public in conversations about mitigation (Participant 9);

A few other emergency managers attended specific events or meetings in order to reach more specialized audiences, including particular interest groups, neighborhoods, church groups, or those who might be affected by a proposed project (n=6). One, for example, met with a condominium community that was investigating a mitigation project,

We were in the process and so the county administrator actually asked us to go down and meet with them, that community, and we actually had like a little HOA, homeowners' association meeting, and we took the contractor with us, and they met with us, and we identified, or tried to talk through how mitigation works, and the whole process (Participant 41).

Still an additional few emergency managers described the virtual techniques and mechanisms their offices were using to disseminate information about mitigation to the public (n=3). For example, one explained,

We really try and see what's going on on the emergency manager side, but even our planning department, here's a website, send us an email, here's a digital form, call us, here's names and numbers you can call and talk to a person. We really try to utilize technology, and we'll have a virtual meeting, or send us a video clip, or whatever you want to show us, because we don't have good turnout for face-to-face meeting anymore, so we're really trying to give them other avenues to give us their input (Participant 39).

Although in many cases, the education and outreach were clearly related to mitigation specifically, it is likely that, in some cases, the role being described pertained more to

preparedness than to mitigation. Several emergency managers, for example, discussed educating the public on resources they would need for survival after a disaster (n=7) or informing the public of the role of emergency management in the community more broadly (n=3).

Planning Role

Emergency managers consistently emphasized their role in the mitigation planning process. Of all of the activities they mentioned, it was not only the most frequently identified (n=37), but also described in the most detail. It is worth noting that mitigation planning, is arguably not even a mitigation activity at all, insofar as it does not itself reduce hazard risk, but it is discussed here because of participants' perception that it their involvement in this process is a key part of their role in mitigation.

In many ways, the planning role incorporates elements of all the roles discussed above: it is largely administrative in nature, it requires emergency managers to promote and to educate the public about mitigation, and it necessitates many of the generic and supporting functions emergency managers reported engaging in, including coordination, facilitation, and relationship-building. However county-level emergency managers went about the planning process, they nearly universally identified planning as being in their domain, unlike any other task or activity. This section will describe emergency managers' planning role as it relates to the planning process, compiling elements of the plan, and plan maintenance.

A number of emergency managers described their involvement in the planning process, including the procedural elements of completing the plan (n=9). Broadly conceived, one emergency manager explained, "[W]e coordinate planning, we assist [municipalities] with plan development" (Participant 37). More specifically, emergency managers described, "facilitating the conversation, bringing the plan forward so folks know what it's about and what it can do for

us, and why it's in place; making sure they understand its purpose" (Participant 18) and meeting "with the cities and villages to discuss their concerns and get an idea as to what projects they'd like to see" (Participant 36). Another emergency manager highlighted the importance of the approval process for mitigation planning:

Obviously, the hard part on any mitigation plan is making it palatable enough to get adopted by the local jurisdiction, recognizing that it points to weakness and it points to things that need repair, and really in some ways it points to liability, but you're really saying we need to have the ability to harden these things long-term, so we need to be holistic in how we do it (Participant 8).

Other emergency managers discussed special circumstances in which changes to the plan needed to be made ("I had to convene special public hearings to be able to do an amendment to our plan" (Participant 7)), and frustrations with parts of this process:

Just right now, we were trying to upload our finances to the shared portal that you're required to do, and oh, by the way, the format of our plan doesn't quite fit so we've got to reformat the plan again and change it so it would fit into the portal (Participant 28).

Regardless of exactly how the planning process was structured, logistics associated with planning, including setting up meetings, arranging communications, liaising among different partners, and information gathering, was an element of planning that many county-level emergency managers (n=10) described as part of their role in planning. As one emergency manager said, "I pool our resources and advise them as to what meetings we need to get everybody together and I come up with the master plan on how to achieve our goals and what the goals would be, and now I'm working with the different groups letting them do the work to achieve the goals" (Participant 38). Another said,

I did a lot of footwork for [the contractor], we had to set up a meeting place to meet with all these people. And of course, there's privacy, I had to make sure it's some private area where we could go and discuss things. And sometimes communication was a factor, because they couldn't speak English, so we had to get translators in there. So that was a role we played (Participant 4).

Emergency managers also described their role in helping to compile components of the mitigation plan itself. The two major sections emergency managers described were hazard and risk identification (n=5), and goal and strategy development (n=3). With respect to hazard identification, one emergency manager said:

We did our whole process, all the public meetings, went through I'll call that a mini-THIRA [threat and hazard identification and risk assessment], but they wanted us to rank everything, put numbers to it, we kind of went through a THIRA process for top ten hazards in the area (Participant 32).

Other emergency managers discussed the development of mitigation goals and strategies: "We come up with our new projects and goals for us to meet for the next five years" (Participant 27). Although a few emergency managers noted writing other parts of the mitigation plan, these were the two most common components, as well as the two most important components of the plan overall.

After the plan itself is completed, emergency managers continue to maintain it, by checking in with planning partners and ensuring that plan timelines are being followed (n=9). One county-level emergency management department developed a program for plan maintenance:

We started establishing something called a plan to project pipeline, which involves, we go out and give them a workshop, as soon as they finish their plan, we establish a workshop with their planning partners to start converting some of their strategies into projects (Participant 30).

Another emergency manager described a somewhat less structured plan maintenance process:

We do an annual report, so sometime around the beginning of October I'll send around to the municipalities and the university, a list of, a copy of the goals and objectives and their projects, and find out what the status of their projects is and what they have done to, during the year, for those goals and objectives that were identified through the planning process. We've used it in a way to do our annual

check-ins with our mitigation plan, as to how people are doing on the action items they had identified during the planning process and see if they've identified any others that have popped up since or finding after an event, what kind of... What sorts of damages have occurred and what sorts of mitigation projects we can bring to help prevent those sorts of damages in the future... As they take on projects oftentimes it'll just be a quick check-in of hey, we're doing the thing, it's going to be, or, usually a major thing like if it's a school district, hey, we're looking to, we got this bond levy and as we develop new schools we're looking to bring in better mitigation principles into those schools so they'll be able to be a shelter site after an earthquake (Participant 25).

Although plan check-ins were reported multiple times, some emergency managers emphasized that they did not necessarily expect municipalities to complete projects that were listed in the plan (n=4). One said, for example,

As far as the planning, I think sometimes [municipalities are] reluctant to put a project in there because they're concerned because they think if it's written in the plan they have to do it. We try very hard to emphasize that, because you list it in the plan, it doesn't mean you have to do it, it means you'd like to do it (Participant 36).

These emergency managers noted the importance of maintaining the plan, even if it did not result in more mitigation. The reasons county-level emergency managers engaged in a planning role, and the reasons they engaged in the other roles described here, will be described below.

Emergency Manager Characteristics and Role

Although a number of common roles emerged from the data, only one participant reported being engaged in all of these roles (Participant 36), and others reported being engaged in as few as one (Participants 1, 2, and 25). Characteristics of participants and their counties, including county size, emergency management staff, years of experience, education, and background were investigated to determine whether these may have affected the roles that were reported. To do so, each participant's reported county size was coded as small, medium or large; their staff size was coded as having more than one staff member, having one full-time staff member, or having one part-time staff member; their reported years of experience were recorded;

their education was coded as having a graduate degree in emergency management, having an undergraduate degree in emergency management, having relevant educational experience, or having no relevant educational experience; and their background was coded as first response-related, or other (the remaining participants had only worked in emergency management). The researcher did not run any statistical analyses to examine the relationships between these characteristics and reported roles, however a visual investigation of this data did not reveal any relationships or patterns.

Explanatory Factors

Chapter Two presented a hypothesized grouping for explanatory factors – municipal factors, including leadership and government characteristics; community factors, including community support and hazard profile; and external factors, including resource availability – however, this research suggests a somewhat different grouping schema. The factors that explain emergency manager role based on the findings of this research would instead include conceptual confusion, response and preparedness orientation, financial resource factors, funding requirements, planning requirements, additional resource factors, competition between mitigation and development, resistance to mitigation, and engagement in mitigation. This section will discuss these categories of findings in more depth.

Conceptual Confusion

When asked to define mitigation, participants' definitions provided some insight into how they conceptualize the roles they play, and the activities they consider to be mitigation.

Mitigation, as defined in Chapter 2, refers to "sustained action to reduce or eliminate risk to people and property from hazards and their effects." Two critical components of this definition, sustained action, and risk reduction or elimination, were absent in either all or many of the

definitions provided by emergency managers. Of the 31 definitions provided by participants, none referenced sustained action, and more than half (n=17) did not reference risk reduction and/or elimination. Instead of risk reduction or elimination, a number of participants discussed impact reduction instead, which is conceptually related but distinct (n=12). The remaining definitions, i.e. the five that failed to include reference to sustained action, risk reduction/elimination, or impact reduction were either confused (for example,

Mitigation, to me, it starts from prevention to providing a service during a disaster and continuing to provide that service after that disaster, and to plan, to have a plan, implement the plans, and once you get through with the disaster, after action report, see what you can do better, see what you need to improve on, what other resources we may need, identify those and make improvements in case we have another disaster (Participant 15)),

or were describing a different concept, like resilience (for example, "I could quote a textbook but I don't want to do that. I'd say, to help adapt to the environment" (Participant 1)). A complete list of definitions is included as Appendix G.

Although many of the activities emergency managers discussed as being mitigation activities did, in fact, constitute mitigation, many were instead preparedness activities, and in some cases, recovery activities. For the purposes of this project, preparedness may be defined as,

A dynamic state of readiness that is dependent on context and the process of completing the activities required to effectively take immediate action to save lives, property, and/or the environment; and to effectively restore, reshape and rebuild the parts of life impacted by disasters (Nojang, 2015),

and recovery defined as, "a differential process by which individuals and households, organizations, and jurisdictions seek to restore, rebuild, and/or reshape that which has been directly or indirectly impacted by a hazard event through pre-event planning and post event action" (Montano, 2017). It is possible that the fact that the definitions emergency managers

provided differed significantly from the definition scholars have agreed upon affected the activities they considered mitigation.

Some of the activities that the researcher would consider to be preparedness or recovery rather than mitigation include acquiring generators (n=11), distributing weather radios (n=4), and doing damage assessments and debris management (n=2). Mitigation seemed to be most often confused with preparedness (n=14), and only occasionally with recovery (n=2). Exemplifying the confusion between preparedness and mitigation, one emergency manager said, "When they have a car kit, when they put stuff in their car to be more prepared in case of an emergency, they're mitigating the effects of something that they might not have full control over" (Participant 9). Another said, "They're preparedness and part of preparedness, like I said, is education, which is mitigation" (Participant 5). The activity misidentification illustrated above suggests that county-level emergency managers may not understand mitigation well enough to engage in it as comprehensively or as actively as scholars might hope. It also suggests that county-level emergency managers were doing even less mitigation than they reported, because at least some of the activities that they included in their time allocation estimates do not constitute mitigation.

Only a few emergency managers (n=4) identified education or expertise regarding mitigation as important, but the conceptual confusion highlighted above suggests that emergency managers are not, in fact, well educated in mitigation. Without understanding mitigation either conceptually or in practice, county-level emergency managers could not be expected to value or know how to drive mitigation. One participant noted that many emergency managers they had encountered were insufficiently knowledgeable about mitigation,

And we certainly need to educate our emergency managers differently about the value of mitigation to the programs, and I don't recall ever seeing any applicant

for a position or for an internship ever reference any knowledge of hazard mitigation (Participant 17).

Another, echoing this sentiment, remarked that they did not necessarily have the experience or expertise necessary to engage in mitigation,

I think that's been part of the problem, is the fact that there's this expertise level that everybody wants you to be at, or you should know how to do this. Well, you gotta remember, this is the first time for me too so I don't know all the answers (Participant 32).

The deficits emergency managers demonstrated in understanding and education of mitigation helps to explain the limited role that they reported playing.

Response and Preparedness Orientation

Another factor that seemed to limited the extent of county-level emergency managers' involvement in mitigation was the response and preparedness orientation they and their departments exhibited. Participants varied considerably with respect to the extent of their response and preparedness orientation, and none specifically articulated that they felt response and preparedness were more important than mitigation. This orientation was, however, demonstrable in statements they made. For example, one emergency manager, describing the amount of time that they spend on mitigation, said, "Unfortunately, it's very low, just because of being one person and having other projects that take priority" (Participant 20). In describing their experiences in the introductory part of the interviews, emergency managers were more likely to discuss their response functions and activities (n=15) than their mitigation functions and activities (n=4).

The way county-level emergency management departments were structured also contributed to a response and preparedness orientation. One emergency manager, whose office was located in a fire department rather than independently operated, said,

I'm a very strong proponent of never having emergency management under any other operational agency, like a response agency. And in Virginia, the average office is going to be under fire. And that really does, the perspective skews that way. You go, we put the water on the fire and we drive the injured to the hospital, and then we go back, and we're done! We clean up our equipment, we restock, we're ready for the next one. And I think that long-term perspective is severely limited when it's under an operational agency (Participant 17).

Of the 42 participants in this study, 13 reported that their department was located within a first response agency. Even though many of these emergency managers were not dissatisfied by this arrangement (n=10), the frustration expressed by the emergency manager quoted above reinforces the importance of those who have some influence over the mitigation process understanding its value. If emergency managers are housed in agencies that do not recognize that emergency management is more than preparedness and response, or if emergency managers themselves do not recognize this, it follows that they would play only a limited role in mitigation.

Financial Resource Factors

One of the most commonly cited reasons emergency managers articulated for not spending more time doing mitigation was the absence of resources, especially financial resources (n=28). One emergency manager, for example, said, "There's some things, mitigation-wise, I think the county could do, but it's the funding that's the tough part" (Participant 3). These participants alluded to the fact that, in contrast to many response, recovery and preparedness activities, many mitigation projects and activities are capital-intensive and expensive. Although structural mitigation is obviously costly, other mitigation activities, including code enforcement, can also be expensive. Especially in poorly-resourced government contexts like low-population rural counties with small tax bases (n=3), without access to outside sources of funding, mitigation may simply be unaffordable. Because funding for mitigation is so limited, and

mitigation itself is so expensive, it follows that county-level emergency managers would spend a significant amount of their allotted mitigation time engaged in fund-seeking activities through their administrative role.

Although some funding for mitigation comes from county and municipal budgets, including matching funds, many mitigation projects are funded by FEMA. Emergency managers from smaller counties described difficulties they had had in accessing this federal funding. In reference to this challenge, one emergency manager said, "I'll tell you where [federal mitigation funding] is going, it's going to the communities that can afford to do [mitigation]. I get that. If they can afford to do that why wouldn't they? I wouldn't sit there and not take the money!" (Participant 7). Another emergency manager from a small, rural county echoed this challenge, saying,

And as a smaller population it seems like it's kind of difficult to, because you've got to have a cost to benefit ratio, and without the population, it's hard for them to justify, probably, the cost to benefit ratio for you (Participant 23).

As this emergency manager explained, funding mitigation projects in these smaller counties may be unappealing for FEMA, because associated cost savings may not be as high as in larger or more densely populated counties.

Several participants struggled to finance non-flood mitigation projects specifically through FEMA (n=4), in spite of the fact that flood mitigation was not appropriate for their community. One emergency manager explained,

FEMA, when they come out with the pre-disaster money, it's pretty much, they know where they'd like to see that spent. For elevation or acquisition of properties, to get things out of floodplains. The notification type systems and stuff, that's been challenging, is the best way to put it. We've been able to do certain things so far with some creative funding sources, to try to get this moving forward (Participant 25).

Although emergency managers still engaged in their administrative role to attempt to access this funding, the limited availability of that funding, and the restrictions associated with it, helps to explain the small percentage of time allocated to mitigation and the limited role of county-level emergency managers in mitigation.

One source of financial resources for mitigation was eligibility for federal funding following disaster declarations. A few emergency managers (n=3) discussed this mitigation-related disaster effect. One emergency manager, for example, explained,

Well, as luck would have it for you, anyway, having had numerous federally declared disasters, one of the benefits, if you want to call it that, from having disasters, is there's hazard mitigation money after the event, some percentage of what the total damage was in your county. And over the years, we've had several projects that have significantly reduced the impacts that flooding has (Participant 24).

In these cases, the administrator role county-level emergency managers play was central to achieving mitigation. Conversely, without having had a hazard event, emergency managers were limited in their ability to access funding. One noted,

It's fortunate that we don't have great flooding issues, but on the other hand, if we did, it would give us something to go after. I live in the county next door, and there's a little town that flooded numerous times, and they have gone after mitigation money to buy housing and knock it down and make green space, that kind of thing... But we just don't have that kind of repetitive issue here in [my county], which is a good thing, but it takes those big mitigation projects off the table (Participant 34).

Even in the absence of access to FEMA funding, it seemed that county-level emergency managers could still access money and other resources from nontraditional sources. A few participants (n=3) described seeking funding from federal agencies other than FEMA to supplement their mitigation resources. One emergency manager said,

But I take the resources that I've got and I do the best that I can do with them and I get a lot of things done. And I'm a big one on using other people's resources. If I

can get something out of the forest service or the BLM or whatever, I'm not hesitant at all about using their resources (Participant 38).

Other respondents looked for local sources of supplemental resources (n=6), including working with state or county partners to locate funding and other resources that could be useful, benefiting from some of the generic coordination, liaising, and relationship-building activities several of them described. One emergency manager explained their experience, saying,

The big urban areas are getting all the funding, you know, we're not. For us to get funding, we've done creative things here, with other grants and stuff to get some of these projects taken care of. I've been very impressed with some of these municipalities, go-get-it type thing, to get projects done, where they use some, save up and use some of their own money, use people to dig up the road and put in bigger culvert pipes to help move water along (Participant 25).

The limited funding available for mitigation prevents county-level emergency managers from being more engaged in mitigation. However, the need to access this funding likely drives the administrative role that they play.

Funding Requirements

Many emergency managers reported struggling to meet federal and state requirements in order to access funding (n=17). What exactly was required varied by state, but for at least one respondent, requirements were so time consuming that they simply did not have time for mitigation. This emergency manager described all the tasks they had to complete in order to continue receiving state funding, saying,

I am required to do and jump through a lot of hoops with emergency planning, training, exercising. I now have to exercise all 32 core capabilities that the federal government has identified, I have to exercise all 32 within a four-year timeframe. So I have to exercise, not the agency, me personally, I have to exercise eight core capabilities each year. In between maintaining the emergency operations plan, maintaining the hazard mitigation plan, maintaining the agency inter-operability communication plan, I have to do all of this. I have to do mitigation, I have to do preparedness, response recovery, and prevention. I have to do all these things. Not to mention, because I'm full-time, my county feels like, well, you know, there's emergency management stuff to do, but a lot of people don't have full-time

emergency managers, so you also need to do the website, you need to do our social media, information management, you need to do asset management, insurance stuff, you pretty much need to do all of it (Participant 7).

Others described challenges meeting the requirements of specific programs, including funding programs (n=8). The challenges participants faced differed by funding source, state, and project type, but included having to expend resources to do research that supports project applications. One emergency manager explained,

When we speak directly about our mitigation plan and the funding that's put behind it, the pre-disaster mitigation funding and the environmental historic preservation research that has to go into any one project, those are big barriers (Participant 18).

Another specific challenge with some federal funding programs is that, to receive funding, applicants must demonstrate historical losses; whereas in many cases, applicants may be aware that a mitigation program or project is necessary, even if no losses have occurred (n=3). One emergency manager said, for example,

They want historical data, so even if there's nothing that historically has impacted a certain structure, but you know that it could be, you're trying to prevent it from happening, there just is no, the funding, there's a grant opportunity, but they are kind of stringent on their criteria and their competitiveness (Participant 20).

County-level emergency managers facing these specific structural challenges might still engage in their administrative role to try to access this grant funding, but their frequent inability to access it functionally limited their involvement in mitigation.

There were several additional challenges participants noted regarding meeting FEMA or state requirements. Sometimes these were relatively minor ("I was finishing up an MBA and I never got officially certified because I didn't have time to take Creative Budgeting. It was kind of absurd, but nonetheless" (Participant 17)), but many created significant barriers to engagement

in mitigation (n=6). One emergency manager, discussing ever-increasing regulatory requirements, said,

I mean, literally, these towns operate off hardly nothing. And the state continues to take more out of us. And the federal government along with the state government continues to put more regulations on us. I don't see how it works in the long-run (Participant 7).

Other emergency managers noted that what may work for the average county is inappropriate for their county (n=3). In these cases, emergency managers acknowledged that mitigation requirements may have value, but they were difficult or impossible for their communities to fulfill. One such emergency manager explained,

Being a rural county, our resident population, you know, of what we have, the funding is just not there. I got municipality that's got 56 people living in it. And there's, they have two roads, that's it. They don't have a building, they don't have equipment, they don't have anything. But yet they're still supposed to do everything like everybody else (Participant 25).

Planning Factors

Although there were numerous challenges to acquiring funding for mitigation, participants noted that it would be impossible to access federal funding at all without having a state- and FEMA-approved mitigation plan. In fact, the reason many emergency managers gave for the value they placed on the mitigation plan is the relationship between maintaining an approved plan and access to state and federal funding (n=13). One of these emergency managers explained, "Every county in the state has to have this MHMP [multi-hazard mitigation plan] because if you don't, you don't get funding" (Participant 16).

How, and the extent to which county-level emergency managers were engaged in the planning process was associated with how this process was organized, which in turn was explained by planning-specific resource factors. Those most engaged were the emergency managers whose county negotiated the planning process independent of a contractor, followed

by emergency managers whose county hired a contractor to direct or assist with their planning process, followed finally by emergency managers whose county participated in a regional planning process. Those who ran the planning process independently generally did so because they had sufficient staff resources. In contrast, those who lacked these staff resources, but, in many cases (n=8) had access to specific grants, hired contractors in order to lead this planning process. The final group, those who participated in a regional planning process, lacked the resources to run their planning process independently, but were located in a region with an active planning group.

Of the emergency managers interviewed for this research, a minority were in counties which completed their mitigation plans independently (n=9). Half of the emergency managers interviewed for this research, however, reported using a consultant for at least part of their planning process (n=21). There was a continuum of involvement in consultant-led or -mediated planning processes for study participants. Of these, several (n=8) reported having the consultant run the entire process, but may have provided information about the community or hazards as requested. Other emergency managers who hired consultants emphasized that they continued to play a meaningful role in the planning process (n=11). For example, one emergency manager described their role in a consultant-led process:

We normally hire a contractor to go through the [mitigation planning] process with us, and they have several meetings where we go over what did we have previously, what are our needs, are there any gaps, provide a full updated information of, like, how many flood events did you have, how many tornadoes from the last one, and then we sit down, and as a region we come up with, we go over the hazards, and when they put them into the program they go, is that accurate? (Participant 27)

Still other emergency managers hired consultants to assist them with part of the planning process, but ran the majority of it themselves (n=2). Finally, a few emergency managers

participated in a regional planning process, rather than conducting such a process for their county specifically (n=4). One such emergency manager explained, "Other than being one of the members at the table when we're updating [the mitigation plan], I don't head that" (Participant 24).

Although the planning mandates that help ensure that all, or nearly all, jurisdictions have hazard mitigation plans, has some obvious advantages, there is arguably at least one major downside. Namely, a few emergency managers reported undertaking mitigation planning processes in order to be eligible for grants, rather than with the intention to actually implement any of the projects or programs described by the plan (n=3). To this point, one emergency manager said,

There's a default position, in order to create a plan that sufficiently addresses enough of the requirements to be eligible for hazard mitigation grants, has a declaration. So what happens is that the minimum becomes the maximum for too many of us (Participant 17).

In spite of this objection to having emergency management funding tied to mitigation planning, the consensus of participants in this study was that there was too little federal funding available, this funding was too difficult to access, and the mechanisms associated with accessing this funding were too complicated.

Additional Resource Factors

Many emergency managers, especially in small counties, reported insufficient staff resources to do as much mitigation as they would like to (n=8). Seven of the participants in this study worked in one-person offices, and four of these were only part-time emergency management employees. For these emergency managers, there simply were not enough hours in their work day to do mitigation in addition to all the other duties they were responsible for. One emergency manager in such a situation said, "So for me, [I wish] I had a partner who could just

go run around and talk to these people all day, let's do this! Let's do this! Yeah, I'll clone myself, you go do this, and I'll do this" (Participant 32). A few emergency managers were also affected by insufficient municipal staff or other municipal resources (n=3). One participant discussed their frustration with the technological and personnel difficulties they faced when trying to do basic mitigation tasks,

When we get the notice of the application time period for grants, getting that out to everybody. Here's another thing, they do them online now, before you can apply online you've gotta take a class. My municipalities, I'm going to say probably at least a third, that's never going to happen. It's just, they're just not that type of skill level. So, that, while doing that on the computer is nice, then we need to do something to provide some help to areas that don't have those abilities, either through internet and/or the skill set for the people there that do it. A lot of the secretaries here are in their sixties and seventies. I just got one, I got one municipality that finally got a computer (Participant 25).

These non-financial resource limitations likely also contributed to the relatively small amount of time county-level emergency managers spent engaged in mitigation. Although not explicitly stated, emergency managers in counties with large emergency management staffs and access to adequate technological resources were apparently engaged more often in mitigation activities.

Competition Between Mitigation and Development

In addition to lack of funding and other resources, county-level emergency managers also had to contend with municipal and community priorities that compete with mitigation, especially development and economic concerns (n=8). One emergency manager, for example, said, "[There is a] fundamental disconnect between planning and development and hazard mitigation/emergency management" (Participant 30). Although distinct from lack of access to resources, this particular challenge may result from mitigation's incompatibility with either maintaining or expanding the community's tax base and financial position. One emergency manager explained how this may happen as a result of buy-outs: "They're losing that massive tax

base because so many of the homes are gone now [due to buy-outs]" (Participant 14). Another emergency manager discussed a related conflict between mitigation and development:

The issue is, I mean, let's face it, if we're really talking about mitigation and alleviating a life hazard along the coast line, letting people build multi-million dollar homes along the beach is not the right way to do it (Participant 41).

This problem was exacerbated by the presence of developers in conversations about mitigation. One emergency manager, for example, described how developers could dominate these discussions.

Right now, the largest voices in our community are the developers, and they, even with this last thing, they like to sell stories, like, oh that will just make property more expensive! And when they're the only one shortcutting, so a resident then can't get water, fuel or anything delivered because they've built this substandard road, but don't have a large voice that likes to spark outrage when those bills come up, and then, so sometimes it just takes, well this is what it really is, and, it's like social media, you know, it's, sometimes the loudest voices are certainly not the most educated or intelligent. Got to counteract that (Participant 31).

Counteracting these development and economic pressures likely requires the deployment of the promoter and public educator roles by county-level emergency managers.

Resistance to Mitigation

In addition to financial and development challenges, emergency managers also discussed the presence of resistance to mitigation in their communities. There was by no means universal resistance to mitigation efforts. Some emergency managers (n=5) felt they had strong support for mitigation. One of these said, "We have an, we're just so lucky, we have an incredible amount of credibility with all our communities, with everything that we do" (Participant 29). Few emergency managers reported active opposition (n=3), but in some cases there was resistance (n=12) to mitigation generally or specific mitigation projects. A number of participants (n=6) described their constituents as "libertarian," or as generally opposed to government projects,

which they might perceive to interfere with their lives. One emergency manager who experienced resistance like this said,

Generally, there's support, but in a rural county like this, there's always some portion of the county that resents that federal government involving themselves in the business, and these rural areas, they're more of a survivalist type element, and want to take care of themselves, very independent and feel like the government is imposing if they take money, then they're obligated to the government, which is not the case in most cases (Participant 38).

At the same, time, some of these county-level emergency managers (n=4) described how educating the public, via the public educator role, about mitigation could change these attitudes. For example, one emergency manager highlighted one potential benefit associated with engaging in public education: "We help create the champions with our neighboring communities so that they have that assistance when they don't have time to focus on it" (Participant 12).

In a few cases (n=4), resistance to mitigation took the form of litigation against the mitigating jurisdiction. Actual or potential lawsuits discouraged mitigation in some particularly litigious counties. One emergency manager described their experience with mitigation litigation,

It's interesting for a jurisdiction, and I can tell you, in the last ten years that I've been here, we have been sued five times at least for things that we did in the form of mitigation to benefit somebody's property's safety and integrity. And some years down the road we end up getting sued because of it. And it's like, well, ok, but your property was even worse before we did the good thing to benefit it. So we have been bitten many times for doing mitigation actions too. We think we're doing the right thing when we move forward with a project, and then somebody comes and sues you for it later (Participant 8).

In some cases, these difficulties may lead county-level emergency managers to engage in their promoter role, but in the case of this study's participants, they led emergency managers to disengagement with the mitigation process. For example, when asked if they would consider trying to promote mitigation in such a context, one emergency manager said, "We identified

[mitigation] and made mention of it [through our mitigation plan], but for me to go out and push the issue is probably not something that would happen on a normal basis" (Participant 41).

Some participants (n=8) also noted that there was confusion about what mitigation was among the community, which could also create animosity or resistance to mitigation. One emergency manager noted, for example, "I think the big one is really trying to get people to understand what mitigation is" (Participant 32). At least two emergency managers described community confusion specifically regarding parks that were designed to be flooded. One outlined their frustration with this experience,

Probably the biggest challenge that we run into has to do with people not understanding why we do mitigation. So every spring, without fail, I listen to the radio, and somebody calls in and says, this million-dollar park we built in the last year or two is under water again, for the umpteenth time, why did we invest all this money in something that's under water again (Participant 22).

When community members are ignorant of mitigation and its benefits, it may be challenging to convince them to approve new mitigation projects or programs. Again, however, by taking an education role, emergency managers reported being able to assuage some of this misunderstanding or hostility toward mitigation. For example, one emergency manager, who fielded similar questions about a flooded park, described their experience educating the governor of their state about mitigation:

They took land and made it a city park, so [the river] would flood that. Now, ten years later... and this is how I met the governor, I had the governor show up at my office going, I'm getting calls about the city park flooding, you're doing nothing about it, and it's like, well sir, it was designed to flood, that was part of the mitigation plan that we did to save the town. And he looked at me and turned to his aide and said, is that true? And he goes, yeah, sounds good (Participant 32).

Although public education may not always result in such a simple and satisfying conclusion for county-level emergency managers, it can be a useful tool for overcoming confusion about mitigation.

Local government officials may also misunderstand mitigation and mitigation programs (n=4). If politicians and other officials do not understand the value of engaging in mitigation, they may not be willing to commit political capital to promoting mitigation projects. As one emergency manager explained,

They're just not familiar with the programs, they're, most of these elected officials are part-time people with full-time jobs and other things going on that taking on this work is not something that they do all the time and they probably don't have the opportunity to familiarize themselves, do some of the research, so they tend to rely on folks like myself to kind of be that guidance for them (Participant 36).

In one extreme example, a local government simply refused to participate in mitigation or emergency management initiatives. This emergency manager described this situation,

Like, one community said, you know, a certain amount of staff time count towards your totals for the declaration and they said, "nope, we're not going to put that in there. That's just our city's policy that we do what we need to do to take care of our community during an emergency and we're not going to put in for that." And we were like, "but do you understand that counts towards our total." So that was a back and forth and they said "nope. We're absolutely not going to put that number in the city's total" (Participant 9).

Although this behavior was not reported by many participants, it reflects the fact that county-level emergency managers may engage in mitigation only at the discretion of the jurisdictions in which they work, and the local governments they work with may be more or less cooperative.

In response to confusion and misunderstanding from local government officials, emergency managers also reported deploying their educator and promoter roles. One emergency manager, for example, said,

You know, my elected officials up here, I'm going to say, average age is probably 68 or 69, I still have a lot of those that didn't even graduate grade school, so I try to write the plan on a sixth grade reading level. They really, when you meet with them, their basic thing is, tell us what we need to know and what you need us to do (Participant 25).

Even when emergency managers knew that education and promotion was necessary to accomplish mitigation in similar situations, it was not always possible. One emergency manager said, for example, "They just don't understand the paperwork and the cost of that, and sometimes I just don't have that time to go out and teach them" (Participant 32).

Engagement in Mitigation

Another factor that county-level emergency managers must navigate in mitigation is the extent of engagement or buy-in exhibited by various community stakeholders. Buy-in reflects the community's support for mitigation, whereas engagement refers to the extent to which the community is aware of and interested in mitigation. One dimension of this factor is political buy-in. When local elected officials focus on the politics of mitigation (n=6), it can limit the extent of engagement in it. For example, one emergency manager discussed the focus some politicians place on the financial consequences of mitigation:

But the other stuff that we try to do they do not, it's just harder to get more of a buy-in, because they don't see the, not the value, but they want to see their dollars be put to good use, especially politicians (Participant 26).

The broader community's relationship with mitigation may also affect emergency managers' ability to play an active role in it. One emergency manager, emphasizing the importance of this dimension, said, "[T]here has to be local involvement, and local buy-in, and I think that's where a lot of efforts fail" (Participant 31).

Some emergency managers were satisfied with the amount of engagement in their communities (n=3). Describing their county, one such emergency manager said, "It's a really proactive community. I can see that other communities may have issues, but everyone in XXX is just very involved" (Participant 11). Many others, however, were frustrated by an absence of engagement or interest in mitigation (n=13). One emergency manager, describing part of their

planning process, said, "Part of the mitigation plan, we have to have a public meeting where we show our mitigation plans. The ones we did four years ago, nobody came to, at all. Not a single person" (Participant 5). In order to defray this lack of engagement, some emergency managers embraced their public education and promoter roles, to mixed success. For example, several emergency managers (n=4) noted that they had taken all FEMA and state-recommended steps to increase turnout at public meetings, but they continued to experience low levels of engagement.

Engagement may also be affected by the experience of hazard events. Several participants reported that their communities were interested in mitigation only in the wake of a hazard event (n=7), whereas they were apathetic about mitigation otherwise. One emergency manager said, "It seems like they're only interested if there's a disaster" (Participant 14). Another described a specific incident that motivated action in one of the municipalities in their county,

So they get kind of complacent. In fact, XXX, who had a tornado in 1981, their tornado siren does not work. And I've been after them for two years to get their radio fixed, and they're dragging their feet. Now, with the tornado outbreak in XXX that we've had, they're all like, well what about my siren! (Participant 34).

Although this represents a challenge in many cases, in instances where county-level emergency managers could effectively capitalize on their promoter or public educator role, they could use this post-disaster time to encourage their communities to implement mitigation. One emergency manager, for example, described how they use their public education role following a disaster to encourage mitigation,

If you can plan, and you can educate people on what to do, what not to do, warnings, everything, I think that sort of mitigation has been our most successful, if you tie it and do mitigation after the disaster. I think that is probably the biggest thing that we had, is people saying, ok now, we had that tornado warning and the sirens went off, and we heeded that, we went into the basement and look, this tree was through the living room (Participant 5).

Relatedly, county-level emergency managers reported that most mitigation projects are for hazards with high salience; that is, hazards that people are most aware of and concerned about (n=12). One emergency manager explained how mitigation projects in their community tended to be tailored to low-impact, high-frequency hazards, rather than high-impact, low-frequency hazards:

So I just think that's something that's interesting to note, that I think people, at least in my experience with my colleagues, they tend to tailor their mitigation activities around what I'd call the usual suspects, and don't really put a lot of energy into really addressing the really worst-case things, because, going back to what I started with, it's really hard to get policy makers to fund something that may not happen in the next 50 to 100 years (Participant 37).

In order to overcome this challenge, county-level emergency managers had to engage in both their public education and promoter roles. One emergency manager noted that they work to do this in their community, explaining,

I want to know what risk is going to look like 100 years from now, or fifty years. I want to be able to tell somebody, if you want to build your house to be here in fifty years, you need to adopt a plan of resilience standard, not a current code. And not just encourage people to think about it in terms of building up to the current code, but also to think about it in terms of your resilience standard (Participant 30).

Conclusion

This chapter has summarized the findings of this research. It was revealed that although emergency managers do not spend a great deal of their time on mitigation, they play several distinct roles within mitigation. These roles are: a support role, an administrative role, a promoter role, a public education role, and a planning role. It also discussed factors that may help explain these roles, which include: conceptual confusion, response and preparedness orientation, financial resource factors, funding requirements, planning requirements, additional resource factors, competition between mitigation and development, resistance to mitigation, and

engagement in mitigation. The next chapter will discuss these findings in the context of the mitigation literature and emergency management broadly.

CHAPTER FIVE: DISCUSSION

This chapter discusses the findings of this research, and contextualizes them in terms of the mitigation literature and the comprehensive emergency management framework. This chapter is organized into four sections. The first section considers the extent to which the typical emergency manager interviewed for this study meets the description of the idealized emergency manager defined in Chapter Two. The second section considers how the findings of this study compare to the findings of a similar study. The third section addresses the gap evident between the idealized emergency manager and actual emergency managers in mitigation. The fourth and final section discusses implications of this work for education, practice, and policy.

Emergency Manager Role as Theorized and as Practiced

Chapter Four discussed the roles that emergency managers reported playing within mitigation. Based on these self-described roles and activities they engaged in, it is possible to evaluate the extent to which they meet the description of an idealized emergency manager in mitigation described in Chapter Two. This section will review these idealized characteristics, and compare them to the characteristics participants identified as being part of their own role in mitigation.

Based on the mitigation literature reviewed for this study, the ideal emergency manager has sought to decrease barriers to mitigation in a number of ways. This emergency manager has looked for and accessed financial resources, both by ensuring they are compliant with federal programs, and also by applying for and managing mitigation grants (Baudoin & Wolde-Georgis, 2015; Becker & Reusser, 2016; Brody et al., 2009; Fois & Forino, 2014; Hakaloba et al., 2016; Hill & Gaillard, 2016; Labossière & McGee, 2017; Manyena, 2013; Nelson, 2014; Raju & van Niekerk, 2013; Rivera & Wamsler, 2014; Sipe & Vella, 2014; Stehr, 2007; Wiek et al., 2010;

Yoon, Youngs & Abe, 2012; Zahran et al., 2010). They have also acquired the resources necessary to increase the organizational capacity of their departments and offices (Brody, Kang & Bernhardt, 2010; Manyena, 2013; Nelson, 2014; Olshansky et al., 2008). They have sought out opportunities to increase their own discretion and control over the mitigation process (Stevens, 2010).

Of their mitigation activities, emergency managers were particularly adamant about their role in reducing financial barriers. Their primary mode of reducing these barriers was engaging in a mitigation planning process in order to be eligible for federal funding. Emergency managers, however, also discussed assisting in, applying for, and maintaining mitigation grants. Virtually every emergency manager who participated in this study had an approved mitigation plan in place (n=41), but there was more variation in their success acquiring grants. Some emergency managers, especially in large and already well-resourced communities, reported having no problem accessing mitigation funding, whereas others, often in small and poorly-resourced communities, struggled to meet the requirements associated with mitigation grants (n=5). In spite of their efforts to access funding, however, financial barriers to mitigation were persistent and seemed to be a critical limiting factor for both mitigation implementation and emergency managers' involvement in mitigation.

Partially as a result of the financial barriers facing emergency managers, emergency management offices frequently lacked sufficient organizational capacity to participate in mitigation. Organizational capacity can be recognized by communities' ability to anticipate hazards, make informed decisions about mitigation, and implement effective policies (Brody, Kang & Bernhardt, 2010), and is affected by access to financial, staff, and technological resources; the ability to communicate and share information; and strong leadership (Brody, Kang

& Bernhardt, 2010). Emergency managers consistently reported challenges with organizational capacity, both in terms of how organizational capacity manifests, and the factors that explain it. Most obviously, few emergency managers reported having implemented effective mitigation policies, and several mentioned the presence of policies that likely increased risk. For example, a few emergency managers (n=3) were located in jurisdictions without building or development codes, and they did not seem to believe this was likely to change.

Many emergency managers also lacked the resources associated with organizational capacity. Several participants were the only staff member in their department, and of these, a number were only part-time employees. Although organizational capacity here specifically refers to the capacity to achieve mitigation, it should be noted that emergency managers had to work to maintain even a low level of organizational capacity generally. For example, in addition to seeking funding for mitigation projects, many emergency managers also reported applying for grants, and spending large amounts of time meeting requirements in order to be eligible for state funding to support emergency management offices and pay staff salaries. In spite of the challenges associated with organizational capacity, emergency managers reported fewer activities associated with increasing mitigation-specific organizational capacity than with acquiring financial resources. Many participants reported being dissatisfied with the extent of their organizational capacity, whether it was because they lacked adequate staffing or technological resources, or because their department did not have sufficient funds for general operations. Some emergency managers reported that they hoped to hire additional staff or to acquire additional resources (n=3), but only one had concrete plans to accomplish these goals.

The idealized emergency manager has been a mitigation advocate. For example, they act as a policy entrepreneur and regularly propose mitigation projects, or they take an approach that

mobilizes others to advocate for mitigation (Birkland, 1996; Labossière & McGee, 2017; Olshansky et al., 2008; Olson & Olson, 1993; Solecki & Michaels, 1994; Stevens, Berke & Song, 2008). They have also developed a policy community that they can deploy in advocating for mitigation (Birkland, 1996). They have demonstrated a dedication and commitment to mitigation as a strategy for reducing hazard impacts (Brody, Kang & Bernhardt, 2010; Stevens, Berke & Song, 2008). When their community does experience a disaster, they have taken advantage of the aftermath as an opportunity to implement more mitigation (Asgary et al., 2007; Baker, 1977; Fois & Forino, 2014; Healy & Malhotra, 2009; Hill & Gaillard, 2013; Labossière & McGee, 2017; Le Masson, 2015; McSweene & Commes, 2011; Olson, Olson & Gawronski, 1998; Paul & Che, 2011; Tunstall, Johnson & Penning-Roswell, 2004).

Here again emergency managers reported mixed success. There was variation among participants, but many of the tasks associated with advocacy were not reported. Few emergency managers reported acting as a policy entrepreneur, although several discussed the importance of this role. It should be noted that the literature identifies having resources to influence mitigation as an important dimension of the policy entrepreneur role (Solecki & Michales, 1994).

Emergency managers were once again stymied by their lack of access to resources, which made engaging as policy entrepreneurs much more difficult. Similarly, few emergency managers were engaged as mobilizers of their communities, although some identified public education as a necessary task to create mobilizers within their communities. No respondents specifically reported being part of a mitigation policy community, but a few emergency managers were engaged in mitigation working groups, which may accomplish some of the same ends. Overall, it seemed that emergency managers understood the importance of advocacy, but were only able to engage in it a limited fashion.

Although emergency managers were not necessarily involved in these specific advocacy roles defined by the literature, some did report engaging as promoters of mitigation in meetings with government officials. They also emphasized the role that they play in advocacy following disasters. Emergency managers saw these events as an opportunity to promote mitigation projects when interest was highest and people were least likely to object. They also deployed their administrative role in this post-disaster period. Emergency managers, however, emphasized their role in increasing the availability of mitigation funding after disasters more than their role in generating engagement and buy-in for mitigation projects in this time period.

In addition to advocacy, the theoretical emergency manager has been attentive to their community, and has tried to increase the independence of their community in making mitigation decisions by working to augment community capacity (Norris-Raynbird, 2005). When making decisions about mitigation, they have centered community needs and preferences (Rubin, Saperstein & Barbee, 1985). They have worked to engage their community in mitigation (Andersson-Sköld & Nyberg, 2016; Evans-Cowley & Gough, 2008; Hakaloba et al., 2016; Le Masson, 2015; Manyena, 2013; McNamara & Des Combes, 2015; Norris-Raynbird, 2005; Olshansky et al., 2008; Oulahen & Doberstein, 2012; Passerini, 2001; Sipe & Vella, 2014; Stevens & Berke & Song, 2010), through the development of coalitions (Birkland, 1998; Goldstein, 2008; Labossière & McGee, 201; Lane, 2000; Lu et al., 2017; Mannakkara, Wilkinson & Potangaroa, 2014; Meo, Ziebro & Patton, 2004; Olson, Olson & Gawronski, 1998; Raju & van Niekerk, 2013; Sipe & Vella, 2014; Weischelgartner & Pigeon, 2015), and an effective media and communications strategy (Baudoin & Wolde-Georgis, 2015; Evans-Cowley & Gough, 2008; Fois & Forino, 2014; Sipe & Vella, 2014; Stevens, Berke & Song, 2010; Weischelgartner & Pigeon, 2016).

For most of the emergency managers interviewed for this research, there was no reason to believe communities were not making decisions independently, although there were some exceptions. Participants mentioned decision-making being influenced by developers in several cases, and the hiring of contractors for some mitigation projects and for mitigation planning may also have influenced independence. In fact, the literature suggests the influence of these two groups – that is, developers and contractors – should be limited in order to increase community independence (Norris-Raynbird, 2005). Outside influences, especially from groups whose interests were not driven by community safety, tended to reinforce community preferences for economic wellbeing over hazard mitigation. As a result, in some instances, the community needs and preferences emergency managers observed were in opposition to mitigation, although in the absence of these influences, their communities may have preferred public safety.

Another reason communities seemed to prefer development to mitigation was the absence of engagement in the mitigation process. Emergency managers reported making attempts to increase engagement, but they encountered numerous barriers. Efforts to increase engagement did include developing media and communications strategies, but effectiveness of these strategies was somewhat mixed. In spite of attempts to modernize communications through social media, emergency managers reported having a difficult time communicating with the public. Although at least one emergency manager reported that, since they had been in their position, their agency's Facebook profile had attracted a significant number of new followers, engagement in activities like public meetings remained low. In contrast to the efforts emergency managers made regarding communications, there was little explicit evidence that emergency managers were working to create coalitions within their communities, likely because there was too little overall engagement for coalitions to be viable.

The ideal emergency manager has also sought to work effectively and improve themselves professionally. They are professionally ethical (Le Masson, 2015), and have taken advantage of educational opportunities to become more technically adept (Brody, Kang & Bernhardt, 2010). They are knowledgeable or experienced about hazard risk and mitigation strategies (Brody, Kang & Bernhardt, 2010; Rubin & Barbee, 1985; Stevens, 2010), as well as about their community's values and priorities (Rubin, Saperstein & Barbee, 1985). Finally, they actively work to 'de-silo' local government so that there is more communication and information-sharing across departments (Brody, Kang & Bernhardt, 2010; Raju & Van Niekerk, 2013).

The final literature-derived factors, effectiveness and improvement in their professional capacity, was present to some extent in all of the interviews, but the effectiveness of participants' professional efforts were hampered by structural factors. With respect to ethical operation, there was no evidence that emergency managers were operating unethically, although it seems unlikely that participants would have reported unethical behavior even if it were present.

Emergency managers did not, for example, report explicit corruption. However, there may have been marginal cases: in at least one case, an emergency manager did report taking actions that they had been specifically instructed not to take by county officials. It is also possible that county relationships with certain stakeholder groups (e.g. developers) may have been present but undisclosed in interviews. To the extent that these actions or relationships may have been unethical, it is also possible that they affected emergency management engagement or effectiveness in mitigation.

There are undoubtedly additional ethical dimensions to mitigation work, especially with respect to policies that prevent low-income communities from addressing hazard risk, a problem

that was alluded to by several participants. Even within communities, it may only be possible to implement mitigation in areas with high property values, leaving residents who are already likely more vulnerable to hazard risks without mitigation protections. Although these ethical dimensions of policy came up in interviews, county-level emergency managers did not report active engagement in efforts to reform policy. In other words, participants seemed to perceive their advocacy role as limited to encouraging local mitigation adoption, rather than as including influencing state or federal mitigation policy.

Emergency managers did not generally report seeking additional discretion over the mitigation process. In most cases, although they seemed frustrated by the amount of mitigation occurring in their counties, they apparently did not identify increased discretion over mitigation as a strategy to increase the amount of mitigation taking place. Emergency managers reported networking and relationship-building across agencies, but it seems that these efforts did not necessarily rise to the level of de-siloing. In addition to networking and relationship-building, meaningful de-siloing would require concerted and coordinated efforts to improve communication and interoperability in a sustained way (Raju & Van Niekerk, 2013). Emergency managers reported challenges they faced as a consequence of departmental silos, especially as they pertained to contradictory and overlapping requirements associated with different permitting agencies. This would suggest that, had emergency managers been attempting to de-silo local government through their relationship-building and networking, they have not succeeded yet.

Participants did report engaging in educational opportunities to improve their understanding of their emergency management role generally, but not necessarily about mitigation specifically. Much of this education represented a requirement to maintain accreditation or state funding, although several emergency managers also reported seeking out

education independent of requirements. Some emergency managers also had either bachelors' degrees or graduate degrees in emergency management or a related field, which was likely helpful for mitigation in at least some cases. However, the extent to which emergency managers failed to accurately conceptualize and define mitigation strongly implies that there were deficiencies with the educations that they had received, whether in a traditional academic setting or in practice-oriented education and training. A summarized comparison between the characteristics of the idealized emergency manager and of participants in this study is provided in Table 5 below.

Although not explicitly discussed as part of the idealized emergency manager role in Chapter Two, Comprehensive Emergency Management and emergency management scholarship would also strongly suggest that emergency managers should play a significant role in mitigation (Britton, 1999; Mileti, 1999). Moreover, emergency managers have tasked themselves with engaging in all four phases of emergency management (FEMA, 2011). Because mitigation is one of four presumably equally important phases, it would follow that emergency managers should spend a significant portion of their time working in mitigation.

Most of the emergency managers interviewed, however, reported that they were not spending as much time working in mitigation as they would like. As noted in Chapter Four, most emergency managers spent well under a quarter of their time on mitigation, and those who reported spending much larger amounts of time often described activities that do not constitute mitigation. The estimates of time allocated for mitigation was also seemingly influenced by mitigation planning processes, which, in at least some cases likely inflated participants' responses.

Table 5
Idealized Activities of Emergency Managers in Mitigation Compared to Activities of Study
Participants in Mitigation

Idealized ActivitiesStudy ParticipantsIncrease availability of financial resourcesParticipants maintained• Maintain compliance with federal, state and local requirementscompliance, applied for and maintained grants. Limited
• Maintain compliance with federal, state and local compliance, applied for and
1
requirements maintained grants. Limited
 Apply for and manage mitigation grants success in increasing
 Increase organizational capacity for mitigation and organizational capacity.
emergency management generally Participants discussed
Engage in mitigation advocacy importance of advocacy but
• Act as a mitigation policy entrepreneur had largely not helped
Mobilize other stakeholders develop advocacy in a
Develop a mitigation policy community meaningful way
Take advantage of 'windows of opportunity' Participants emphasized
Work with their community importance of community,
• Center community needs and preferences in decision but struggled with
making engagement and with
 Engage community in mitigation decision making improving community independence
 Develop community coalitions
Develop effective media and communications strategies Participants engaged in
Improve professionalization activities to improve
Operate ethically professionalization but were
• Seek educational opportunities in mitigation often stymied by structural
• Develop knowledge or experience in mitigation factors
De-silo local government

Findings and Analysis from a Similar Study

As noted in Chapter Two, a study was published in March of 2019 which sought to address a similar research question (Samuel & Siebeneck, 2019). In order to avoid biasing the data analysis of this research, this paper was not reviewed until after data analysis was complete. Yet, the similarities between that paper and this research are notable. Samuel and Siebeneck identified six mitigation roles described by emergency managers: administrator, collaborator, coordinator, fund seeker, advocator, and public educator, as well as an overarching leadership role. There are notable commonalities, especially the presence of the administrator, promoter/advocate, public educator, and generic/collaborator and coordinator roles. In this

research, "fund seeker" was largely subsumed under the administrator and planner roles. Although Samuel and Siebeneck did not specifically identify a planner role, they did find that emergency managers were heavily involved in the mitigation planning process. Samuel and Siebeneck specifically suggested that, due to the large number of their participants who identified collaborator, coordinator, and administrator mitigation roles, emergency managers should play a leadership role in hazard mitigation planning.

It is also important to note that there are several key methodological and analytical differences between the studies. With respect to study design, Samuel and Siebeneck interviewed both municipal and county-level emergency managers, and all of their participants were located in North Central Texas. In contrast, the researcher of this study interviewed only county-level emergency managers across regions of the United States. It is also notable that, like many of this study's participants, Samuel and Siebeneck defined mitigation as "actions taken to reduce the loss of life and property by lessening the impact of disasters," excluding the sustained or long-term dimension of the concept. The authors also fail to identify specific mitigation activities outside of planning, so it is impossible to know whether this conceptualization of mitigation influenced their analysis. In at least two of their interviews, emergency managers seemed to confuse mitigation and preparedness, but the causes and potential consequences of this confusion were not examined.

Moreover, although the findings were quite similar, in contrast to Samuel and Siebeneck, this study found that, in aggregate, emergency managers were engaged in a support capacity rather than a leadership capacity. Notably again, however, Samuel and Siebeneck do not specifically identify activities that constitute leadership, so although their participants described their role using 'leadership' language, it is difficult to determine what they feel leadership looks

like in practice. The studies agree, however, that emergency management engagement in mitigation could be more substantial. Where Samuel and Siebeneck propose emergency management-centric solutions to challenges for emergency management engagement, however, this research instead suggests that these challenges are largely external. The following sections discuss the reasons challenges emerged, and implications of this research for some of these external communities.

<u>Understanding the Gap Between Theory and Practice</u>

Although the emergency managers interviewed for this research bore some resemblance to the idealized emergency manager described in Chapter Two, there are a number of key differences. The idealized roles emergency managers were fulfilling with at least some degree of effectiveness included seeking funding for mitigation, and making efforts toward improving advocacy, communications, de-siloing, and their own educations in emergency management and mitigation. In contrast, emergency managers often struggled to increase organizational capacity, increase the extent of mitigation advocacy, were not members of mitigation policy communities, were not explicitly seeking to increase the independence of their communities, had not developed communications strategies that effectively increased engagement, had not increased their discretion over the mitigation process, had not been successfully fully de-siloed government agencies, and had only a limited education pertaining to mitigation.

Moreover, in spite of the fact that mitigation is purportedly one of four equally important phases in emergency management according to the Comprehensive Emergency Management framework, emergency managers generally reported spending a relatively small amount of time on mitigation-related activities. In fact, several emergency managers contacted for participation in this study declined on the grounds that they considered mitigation to be outside of their

professional purview (n=5). The fact that emergency managers are, by and large, not doing as much mitigation as other emergency management activities, and are involved in mitigation in a support rather than leadership capacity, creates several issues. First, it suggests that, because emergency managers themselves are not doing 'enough' mitigation, there may not be enough mitigation being done overall. It also raises questions about the differences between the roles played by theoretical and actual emergency managers in mitigation, including why they are not doing as much mitigation as researchers would expect. Each of these issues will be addressed in the subsections below.

Adequacy of Mitigation Efforts

Regardless of how much mitigation work emergency managers are doing, there is consensus within the scholarly community that there is not enough mitigation being done overall (see for example: Alesch & Petak, 2002; Deegan, 2007; Rubin & Barbee, 1985). The relative lack of mitigation puts communities at risk today, and these risks will only be amplified as development pressures and the disaster-related effects of climate change put more people in harm's way. It is therefore critical that emergency management scholars and practitioners identify strategies to increase the amount of mitigation being undertaken.

To the extent that the purpose of the Comprehensive Emergency Management framework is to encourage emergency managers to be meaningfully engaged in all four phases of emergency management, it is clear that challenges remain. Although the reason enough mitigation is not being done may be attributed to a widely distributed network of people, it is still important to address whether weaknesses in Comprehensive Emergency Management contributes to this inadequacy. Were emergency managers to spend more of their time in mitigation, it is possible that more mitigation would be implemented.

Why Do Theorized and Actual Emergency Managers Differ?

The differences between the theorized and actual emergency manager in mitigation are likely the result of at least three different challenges. The overarching barrier to mitigation, and to emergency managers playing a more robust role in mitigation, is resource limitations. Having inadequate resources affects virtually every aspect of emergency managers' roles, from the time they spend trying to access additional resources, to their lack of effectiveness across their work in mitigation. Resource limitations prevent emergency managers from having the staffing, time and funding necessary to create community coalitions and policy communities, and to develop and deploy more effective communications and de-siloing strategies. They make seeking out additional knowledge and education much more difficult. Fundamentally, they limit the extent to which emergency managers may expand their organizational capacity, even if they desire to do so.

A second challenge arguably comes from emergency managers themselves. Specifically, some emergency managers consider mitigation to be outside of their purview entirely, and have therefore not sought out opportunities to play a larger or more effective role in mitigation.

Although in some cases, emergency managers seemed to be complacent about their professional limitations, it is likely that, for many, this seeming complacency was actually due to an understanding of their resource limitations. Participants in this study reported being frustrated by the lack of control they had over mitigation processes, although few, if any, discussed a plan or even a desire to increase the control they exercised. Emergency managers are also limited in their approach to mitigation by the absence of professional guidelines for how they ought to engage in mitigation. For example, neither the International Association of Emergency Managers nor the National Association of Emergency Managers, the two largest emergency management

associations in the United States, provides robust guidance on emergency manager roles and responsibilities during mitigation. If they had access to resources or documents clearly articulating what their appropriate roles and responsibilities in mitigation were, it seems likely that emergency managers would engage in those roles more often and more effectively.

Finally, it seems that emergency managers may simply lack the knowledge they would need to be more effective players in mitigation. Although there were areas in which they could have played a more substantial role (for example, establishing or participating in policy communities, seeking out greater knowledge of mitigation, and effectively de-siloing local agencies), it seems emergency managers were not always aware this was the case. Notably, these were all barriers to mitigation that were not frequently noted by emergency managers, in contrast to financial challenges and difficulties meeting requirements, for example. If emergency managers do not recognize barriers to mitigation or activities they could be engaged in, it follows that they would not seek out solutions to those barriers, or engage in those activities.

The fact that participants did not pursue the above mitigation-facilitating activities is likely at least partly the result of structural and cultural forces in the emergency management profession. Although some states or local emergency management offices may require that emergency managers have specific certifications, training or education (McEntire, 2018), there is no required national, standardized accreditation system for emergency managers (EMAP, 2019). As a result, knowledge about emergency management generally, and mitigation specifically, is not evenly distributed throughout emergency management practice. In addition, many emergency managers come from first response backgrounds – nearly one in two emergency managers interviewed for this research, for example, had backgrounds in first response. As discussed below, the response-orientation that may result from a first response background likely biases

emergency manager knowledge in favor of response and preparedness, to the detriment of mitigation.

Why Are Emergency Managers Doing Too Little Mitigation?

Chapter Four addressed some of the explanatory factors that influence both the amount of mitigation being done, and the roles emergency managers play within mitigation. However, these factors should be contextualized by considering the external forces that give rise to them.

Emergency managers operate in a space that is underappreciated, underfunded, and often misunderstood (Labadie, 1984). They must attend to differing and frequently changing laws and guidance at the local, state, and federal levels (Wamsley & Schroeder, 1996). Emergency management has also not been fully professionalized (e.g., Carlson, 2015; Cwiak, 2019; Farris & McCreight, 2014), which contributes to the fact that emergency managers may lack education and guidance about their roles and responsibilities that would benefit them.

As discussed in Chapter Four, emergency managers also tend to be oriented to response and preparedness rather than to mitigation. Not only does this affect which activities they focus on, but it also reflects their expertise. Mitigation requires a different skill set than response and preparedness, relationships with different stakeholders, different kinds of resources, and responsiveness to a different set of challenges (FEMA, 2018). When emergency management offices have only one emergency manager, it is impossible for that person to specialize in any of the phases. Moreover, when the emergency management profession draws so many of its practitioners from first response fields (in this study, 18 out of 42 participants had first response backgrounds), it is likely that any specialization they do have will be in response or preparedness. If emergency managers are to engage in mitigation, the response and preparedness orientation must shift. This could result from increasing the resources of emergency management

departments so that mitigation (and recovery) specialists may be hired, or from further professionalization which incorporates a focus on mitigation (and recovery) education.

Perhaps one of the reasons that emergency managers did not spend more time on mitigation, and were not more effective in getting mitigation implemented, is the fact that participants did not seem to have a robust understanding of mitigation as a concept. In fact, as discussed in Chapter Four, emergency managers defined mitigation in ways that differed significantly from the most commonly accepted definitions of mitigation. They also described activities that suggested they were thinking about mitigation in nonstandard ways. This is not necessarily their fault. Neither the academy nor the practitioner community has defined mitigation consistently, if they have defined it at all. As discussed above, Samuel and Siebeneck (2019), for example, offered a definition of mitigation more in line with the respondents in this research than the definition provided here.

Guidance and definitions from FEMA do not make matters much clearer. Unlike

Comprehensive Emergency Management, FEMA now recognizes (apparently in some, but not all cases) five, rather than four, "mission areas." In addition to mitigation, response and recovery, FEMA has included prevention and protection, and removed preparedness. Here, prevention and protection are both oriented to terrorism and homeland security hazards. Moreover, a brief search of FEMA's website also revealed several significantly different definitions of mitigation, including not only the definition used in this study, but also, from the National Mitigation

Framework, "the capabilities necessary to reduce the loss of life and property by lessening the impact of disasters" (Department of Homeland Security, 2016) and,

Mitigation is the effort to reduce loss of life and property by lessening the impact of disasters. In order for mitigation to be effective we need to take action now—before the next disaster—to reduce human and financial consequences later (analyzing risk, reducing risk, and insuring against risk). It is important to know

that disasters can happen at any time and any place and if we are not prepared, consequences can be fatal (Federal Emergency Management Agency, 2018).

These definitions are clearly related to the definition that has been roughly agreed upon by emergency management scholars. However, they are notably missing any reference to sustained or long-term efforts. If the definitions of mitigation being used do not include an element about actions being sustained or long-term, virtually any emergency management activity could be deemed a mitigation activity. For example, an effectively executed response could be considered mitigation, because there will likely be fewer impacts in a good response than in a poor response. To make matters worse, some of the "core capabilities" FEMA identifies as being mitigation include activities that would be described by most emergency management scholars as preparedness activities, including planning, and public information and warning. With these definitions from FEMA in place, it is no wonder that emergency managers displayed conceptual confusion about mitigation. Contributing to this confusion is the fact that FEMA, as one participant noted explicitly, provides mitigation funding for at least one preparedness activity, acquiring generators, through the Hazard Mitigation Grant Program (HMGP) (FEMA, 2015).

Likely as a result, one of the most notable features of emergency managers' responses to questions about the mitigation tasks they or their counties undertake is the fact that so many of their answers represent activities that the scholarly community does not consider mitigation. Of the at least 57 different activities reported, almost half represent preparedness, recovery, or non-emergency management related activities (n=24). At least eleven participants identified acquiring generators, and six identified hazard notification as mitigation activities they have undertaken or would like to undertake, both of which are preparedness activities. At least two respondents also identified damage assessments and debris management – both recovery activities – as mitigation activities. More than half of participants also identified education and

outreach as a mitigation activity. Indeed, in some cases this may constitute a mitigation activity, but, as discussed briefly in Chapter Four, in many cases is more likely a preparedness activity.

This definitional confusion leads to another set of questions: if emergency managers are uniformly describing mitigation in unconventional ways, what does this suggest about the definition of mitigation being used in this research? Is it wrong? If wrong, how should it be modified? Is it sufficiently common? Are the academic and practitioner communities using it consistently? Would calling the constellation of activities associated with mitigation something different be helpful? Should mitigation and preparedness be combined in order to reduce confusion?

It is beyond the scope of this research to evaluate the extent to which the current definition of mitigation is the best possible, or most complete, definition. Such a definition would lead researchers and practitioners to be able to categorize activities as being exclusively in the mitigation domain. Based on this study, it is clear that the definitions emergency managers are using does not meet this criterion. It seems likely that, were the 'correct' definition of mitigation in wider usage, fewer emergency managers would confuse mitigation and mitigation activities with other phases. Researchers and government agencies should certainly use a consistent definition and identify common activities when discussing mitigation. The definition and description of such a critical element of emergency management scholarship and practice should not be modified without significant support and justification. These factors and their implications will be discussed at greater length in the final section of this chapter.

<u>Implications for Communities of Policy and Practice</u>

The findings of this research suggest that emergency managers play a number of important roles in mitigation, but the extent to which they engage in these roles, and the

effectiveness of their participation in mitigation processes, seems to be limited. Limitations are, in turn, affected by a number of factors established in the findings chapter. Reducing barriers to county-level emergency managers' participation and effectiveness in mitigation will require stakeholders across emergency management functions to engage in a number of coordinated activities. Specifically, agreeing upon a definition of mitigation, recognizing mitigation as a critical function of local government, and rethinking current approaches to mitigation planning will be critical.

Implications for Policy

A key challenge to county-level emergency managers' ability to engage more often and more effectively in mitigation is the extent to which policy incentivizes and emphasizes the importance of mitigation. Not only must mitigation specifically be recognized as a critical element of emergency management, but emergency management must be recognized as a critical function of local government. This recognition should compel multiple levels of government to shift emergency management organization and resourcing through changes to policy.

Emergency management must first be recognized as a unique and important role within local government. Although other agencies or organizations, including response and planning agencies, have roles within emergency management, there are functions of emergency management that cannot effectively be done by other departments. It therefore impedes emergency management functions to locate emergency management agencies within other agencies or departments. Although doing so may create efficiencies in certain tasks and activities, it subordinates many other critical emergency management activities to other departmental priorities. An emergency management office located within the fire department is

more likely to be perceived as a response office; whereas an emergency management office located within the planning department is more likely to be perceived as a mitigation office.

Of course, in order to be an independent office, emergency management must have sufficient resources. Minimally, emergency management offices must have the resources to engage in all aspects of their role, and across all phases of the emergency management cycle. Without adequate staff resources, emergency managers may struggle to find the time necessary to meet the requirements for accreditation, which may be a prerequisite for state or federal funding. Staff resources are also important because different skill sets and knowledge bases are required for emergency management's various phases and activities. A planning expert, for example, may not also have expertise in running an emergency operations center, and cannot realistically be expected to do so. It is therefore important that there be enough staff with appropriate expertise to accomplish all the activities of emergency management.

Another dimension of recognition of emergency management as a critical function of local government is the way local government ranks community priorities. This is especially true with respect to development. Local governments certainly may have strong incentives for encouraging development at the expense of public safety, but this conflict will handicap emergency management's ability to encourage and implement mitigation. State and federal governments can help communities overcome some of these incentives for unsafe development by providing a greater amount of supplemental mitigation funding. Although in many cases this funding can and should come from local governments, state and federal governments must also commit to resourcing local emergency management. In many cases, there are not enough resources at the local level to go around, and state and federal governments can help bridge the gap.

Funding and increasing staff resources would also likely increase the quality of plans, by allowing emergency management agencies to take full control over the planning process, allowing it to proceed as it was designed to. At least one participant in a county in which the mitigation planning process was run independently acknowledged that they needed substantial staff time to do so, and they were lucky to be in such a position. If states and FEMA are going to continue to require an extensive planning process in order for communities to be eligible for mitigation funding, they must recognize the challenges inherent in such a process, and consider how to provide support that does not compromise its integrity. In the absence of changes to mitigation planning requirements or increases in resource availability, emergency management offices will almost certainly continue to rely on consultants. Although consultants may be able to meet the requirements of the planning process, their presence raises questions about the integrity of the process, especially with respect to plan generation and approval by the community.

Increasing availability of funding for local emergency management generally, and for mitigation specifically, may, however, require governments to change the way they make funding determinations. Criteria for funding that emergency managers interviewed for this research identified as challenging included establishing historical precedent and a favorable cost-benefit analysis. With respect to this first challenge, it is critical for funding programs to acknowledge the changing risk communities face. Where hazard risk has been identified, it is reasonable to fund mitigation, even in the absence of historical losses, especially because development pressures and climate change have affected the risk profile of many communities in ways that have not yet resulted in hazard events. With respect to the second challenge, the need to establish a favorable cost-benefit analysis for implementation of mitigation may result in low-density communities disproportionately failing to receive funding. Although a favorable return

on investment is frequently cited as a reason to engage in mitigation, this emphasis leaves communities where investments in mitigation have a low or negative return unprotected or under-protected from hazard risk. Thus, increasing the amount of available funding overall, and changing state and federal orientations to mitigation are both important for helping local governments implement mitigation.

There are a number of ways federal policy could be changed to accommodate increased mitigation, but one that emerged consistently from these interviews (n=8) was adjustments to the FEMA match for mitigation projects. Currently, for most Hazard Mitigation Assistance funds, FEMA requires local governments to fund 25% of mitigation projects themselves, but will cover the remaining 75% of costs. In some places, this match amount is unaffordable. Moreover, because many of the communities for which a 25% match is unaffordable reported broader community resource limitations, this requirement tends to exacerbate existing inequalities in vulnerability.

With respect to availability of funding, some progress appears to be being made. A few emergency managers (n=3) noted that FEMA policy appears to be shifting with the Disaster Recovery Reform Act (DRRA), which expands the number of mitigation activities eligible for federal funding assistance (Webster & Lindsay, 2019). Although the DRRA does not necessarily address some specific criticisms of federal mitigation programs, and it is unknown what the increase in funding availability will be, these changes are encouraging, and seemed particularly encouraging to some of the emergency managers interviewed for this research. It is hoped that FEMA will, in fact, meaningfully increase the amount of assistance available to counties for mitigation through the DRRA. Another resource communities are using to fund mitigation is the Department of Housing and Urban Development's (HUD) Community Development Block

Grant Mitigation (CDBG-MIT) funding program. In 2019, \$6.875 billion was made available to communities that had experienced disasters in previous years for mitigation specifically designed with the effects of climate change in mind (HUD, 2019). Both the amount of funding and the emphasis on developing 'resilience' to hazards and climate change may represent a shift in thinking and future mitigation resource availability (Moore, 2019).

Implications for Education

The variation in emergency managers' definitions of mitigation echoes the variation in definitions provided by official sources. In 2008 for example, Wayne Blanchard identified at least 31 unique definitions for "mitigate" and "mitigation" (Blanchard, 2008). An examination of FEMA's website reveals at least seven definitions (FEMA 2001, 2005, 2011, 2013, 2018, 2019 & 2020), some of which capture all the important dimensions of mitigation (e.g., "Mitigation is defined as sustained action taken to reduce or eliminate long-term risk to people and property from hazards and their effects" (FEMA, 2011)), and some of which do not (e.g., "Flood hazard mitigation is defined as all actions that can be taken to reduce property damage and the threat to life and public health from flooding" (FEMA, 2013)). The literature, moreover, consistently fails to define mitigation, and occasionally (including by Samuel and Siebeneck, 2019) to accurately define mitigation. All parties involved in promoting, funding, and expecting mitigation must agree on how mitigation out to be defined, whether that definition is the same or similar to the one used here or not. It is unacceptable to continuously redefine mitigation, especially without good cause; to fail to define mitigation in literature, or to otherwise assume that the audience understands mitigation to mean what the author intends; or to include activities that do not meet the definition of mitigation in mitigation funding initiatives or programs. One dimension of this must be including an accurate and consistent definition of mitigation in education and training

for emergency managers. This effort will be undermined, however, if the mitigation programs that emergency managers interact with operationalize mitigation in incongruous or inaccurate ways.

Another dimension of education is providing emergency managers with the documents and resources they need to understand what their roles and responsibilities in mitigation are. Professional emergency management organizations like the National Emergency Management Association (NEMA) and the International Association of Emergency Managers (IAEM), or state and federal emergency management agencies like FEMA, could provide this guidance. If county-level emergency managers are expected to engage in mitigation by professional and governmental organizations, it is appropriate for these same organizations to specify how they are expected to engage.

Conclusion

This chapter has contextualized the findings from this research within the emergency management literature. It compared the theorized emergency management from previous scholarship to the emergency managers who participated in this study, and considered how and why their mitigation roles differed. It concluded with a discussion of the implications these findings have for education, practice and policy.

CHAPTER SIX: CONCLUSION

Although disaster researchers have studied mitigation along a number of important dimensions, little scholarship has specifically examined the county-level emergency management practitioner's role in mitigation. This study sought to address this gap in the literature. Interviews with 42 emergency managers across the United States revealed that, although they spent disproportionately little time on mitigation compared to other phases of emergency management, county-level emergency managers played several distinct roles in mitigation. These roles included a generic emergency management role, a support role, an administrative role, a promoter role, a public educator role, and a planner role. These roles, and the limited extent of emergency manager involvement in mitigation, were explained in part by several factors, including conceptual confusion, resource factors, competing priorities, stakeholder relationships to mitigation, structural factors, meeting requirements, and environmental factors.

This research suggests that, although emergency managers are engaged in many of the tasks, activities and roles that the mitigation literature might predict, there is room for improvement. Emergency managers could expand their role and improve their engagement in mitigation, but they cannot do it alone. Resourcing emergency management offices appropriately, earmarking funds for mitigation, and structuring emergency management agencies properly across levels of government would help eliminate some of the barriers to mitigation described by participants in this study.

Significance of Study for Emergency Management

A primary purpose of this study was to inform the emergency management discipline and profession about the roles that county-level emergency managers play within mitigation.

Although the mitigation roles and activities emergency managers engage in may be inferred from

the literature to some degree, the precise nature of these roles and activities has not heretofore been thoroughly investigated. Not only did this study evaluate the roles played by emergency managers within mitigation, but it also considered the factors that help explain those roles, especially including barriers to mitigation. These factors largely conform to the factors identified by previous literature, providing important confirmatory information to researchers and practitioners alike.

Using this knowledge of both emergency manager roles and the factors that inform these roles, scholars, professionals, and policy makers can build on current researchers and practices. The challenges highlighted by emergency managers strongly suggest that policy makers and government officials ought to provide more resources for mitigation. All parties involved in formulating mitigation policy and programs, disseminating training and education, or engaged in research and scholarship must also be consistent and accurate in their definitions of mitigation. This research also suggests that, in order to more closely mirror the idealized emergency manager induced from the literature, actual emergency managers need more formal guidance from professional resources

Advancing mitigation practice is a critical goal, especially in the context of a rapidly changing climate and hazardscape. Although this study was exploratory, and therefore not generalizable, it provides an important foundation for future research on the ways in which emergency managers can (and should) engage in hazard mitigation. Moreover, in light of the similarities between the findings of this research and Samuel and Siebeneck's (2019) findings, it may be possible to begin to develop constructs, if not models, of emergency manager roles in mitigation. With these goals in mind, future scholars can address related questions and research topics that can inform practice and policy.

Recommendations for Future Research

Although this research helps to establish certain information about the role that emergency managers play within mitigation, it also raises a number of important questions that future research should address. To the extent that this research, in combination with Samuel and Siebeneck's (2019) research, suggests the presence of particular emergency manager roles in mitigation, strengthening and further supporting the presence of these roles will require more research. This research may include investigating whether emergency managers at other levels of government understand their roles as similar to or different from county-level emergency managers. It may also be appropriate to conduct quantitative research to help determine the prominence of each role, as well as their effects on mitigation. Furthermore, although emergency managers reported the amount of their time they spent on mitigation, it would be helpful to have a more specific accounting of precisely what they were doing and for how much time, so that scholars could have a more accurate understanding of how much mitigation is actually being done. Additionally, although this study touched on the location of the emergency management office within county government, a more rigorous analysis of how this location affects role or time spent on mitigation would be beneficial. Similarly, evaluating emergency manager role in the context of the structure of their county's government may explain some of the variation among responses (see for example: Benton, 2002; Benton, 2003).

In addition to understanding what is happening in emergency management within mitigation, a goal of research ought to be understanding how to increase the amount and effectiveness of mitigation that is taking place. With respect to this latter goal, research questions oriented to understanding which factors influence mitigation outcomes, potentially including emergency manager mitigation roles, but also factors like other emergency manager

characteristics, resource availability, socioeconomic characteristics of the community, hazard profile, and others. If indeed the goal of mitigation research is to increase the extent and effectiveness of mitigation, it will also be important to investigate how much the actions of emergency managers ultimately affect mitigation. It may also be appropriate to consider how other frameworks (e.g. disaster risk reduction, resilience, sustainable development) vary in effectiveness.

While exploratory and representing only a first step, this study addressed previously under-addressed questions about the role emergency managers play in mitigation, and also raised important new questions. It is hoped that, through this research, mitigation scholarship, practice and policy will be advanced.

REFERENCES

Abrahams, D. (2014). The barriers to environmental sustainability in post-disaster settings: a case study of transitional shelter implementation in Haiti. *Disasters*, 38(1).

Ahrens, J., & Rudolph, P. M. (2006). The importance of governance in risk reduction and disaster management. *Journal of contingencies and crisis management*, 14(4), 207-220.

Alesch, D. J., & Petak, W. J. (2002). Overcoming obstacles to implementation: addressing political, institutional and behavioral problems in earthquake hazard mitigation policies. *Earthquake Engineering and Engineering Vibration*, *1*(1), 152-158.

Alexander, D. (2013). An evaluation of medium-term recovery processes after the 6 April 2009 earthquake in L'Aquila, Central Italy. *Environmental Hazards*, 12(1), 60-73.

Allan, P., & Bryant, M. (2011). Resilience as a framework for urbanism and recovery. *Journal of Landscape Architecture*, 6(2), 34-45.

Andersson-Sköld, Y., & Nyberg, L. (2016). Effective and Sustainable Flood and Landslide Risk Reduction Measures: An Investigation of Two Assessment Frameworks. *International Journal of Disaster Risk Science*, 7(4), 374-392.

Asgary, A., Hajinejad, A., Rafieian, M., & Badri, A. (2007). Lost and Used Post Disaster Development Opportunities in the Bam Earthquake and the Role of Stakeholders. *Post-disaster reconstruction*, 1000-1012.

Auf der Heide, E. (1989). Disaster Response: Principles of Preparation and Coordination. St. Louis: Mosby.

Baker, E. J. (1977). Public attitudes toward hazard zone controls. *Journal of the American Institute of Planners*, 43(4), 401-408.

Barbier, E. B. (2006). Natural barriers to natural disasters: replanting mangroves after the tsunami. *Frontiers in Ecology and the Environment*, 4(3), 124-131.

Baudoin, M. A., & Wolde-Georgis, T. (2015). Disaster risk reduction efforts in the Greater Horn of Africa. *International Journal of Disaster Risk Science*, 6(1), 49-61.

Becker, S. L., & Reusser, D. E. (2016). Disasters as opportunities for social change: Using the multi-level perspective to consider the barriers to disaster-related transitions. *International Journal of Disaster Risk Reduction*, 18, 75-88.

Benton, J. E. (1996). Fiscal Aid and Mandates: The County Experiences. In D. C. Menzel (ed.) *The American County: Frontiers in Knowledge* (149-165). Tuscaloosa: University of Alabama Press.

- Benton, J. E. (2002). County Government Service Delivery: Does Government Structure Matter?. *Public Administration Review*, *62*(4), 471-479.
- Benton, J. E. (2003). County government structure and county revenue policy: What's the connection?. *State and Local Government Review*, *35*(2), 78-89.
- Benton, J. E. & Menzel, D. C. (1993). County Services: The Emergency of Full-Service Government. In D. R. Berman (ed.) *County Governments in an Era of Change* (53-69). Westport, CT: Greenwood Press.
- Berke, P. R. (1996). Enhancing plan quality: Evaluating the role of state planning mandates for natural hazard mitigation. *Journal of environmental planning and management*, 39(1), 79-96.
- Berke, P. R., Kartez, J., & Wenger, D. (1993). Recovery after disaster: achieving sustainable development, mitigation and equity. *Disasters*, 17(2), 93-109.
- Berke, P. R., Roenigk, D. J., Kaiser, E. J., & Burby, R. (1996). Enhancing plan quality: Evaluating the role of state planning mandates for natural hazard mitigation. *Journal of environmental planning and management*, 39(1), 79-96.
- Berke, P., Smith, G., & Lyles, W. (2012). Planning for resiliency: Evaluation of state hazard mitigation plans under the disaster mitigation act. *Natural Hazards Review*, 13(2), 139-149.
- Berke, P. R., Song, Y., & Stevens, M. (2009). Integrating hazard mitigation into new urban and conventional developments. *Journal of Planning Education and Research*, 28(4), 441-455.
- Berrang-Ford, L., Ford, J. D., & Paterson, J. (2011). Are we adapting to climate change?. *Global environmental change*, 21(1), 25-33.
- Blanchard, B. W. (2008, October). *Guide to emergency management and related terms, definitions, concepts, acronyms, organizations, programs, guidance, executive orders & legislation: A tutorial on emergency management, broadly defined, past and present.* Washington, DC: Federal Emergency Management Agency.
- Birkland, T. A. (1996). Natural disasters as focusing events: Policy communities and political response. *International journal of mass emergencies and disasters*, 14(2), 221-243.
- Birkland, T. A. (1998). Focusing events, mobilization, and agenda setting. *Journal of public policy*, 18(1), 53-74.
- Bolton, P. A., & Orians, C. E. (1998). Local Earthquake Mitigation Programs -- Perceptions of their Effectiveness Following the Loma Prieta Earthquake. *The Loma Prieta, California, earthquake of October 17, 1989, 4, 35.*

- Bourque, L. B., Siegel, J. M., Kano, M., & Wood, M. M. (2007). Morbidity and mortality associated with disasters. In *Handbook of disaster research* (pp. 97-112). Springer, New York, NY.
- Bouwer, L. M. (2011). Have disaster losses increased due to anthropogenic climate change?. *Bulletin of the American Meteorological Society*, 92(1), 39-46.
- Britton, N. R. (1999). Whither the emergency manager?. *International Journal of Mass Emergencies and Disasters*, 17(2), 223-235.
- Brody, S. D. (2003). Are we learning to make better plans? A longitudinal analysis of plan quality associated with natural hazards. *Journal of Planning Education and Research*, 23(2), 191-201.
- Brody, S. D., Bernhardt, S. P., Zahran, S., & Kang, J. E. (2009). Evaluating local flood mitigation strategies in Texas and Florida. *Built Environment*, *35*(4), 492-515.
- Brody, S. D., Kang, J. E., & Bernhardt, S. (2010). Identifying factors influencing flood mitigation at the local level in Texas and Florida: the role of organizational capacity. *Natural hazards*, *52*(1), 167-184.
- Brouwer, R., Akter, S., Brander, L., & Haque, E. (2007). Socioeconomic vulnerability and adaptation to environmental risk: a case study of climate change and flooding in Bangladesh. *Risk Analysis: An International Journal*, 27(2), 313-326.
- Bryant, A. (2002). Regrounding Grounded Theory. *The Journal of Information Technology Theory and Application*, 4(1), 25-42.
- Burby, R. J. (2006). Hurricane Katrina and the paradoxes of government disaster policy: bringing about wise governmental decisions for hazardous areas. *The Annals of the American Academy of Political and Social Science*, 604(1), 171-191.
- Burby, R. J., & Dalton, L. C. (1994). Plans can matter! The role of land use plans and state planning mandates in limiting the development of hazardous areas. *Public administration review*, 229-238.
- Burby, R. J., May, P. J., Berke, P. R., Kaiser, E. J., Dalton, L. C., & French, S. P. (1997). *Making governments plan: State experiments in managing land use*. Baltimore: JHU Press.
- Cannon, T., Twigg, J., & Rowell, J. (2003). Social vulnerability, sustainable livelihoods and disasters. *Londres: DFID*.
- Canton, L.G. (2007). *Emergency Management: Concepts and Strategies for Effective Programs*. Hoboken, NJ: Wiley.

Carlson, J. (2015). Exploring the professionalization of the emergency management field: A qualitative analysis. (Doctoral dissertation). Minneapolis, MN: Capella University.

CEMHS, 2018. Spatial Hazard Events and Losses Database for the United States, Version 16.1. Phoenix, AZ: Center for Emergency Management and Homeland Security, Arizona State University.

Chambliss, D. F., & Schutt, R. K. (2006). *Making sense of the social world: Methods of investigation*. Thousand Oaks, CA: Sage Publications.

Chang, Y., Wilkinson, S., Seville, E., & Potangaroa, R. (2010). Resourcing for a resilient post-disaster reconstruction environment. *International Journal of Disaster Resilience in the Built Environment*, 1(1), 65-83.

Charmaz, K. (2006). Constructing grounded theory: A practical guide through qualitative analysis. Thousand Oaks, CA: Sage Publications.

Cigler, B. A. (1993). The Special Problems of Rural County Governments. In D. R. Berman (ed.) *County Governments in an Era of Change* (89-106). Westport, CT: Greenwood Press.

Cigler, B. A. (1995). County Government in the 1990s. *State and Local Government Review*, 54(1), 3-11.

Col, J. M. (2007). Managing disasters: The role of local government. *Public administration review*, 67, 114-124.

Comerio, M. C. (1992). Impacts of the Los Angeles retrofit ordinance on residential buildings. *Earthquake Spectra*, 8(1), 79-94.

Costanza, R., Pérez-Maqueo, O., Martinez, M. L., Sutton, P., Anderson, S. J., & Mulder, K. (2008). The value of coastal wetlands for hurricane protection. *AMBIO: A Journal of the Human Environment*, 37(4), 241-248.

Cutter, S. L., Ahearn, J. A., Amadei, B., Crawford, P., Eide, E. A., Galloway, G. E., Goodchild, M. F., Kunreuther, H. C., Li-Vollmer, M., Scrimshaw, S. C., Stanley, E. M., Whitney, G., & Zoback, M. L. (2013). Disaster resilience: A national imperative. *Environment: Science and Policy for Sustainable Development*, 55(2), 25-29.

Cutter, S. L., & Emrich, C. (2005). Are natural hazards and disaster losses in the US increasing?. *EOS, Transactions American Geophysical Union*, 86(41), 381-389.

Cwiak, C. L. (2019). Framing higher education and disciplinary efforts through a professionalization lens. *Journal of Emergency Management*, 17(1), 61-66.

Deegan, M. A. (2007). Exploring US flood mitigation policies: a feedback view of system behavior (Doctoral dissertation, Rockefeller College of Public Affairs and Policy, Department of Public Administration and Policy).

Denzin, N. K., & Lincoln, Y. S. (2000). *The Sage handbook of qualitative research*. Sage Publications Ltd.

Department of Homeland Security (June, 2016). *National Mitigation Framework: Second Edition*. Retrieved from https://www.fema.gov/media-library-data/1466014166147-11a14dee807e1ebc67cd9b74c6c64bb3/National Mitigation Framework2nd.pdf

DeSantis, V. S., & Renner, T. (1993). Governing the county: Authority, structure, and elections. In *Contributions in Political Science* (ed. D. R. Berman), 15-28. New York: Greenwood.

Deyle, R. E., Chapin, T. S., & Baker, E. J. (2008). The proof of the planning is in the platting: An evaluation of Florida's hurricane exposure mitigation planning mandate. *Journal of the American Planning Association*, 74(3), 349-370.

Deyle, R. E., & Smith, R. A. (1998). Local government compliance with state planning mandates: The effects of state implementation in Florida. *Journal of the American Planning Association*, 64(4), 457-469.

Diggins, W., Wright, J. D., & Rossi, P. H. (1979). Local elites and City Hall: The case of natural disaster risk-mitigation policy. *Social Science Quarterly*, 60(2), 203-217.

Disaster Mitigation Act of 2000, Pub. L. No. 106-390, 114 Stat. 1552 (2000).

Drabek, T. E., & Hoetmer, G. J. (1991). *Emergency Management: Principles and Practices for Local Government*. Washington, DC: International City Management Association.

Drabek, T. E., Mushkatel, A. H., & Kilijanek, T. S. (1983). *Earthquake mitigation policy: The experience of two states*. Boulder: Institute of Behavioral Science, University of Colorado.

Duxbury, J., & Dickinson, S. (2007). Principles for sustainable governance of the coastal zone: In the context of coastal disasters. *Ecological Economics*, 63(2-3), 319-330.

Dynes, R. R., Quarantelli, E. L., & Kreps, G. A. (1972). *A perspective on disaster planning* (No. DRC-SER-11). Ohio State University Columbus Disaster Research Center.

Emergency Management Assessment Program (2019). *What is EMAP?* Retrieved from https://www.emap.org/index.php/what-is-emap

Evans-Cowley, J. S., & Gough, M. Z. (2007). Is Hazard Mitigation Being Incorporated into Post-Katrina Plans in Mississippi?. *International Journal of Mass Emergencies and Disasters*, 25(3), 177.

Evans-Cowley, J. S., & Gough, M. Z. (2008). Citizen Engagement in Post-Hurricane Katrina Planning in Harrison County, Mississippi. *Cityscape: a Journal of Policy Development and Research*, 10(3), 21.

Fan, L. (2013). Disaster as opportunity? Building back better in Aceh, Myanmar and Haiti. Humanitarian Policy Group.

Farris, D., & McCreight, R. (2014). The professionalization of emergency management in institutions of higher education. *Homeland Security & Emergency Management*, 11, 73-94.

Federal Emergency Management Agency (1986). *Making mitigation work: A handbook for state officials*. Washington DC.

Federal Emergency Management Agency (2001). *FEMA Earthquake Preparedness Workshop*. Retrieved from https://www.fema.gov/news-release/2001/05/14/fema-earthquake-preparedness-workshop

Federal Emergency Management Agency (2005). *Mitigation Can Help Reduce Future Disaster Damage*. Retrieved from https://www.fema.gov/news-release/2005/11/22/mitigation-can-help-reduce-future-disaster-damage

Federal Emergency Management Agency (2007). *Principles of Emergency Management Supplement*. Retrieved from https://www.fema.gov/media-library-data/20130726-1822-25045-7625/principles_of_emergency_management.pdf

Federal Emergency Management Agency (2011). *A Comprehensive Emergency Management Training and Education System*. Retrieved from https://www.fema.gov/media-library-data/20130726-1822-25045-

4021/a_comprehensive_emergency_mangement_training_and_education_system.pdf

Federal Emergency Management (2013). *Disaster Operations and Hazard Mitigation*. Retrieved from https://www.fema.gov/media-library-data/20130726-1535-20490-2458/unit10.pdf

Federal Emergency Management Agency (2015). *Hazard Mitigation Grant Program*. Retrieved from https://www.fema.gov/media-library-data/1437513326617-c124385de1b6061509f775a164c9aabd/FEMA_HMA_HMGP_tri_2015_508.pdf

Federal Emergency Management Agency (2017). IS-393.B: Introduction to Hazard Mitigation. Retrieved from: https://training.fema.gov/emiweb/is/is393a/is393.a-glossary.pdf

Federal Emergency Management Agency (2018). What is Mitigation? Retrieved from https://www.fema.gov/what-mitigation

Federal Emergency Management Agency (2019). *Protecting Yourself Through Mitigation*. Retrieved from https://www.fema.gov/protecting-yourself-through-mitigation

- Federal Emergency Management Agency (2020). *Hazard Mitigation Planning*. Retrieved from https://www.fema.gov/hazard-mitigation-planning
- Fisher, M. A., Brocher, T. M., Hyndman, R. D., Trehu, A. M., Weaver, C. S., Creager, K. C., Croson, R. S., Parsons, T., Cooper, A. K., Mosher, D., Spence, G., B. C. Zelt, Hammer, P. T., ten Brink, U., Pratt, T. L., Miller, K. C., Childs, J. R., Cochrane, G. R., Chopra, S., & Walia, R. (1999). Seismic survey probes urban earthquake hazards in Pacific Northwest. *Eos, Transactions American Geophysical Union*, 80(2), 13-17.
- Ford, J. D., Berrang-Ford, L., & Paterson, J. (2011). A systematic review of observed climate change adaptation in developed nations. *Climatic change*, 106(2), 327-336.
- Fois, F., & Forino, G. (2014). The self-built ecovillage in L'Aquila, Italy: community resilience as a grassroots response to environmental shock. *Disasters*, 38(4), 719-739.
- Garnett, J. D., & Moore, M. (2010). Enhancing disaster recovery: Lessons from exemplary international disaster management practices. *Journal of Homeland Security and Emergency Management*, 7(1).
- George, A. L., & Bennett, A. (2005). *Case studies and theory development in the social sciences* (No. 303.1). Cambridge: MIT Press.
- Glaser, B. (1978). Theoretical sensitivity. *Advances in the methodology of grounded theory*. Mill Valley, CA: Sociology Press.
- Glaser, B. G. (1998). *Doing grounded theory: Issues and discussions*. Mill Valley, CA: Sociology Press.
- Glaser, B. G., & Strauss, A. L. (1967). *The discovery of grounded theory: strategies for qualitative theory*. New Brunswick: Aldine Transaction.
- Godschalk, D. R. (2003). Urban hazard mitigation: creating resilient cities. *Natural hazards review*, 4(3), 136-143.
- Godschalk, D. R., Brody, S., & Burby, R. (2003). Public participation in natural hazard mitigation policy formation: challenges for comprehensive planning. *Journal of environmental planning and management*, 46(5), 733-754.
- Godschalk, D. R., Brower, D. J., & Beatley, T. (1989). *Catastrophic coastal storms: Hazard mitigation and development management*. Duke University Press.
- Goldstein, B. E. (2008). Skunkworks in the embers of the Cedar Fire: enhancing resilience in the aftermath of disaster. *Human Ecology*, 36(1), 15-28.
- Greenberg, M. R., Weiner, M. D., Noland, R., Herb, J., Kaplan, M., & Broccoli, A. J. (2014). Public support for policies to reduce risk after Hurricane Sandy. *Risk analysis*, *34*(6), 997-1012.

- Haddow, G., Bullock, J., & Coppola, D. P. (2017). *Introduction to emergency management*. Butterworth-Heinemann.
- Hakaloba, J. M., Mumba, J., Dambe, R., & Michelo, C. (2016). Examining the integration process of the community based institutions and organisations as a response strategy for 'disaster reduction': A case of the Kazungula District, Zambia. *International journal of disaster risk reduction*, 17, 273-279.
- Hammersley, M. (2012). Troubling theory in case study research. *Higher education research & development*, 31(3), 393-405.
- Harris, L. M., McGee, T. K., & McFarlane, B. L. (2011). Implementation of wildfire risk management by local governments in Alberta, Canada. *Journal of Environmental Planning and Management*, *54*(4), 457-475.
- Healy, A., & Malhotra, N. (2009). Myopic voters and natural disaster policy. *American Political Science Review*, 103(3), 387-406.
- Henstra, D. (2010). Evaluating local government emergency management programs: What framework should public managers adopt?. *Public Administration Review*, 70(2), 236-246.
- Highfield, W. E., & Brody, S. D. (2017). Determining the effects of the FEMA Community Rating System program on flood losses in the United States. *International Journal of Disaster Risk Reduction*, 21, 396-404.
- Hill, M., & Gaillard, J. C. (2013). Integrating disaster risk reduction into post-disaster reconstruction: A long-term perspective of the 1931 earthquake in Napier, New Zealand. *New Zealand Geographer*, 69(2), 108-119.
- Hoblitt, R. P., Miller, C. D., & Scott, W. E. (1987). *Volcanic hazards with regard to siting nuclear-power plants in the Pacific Northwest* (pp. 1-194). Reston, VA: US Geological Survey.
- Hoene, C., Baldassare, M., & Shires, M. (2002). The development of counties as municipal governments: A case study of Los Angeles County in the twenty-first century. *Urban Affairs Review*, *37*(4), 575-591.
- Holdeman, E. & Patton, A. (December, 2008). Project Impact Initiative to Create Disaster-Resistant Communities Demonstrates Worth in Kansas Years Later. *Emergency Management*. Retrieved from https://www.govtech.com/em/disaster/Project-Impact-Initiative-to.html.
- Housing and Urban Development. Allocations, Common Application, Waivers, and Alternative Requirements for Community Development Block Grant Mitigation Grantees. Fed. Reg. 18607, 45838-45871. (Aug. 30, 2019).
- Hunt, A., & Watkiss, P. (2011). Climate change impacts and adaptation in cities: a review of the literature. *Climatic change*, 104(1), 13-49.

- Ingram, J. C., Franco, G., Rumbaitis-del Rio, C., & Khazai, B. (2006). Post-disaster recovery dilemmas: challenges in balancing short-term and long-term needs for vulnerability reduction. *Environmental Science & Policy*, *9*(7), 607-613.
- Intergovernmental Panel on Climate Change. (2012). *Managing the Risks of Extreme Events and Disasters to Advance Climate Change Adaptation*. (Special Report, Field, C. B., Barros, V., Stocker, T. F., Qin, D., Dokken, D. J., Ebi, K. L., Mastrandrea, M. D., Mach, K. J., Plattner, G. K., & Allen, S. K. (Eds.).) Cambridge: Cambridge University Press.
- Intergovernmental Panel on Climate Change. (2018). Global Warming of 1.5 °C, an IPCC special report on the impacts of global warming of 1.5 °C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty. Geneva, Switzerland: IPCC.
- Johnston, J. M., Pagano, M. A., & Russo Jr, P. A. (2000). State limits and state aid: An exploratory analysis of county revenue structure. *State and Local Government Review*, *32*(2), 86-97.
- Kang, J. E., Peacock, W. G., & Husein, R. (2010). An assessment of coastal zone hazard mitigation plans in Texas. *Journal of Disaster Research*, 5(5), 520-28.
- Kates, R. W., & Burton, I. (1986). *Geography, Resources and Environment. Vol. 2: Themes from the Work of Gilbert F. White.*
- Kennedy, J., Ashmore, J., Babister, E., & Kelman, I. (2008). The meaning of 'build back better': evidence from post-tsunami Aceh and Sri Lanka. *Journal of contingencies and crisis management*, 16(1), 24-36.
- King, D., Gurtner, Y., Firdaus, A., Harwood, S. & Cottrell, A. (2016). Land use planning for disaster risk reduction and climate change adaptation: Operationalizing policy and legislation at local levels. *International Journal of Disaster Resilience in the Build Environment, 7*(2), 158-172.
- Kita, S. M. (2017). Urban vulnerability, disaster risk reduction and resettlement in Mzuzu city, Malawi. *International Journal of Disaster Risk Reduction*, 22, 158-166.
- Klima, K., & Jerolleman, A. (2017). A rose by any other name communicating between hazard mitigation, climate adaptation, disaster risk reduction, and sustainability professionals. *Journal of Environmental Studies and Sciences*, 7(1), 25-29.
- Labadie, J. R. (1984). Problems in local emergency management. *Environmental Management*, 8(6), 489-494.

Labossière, L. M., & McGee, T. K. (2017). Innovative wildfire mitigation by municipal governments: Two case studies in Western Canada. *International Journal of Disaster Risk Reduction*, 22, 204-210.

Landry, C. E., & Li, J. (2011). Participation in the community rating system of NFIP: Empirical analysis of North Carolina counties. *Natural Hazards Review*, 13(3), 205-220.

Lane, H. (2000). Sustainable development versus economic growth: A case study on natural disaster in Nicaragua. *The Journal of Environment & Development*, 9(2), 175-182.

Le Masson, V. (2015). Considering vulnerability in disaster risk reduction plans: From policy to practice in Ladakh, India. *Mountain Research and Development*, 35(2), 104-114.

Lin, S., Shaw, D., & Ho, M. C. (2008). Why are flood and landslide victims less willing to take mitigation measures than the public? *Natural Hazards*, 44(2), 305-314.

Lindell, M. K., Prater, C. S., Perry, R. W., & Nicholson, W. C. (2006). Fundamentals of Emergency Management.

Lu, Y., Xu, D., Wang, Q., & Xu, J. (2017). Multi-stakeholder collaboration in community post-disaster reconstruction: case study from the Longmen Shan Fault area in China. *Environmental Hazards*, 1-22.

MacAskill, K., & Guthrie, P. (2016). Disaster risk reduction and empowering local government–a case comparison between Sri Lanka and New Zealand. *International Journal of Disaster Resilience in the Built Environment*, 7(4), 318-329.

Maly, E. (2017, February). Rethinking "Build Back Better" in housing reconstruction: A proposal for "People Centered Housing Recovery". In *IOP Conference Series: Earth and Environmental Science* (Vol. 56, No. 1, p. 012025). IOP Publishing.

Mannakkara, S., Wilkinson, S., & Potangaroa, R. (2014). Build back better: implementation in Victorian bushfire reconstruction. *Disasters*, 38(2), 267-290.

Manyena, S. B. (2013). Disaster event: Window of opportunity to implement global disaster policies?. *Jàmbá: Journal of Disaster Risk Studies*, 5(1), 1-10.

Martin, L. L. (1993). States and Counties: Adversaries or Partners? The Florida Perspective. In *Annual Conference of the American Society for Public Administration*. *Los Angeles* (pp. 7-11).

Massey, J. S., & Smith, G. (1994). No staff, no time, no money: Can budget reform work in a rural county?. *International Journal of Public Administration*, 17(5), 811-829.

Maxwell, J. A. (2005). *Qualitative research design: An interactive approach* (2nd ed.). Thousand Oaks, CA: Sage Publications.

May, P., Burby, R., Ericksen, N., Handmer, J., Dixon, J., Michaels, S., & Smith, D. I. (2013). *Environmental management and governance: Intergovernmental approaches to hazards and sustainability*. Routledge.

McEntire, D. A. (2004). Development, disasters and vulnerability: a discussion of divergent theories and the need for their integration. *Disaster Prevention and Management: An International Journal*, 13(3), 193-198.

McEntire, D. A. (2007). Local emergency management organizations. In *Handbook of disaster research* (pp. 168-182). Springer, New York, NY.

McEntire, D. A., & Dawson, G. (2007). The intergovernmental context. In W. Waugh & K.J. Tierney (Eds.), *Emergency management: Principles and practice for local government* (pp. 57-70). ICMA, Washington, DC.

McEntire, D. A. (2018). *Learning more about the emergency management professional*. FEMA Higher Education Program.

McGee, T. K. (2011). Public engagement in neighborhood level wildfire mitigation and preparedness: Case studies from Canada, the US and Australia. *Journal of environmental management*, 92(10), 2524-2532.

McNamara, K. E., & Des Combes, H. J. (2015). Planning for community relocations due to climate change in Fiji. *International Journal of Disaster Risk Science*, 6(3), 315-319.

McSweeney, K., & Coomes, O. T. (2011). Climate-related disaster opens a window of opportunity for rural poor in northeastern Honduras. *Proceedings of the National Academy of Sciences*, 108(13), 5203-5208.

Menzel, D. C., Marando, V. L., Parks, R. B., Waugh, W. L., Cigler, B. A., Svara, J. H., Reeves, M. M., Benton, J. E., Thomas, R. D., Streib, G., Schneider, M., & Salant, T. J. (1992). Setting a Research Agenda for the Study of the American County. *Public Administration Review*, *52*(2), 173-182.

Meo, M., Ziebro, B., & Patton, A. (2004). Tulsa turnaround: From disaster to sustainability. *Natural Hazards Review, 5*(1), 1-9.

Mercer, J. (2010). Disaster risk reduction or climate change adaptation: are we reinventing the wheel? *Journal of International Development: The Journal of the Development Studies Association*, 22(2), 247-264.

Meyer, C. B. (2001). A case in case study methodology. Field methods, 13(4), 329-352.

Mileti, D. (1999). Disasters by design: A reassessment of natural hazards in the United States. Joseph Henry Press.

Mockrin, M. H., Stewart, S. I., Radeloff, V. C., Hammer, R. B., & Alexandre, P. M. (2015). Adapting to wildfire: rebuilding after home loss. *Society & Natural Resources*, 28(8), 839-856.

Moore, R. (Aug. 30, 2019). \$6.875 Billion available for climate resilience from HUD. *National Resources Defense Council. https://www.nrdc.org/experts/rob-moore/6875-billion-available-climate-resilience-hud*

Morris-Oswald, T., & Sinclair, A. J. (2005). Values and floodplain management: Case studies from the Red River Basin, Canada. *Global Environmental Change Part B: Environmental Hazards*, 6(1), 9-22.

Moser, S. C., & Ekstrom, J. A. (2010). A framework to diagnose barriers to climate change adaptation. *Proceedings of the national academy of sciences*, 201007887.

Munich Re. (2013). *Topics Geo: Natural Catastrophes 2012: Analyses, Assessments, Positions. Munich*. Germany: Munich Re Group. Available at www.munichre.com/publications/302-07742 en.pdf.

National Academy of Public Administration (1980). *Metropolitan Governance: A Handbook for Local Government Study Commissions*. National Academy of Public Administration: Washington, DC.

National Governors' Association. Center for Policy Research, & United States. Defense Civil Preparedness Agency. (1979). *Comprehensive emergency management: A governor's guide*. [Department of Defense], Defense Civil Preparedness Agency.

National League of Cities. (2017). Cities 101 – Types of Local Governments. Retrieved from: https://www.nlc.org/resource/cities-101-types-of-local-governments.

National Research Council. (1999). The impacts of natural disasters: A framework for loss estimation. National Academy Press.

Nelson, M. (2014). Using land swaps to concentrate redevelopment and expand resettlement options in post-Hurricane Katrina New Orleans. *Journal of the American Planning Association*, 80(4), 426-437.

Neuvel, J. M., & van den Brink, A. (2009). Flood risk management in Dutch local spatial planning practices. *Journal of Environmental Planning and Management*, 52(7), 865-880.

Nicholls, R. J., & Cazenave, A. (2010). Sea-level rise and its impact on coastal zones. *Science*, 328(5985), 1517-1520.

Nojang, E. (2015). *The Context and Concept of Individual and Household Preparedness: The Case of Fako Division in Cameroon.* (Doctoral dissertation). Fargo, ND: North Dakota State University.

Norris-Raynbird, C. (2005). A mitigation tale of two Texas cities. *International Journal of Mass Emergencies and Disasters*, 23(2), 37.

O'Brien, G., O'Keefe, P., Rose, J., & Wisner, B. (2006). Climate change and disaster management. *Disasters*, 30(1), 64-80.

Olshansky, R. B., Johnson, L. A., Horne, J., & Nee, B. (2008). Longer view: Planning for the rebuilding of New Orleans. *Journal of the American Planning Association*, 74(3), 273-287.

Olson, R. S., & Olson, R. A. (1993). 'The Rubble's Standing Up'in Oroville, California: The Politics of Building Safety. *International journal of mass emergencies and disasters, 11*(2), 163-188.

Olson, R. S., Olson, R. A., & Gawronski, V. T. (1998). Night and day: Mitigation policymaking in Oakland, California, before and after the Loma Prieta earthquake. *International journal of mass emergencies and disasters*, *16*(2), 145-179.

Oulahen, G., & Doberstein, B. (2012). Citizen Participation in Post-disaster Flood Hazard Mitigation *Planning in Peterborough, Ontario, Canada. Risk, Hazards & Crisis in Public Policy*, 3(1), 1-26.

Ozcevik, O., Turk, S., Tas, E., Yaman, H., & Beygo, C. (2009). Flagship regeneration project as a tool for post-disaster recovery planning: the Zeytinburnu case. *Disasters*, 33(2), 180-202.

Pagano, M. A., & Johnston, J. M. (2000). Life at the bottom of the fiscal food chain: Examining city and county revenue decisions. *Publius: The Journal of Federalism*, 30(1), 159-170.

Park, K. O. (1996). Determinants of County Government Growth. In D. C. Menzel (ed.) *The American County: Frontiers in Knowledge*, (34-50). University of Alabama Press.

Passerini, E. (2001). Who is to blame for the failures of sustainable reconstruction projects?. *Natural hazards review*, 2(2), 45-53.

Paton, D. (2000). Emergency Planning: Integrating community development, community resilience and hazard mitigation. *Journal of the American Society of Professional Emergency Managers*, 7, 109-118.

Paul, B. K., & Che, D. (2011). Opportunities and challenges in rebuilding tornado-impacted Greensburg, Kansas as "stronger, better, and greener". *GeoJournal*, 76(1), 93-108.

Pearce, L. (2003). Disaster management and community planning, and public participation: how to achieve sustainable hazard mitigation. *Natural hazards*, 28(2-3), 211-228.

Perry, R. W. (2003). Incident management systems in disaster management. *Disaster prevention and management: An international journal*, 12(5), 405-412.

Petak, W. J. (1985). Emergency management: A challenge for public administration. *Public Administration Review*, 45, 3-7.

Phillips, B. D. (2005). Disaster as a discipline. *International Journal of Mass Emergencies and Disasters*, 23(1), 85.

Pielke Jr, R., Prins, G., Rayner, S., & Sarewitz, D. (2007). Climate change 2007: Lifting the taboo on adaptation. *Nature*, 445(7128), 597.

Platt, R. H., Salvesen, D., & Baldwin, G. H. (2002). Rebuilding the North Carolina coast after Hurricane Fran: Did public regulations matter?. *Coastal Management*, 30(3), 249-269.

Pomeroy, R. S., Ratner, B. D., Hall, S. J., Pimoljinda, J., & Vivekanandan, V. (2006). Coping with disaster: rehabilitating coastal livelihoods and communities. *Marine Policy*, 30(6), 786-793.

Posner, A. J., & Georgakakos, K. P. (2017). Quantifying the impact of community-scale flood mitigation. *International Journal of Disaster Risk Reduction*, 24, 189-208.

Powell, P. J. (2011). Post-disaster reconstruction: A current analysis of Gujarat's response after the 2001 earthquake. *Environmental hazards*, 10(3-4), 279-292.

Prater, C. S., & Lindell, M. K. (2000). Politics of hazard mitigation. *Natural Hazards Review*, 1(2), 73-82.

Quarantelli, E. L. (2006). Catastrophes are different from disasters: some implications for crisis planning and managing drawn from Katrina. *Understanding Katrina: Perspectives from the social sciences*. Available at: http://understandingkatrina.ssrc.org/Quarantelli

Quarantelli, E. L., & Munasinghe, M. (1990). Disaster response: Generic or agent-specific?. In *Colloquium on the Environment and Natural Disaster Management* (pp. 97-105). World Bank. Environmental Policy and Research Division.

Rahmstorf, S. (2007). A semi-empirical approach to projecting future sea-level rise. *Science*, *315*(5810), 368-370.

Raju, E., & Van Niekerk, D. (2013). Intra-governmental coordination for sustainable disaster recovery: A case-study of the Eden District Municipality, South Africa. *International Journal of Disaster Risk Reduction*, 4, 92-99.

Reddy, S. D. (2000). Factors influencing the incorporation of hazard mitigation during recovery from disaster. *Natural Hazards*, 22(2), 185-201.

Ritchie, J., & Lewis, J. (2003). *Qualitative research practice: A guide for social science students and researchers*. Thousand Oaks, CA: Sage Publications.

Rivera, C., & Wamsler, C. (2014). Integrating climate change adaptation, disaster risk reduction and urban planning: A review of Nicaraguan policies and regulations. *International Journal of Disaster Risk Reduction*, 7, 78-90.

Robert T. Stafford Act, Pub. L. No. 100-707, 102 Stat. 4689 (1988).

Rose, A., Porter, K., Dash, N., Bouabid, J., Huyck, C., Whitehead, J., ... & Tobin, L. T. (2007). Benefit-cost analysis of FEMA hazard mitigation grants. *Natural hazards review*, 8(4), 97-111.

Rubin, C. B. (2012). *Emergency management: The American experience 1900-2010*. Oxfordshire, UK: Routledge.

Rubin, C. B., & Barbee, D. G. (1985). Disaster recovery and hazard mitigation: Bridging the intergovernmental gap. *Public administration review*, 45, 57-63.

Rubin, C. B., Saperstein, M. D., & Barbee, D. G. (1985). *Community recovery from a major natural disaster*. University of Colorado Natural Hazards research and Applications Information Center, Boulder, Colo.

Ruddin, L. P. (2006). You can generalize stupid! Social scientists, Bent Flyvbjerg, and case study methodology. *Qualitative inquiry*, 12(4), 797-812.

Ruggiero, P. (2012). Is the intensifying wave climate of the US Pacific Northwest increasing flooding and erosion risk faster than sea-level rise?. *Journal of Waterway, Port, Coastal, and Ocean Engineering*, 139(2), 88-97.

Salant, T. J. (1989). *Arizona County Government: A Study of Contemporary Issues*. Tuscon: University of Arizona Press, Office of Community and Public Service.

Salant, T. J. (1993). Shifting Roles in County-State Relations. In D. R. Berman (ed.) *County Governments in an Era of Change* (107-121). Westport, CT: Greenwood Press.

Samuel, C. & Siebeneck, L. K. (2019). Roles revealed: An examination of the adopted roles of emergency managers in hazard mitigation planning and strategy implementation. *International Journal of Disaster Reduction*, 39, 1-11.

Sanderson, D. (2000). Cities, disasters and livelihoods. *Risk Management*, 2(4), 49-58.

Scanlon, T. J. (1988). Disaster's little known pioneer: Canada's Samuel Henry Prince. *International Journal of Mass Emergencies and Disasters*, 6(3), 213-232.

Schneider, S. K. (1995). Flirting with disaster: Public management in crisis situations. Armonk, NY: ME Sharpe.

Schneider, R. O. (2002). Hazard mitigation and sustainable community development. *Disaster Prevention and Management: An International Journal*, 11(2), 141-147.

Seidman, I. E. (1991). *Interviewing as qualitative research: A guide for researchers in education and the social sciences*. New York: Teachers College Press.

Sekwat, A. (1996). The Use of Capital Budgeting Decision Models by County Governments: A Survey. *State & Local Government Review*, 180-192.

Silvestri, J. H., & Nelson, M. S. (1999). McQuillin Municipal Corporation, Third rev. ed. *The Law of Municipal Corporations* 1(1.01-3A.26).

Sipe, N., & Vella, K. (2014). Relocating a flood-affected community: good planning or good politics?. *Journal of the American Planning Association*, 80(4), 400-412.

Montano, S. (2017). A Foundation for Factors that Explain Volunteer Engagement in Response and Recovery: The Case of Flooding in East Texas 2016 (Doctoral dissertation, North Dakota State University).

Solecki, W. D., & Michaels, S. (1994). Looking through the postdisaster policy window. *Environmental Management*, 18(4), 587-595.

Somers, S., & Svara, J. H. (2009). Assessing and managing environmental risk: Connecting local government management with emergency management. *Public Administration Review*, 69(2), 181-193.

Steel, B. S., & Lovrich, N. P. (2000). Growth management policy and county government: Correlates of policy adoption across the United States. *State and Local Government Review*, *32*(1), 7-19.

Steele, R., Arno, S. F., & Geier-Hayes, K. (1986). Wildfire patterns change in central Idaho's ponderosa pine-Douglas-fir forest. *Western Journal of Applied Forestry*, *1*(1), 16-18.

Stehr, S. D. (2007). The changing roles and responsibilities of the local emergency manager: An empirical study. *International Journal of Mass Emergencies and Disasters*, 25(1), 37-55.

Stevens, M. (2010). Implementing natural hazard mitigation provisions: Exploring the role that individual land use planners can play. *CPL bibliography*, 24(4), 362-371.

Stevens, M. R., Berke, P. R., & Song, Y. (2008). Protecting people and property: the influence of land-use planners on flood hazard mitigation in New Urbanist developments. *Journal of environmental planning and management*, 51(6), 737-757.

Stevens, M. R., Berke, P. R., & Song, Y. (2010). Public participation in local government review of development proposals in hazardous locations: does it matter, and what do local government planners have to do with it?. *Environmental Management*, 45(2), 320-335.

Strauss, A. L. (1987). Qualitative analysis for social scientists. Cambridge University Press.

Strauss, A., & Corbin, J. (1990). *Basics of qualitative research*. Sage publications.

Strauss, A., & Corbin, J. (1998). Basics of qualitative research: Procedures and techniques for developing grounded theory. Thousand Oaks: Sage.

Taylor, S. J., & Bogdan, R. (1998). In-depth interviewing. *Introduction to qualitative research methods*, *3*, 87-116.

Townshend, I., Awosoga, O., Kulig, J., & Fan, H. (2015). Social cohesion and resilience across communities that have experienced a disaster. *Natural Hazards*, 76(2), 913-938.

Tunstall, S. M., Johnson, C. L., & Penning-Rowsell, E. C. (2004, February). Flood hazard management in England and Wales: from land drainage to flood risk management. In *World Congress on Natural Disaster Mitigation* (pp. 19-21).

US Census Bureau. (2018). QuickFacts. Retrieved from https://www.census.gov/quickfacts

US Department of Health and Human Services. (2013). 2013 NCHS Urban-Rural Classification Scheme for Counties. Retrieved from

https://www.cdc.gov/nchs/data/series/sr_02/sr02_166.pdf

Van Aalst, M. K. (2006). The impacts of climate change on the risk of natural disasters. *Disasters*, 30(1), 5-18.

Van der Waldt, G. (2013). Disaster Risk Management: Disciplinary status and prospects for a unifying theory. *Jamba: journal of disaster risk studies*, 5(2), 1-11.

Wamsley, G. L., & Schroeder, A. D. (1996). Escalating in a quagmire: The changing dynamics of the emergency management policy subsystem. *Public Administration Review*, *56*(3), 235.

Waugh, W. L. (1993). Coordination or control: organizational design and the emergency management function. *International Journal of Disaster Prevention and Management*, 2(4), 17-31.

Waugh Jr, W. L. (1994). Regionalizing emergency management: Counties as state and local government. *Public Administration Review*, 253-258.

Waugh Jr, W. L., & Streib, G. (2006). Collaboration and leadership for effective emergency management. *Public administration review*, 66, 131-140.

- Webster, E. M., & Lindsay, B. R. (July, 2019). *The Disaster Recovery Reform Act of 2019 (DRRA): A Summary of Selected Statutory Provisions*. (CRS Report R45819). Retrieved from https://www.everycrsreport.com/files/20190708_R45819_a0c5be964c60f0090fb3c22048fded6a8e268693.pdf
- Weichselgartner, J., & Pigeon, P. (2015). The role of knowledge in disaster risk reduction. *International Journal of Disaster Risk Science*, 6(2), 107-116.
- White, G. F., & Haas, J. E. (1975). Assessment of research on natural hazards.
- Wiek, A., Ries, R., Thabrew, L., Brundiers, K., & Wickramasinghe, A. (2010). Challenges of sustainable recovery processes in tsunami affected communities. *Disaster Prevention and Management: An International Journal*, 19(4), 423-437.
- Wood, N. J., Burton, C. G., & Cutter, S. L. (2010). Community variations in social vulnerability to Cascadia-related tsunamis in the US Pacific Northwest. *Natural Hazards*, 52(2), 369-389.
- Wu, J. Y., & Lindell, M. K. (2004). Housing reconstruction after two major earthquakes: The 1994 Northridge earthquake in the United States and the 1999 Chi-Chi earthquake in Taiwan. *Disasters*, 28(1), 63-81.
- Yoon, D. K., Youngs, G. A., & Abe, D. (2012). Examining factors contributing to the development of FEMA-approved hazard mitigation plans. *Journal of Homeland Security and Emergency Management*, 9(2).
- Zahran, S., Brody, S. D., Highfield, W. E., & Vedlitz, A. (2010). Non-linear incentives, plan design, and flood mitigation: the case of the Federal Emergency Management Agency's community rating system. *Journal of Environmental Planning and Management*, 53(2), 219-239.
- Zhang, Y., Lindell, M. K., & Prater, C. S. (2009). Vulnerability of community businesses to environmental disasters. *Disasters*, *33*(1), 38-57.

APPENDIX A: EXCERPTS OF FEDERAL POLICY RELATED TO MITIGATION

National Mitigation Framework Suggested Local Government Roles and Responsibilities Roles and Responsibilities of Local Governments that Advance Mitigation

Work with the Federal Government to inform the assessment, development, and coordination of mitigation core capabilities

Provide leadership to promote, integrate, and enable an outcome of state and community empowerment to risk reduction and/or adaptation across all mission areas.

Use regulatory authorities and provide funds, incentives, expertise, and leadership to promote the development, implementation, and assessment of mitigation core capabilities. For example, use financial incentives and targeted capital improvement projects to reduce long-term vulnerabilities.

Contribute to the general understanding of risk through the collection, development, analysis, and sharing of information about threats, hazards, and vulnerabilities, as well as through constant evaluation and enhancement of risk assessment methodologies.

In coordination with other mission areas, develop, fund, and deliver training curricula for grades K-12, trade/technical schools, colleges and universities, continuing education, and the whole community to develop proficiency in understanding risks and mitigation.

Engage with local leaders and planners to share perspectives on localized threats and hazards, vulnerabilities, and priorities for incorporating mitigation into community planning and development, as well as continuity and recovery plans, therefore making achieving resilience a part of the community both before and after a disaster.

Assess risks and disaster resilience. Maintain awareness of threats, hazards, and vulnerabilities.

Incorporate resilience principles and priorities into ongoing activities, including family preparedness plans, economic and community planning and development, construction and assessment of infrastructure, comprehensive plans, disaster response and recovery support, homeland security research and development, training, and exercises. Identify leaders who will be responsible for applying mitigation capabilities to these areas and identify ways to incentivize integration into existing organizational processes.

Acquire funding or resources and take action to reduce risk through projects, such as home elevation, or processes, such as enforcing building codes.

Provide functional capacity and technical expertise to implement long-term vulnerability reduction projects across the whole community, whether engineering a bridge to withstand an earthquake, planning a future development for resilience, or building redundancies into critical infrastructure and lifeline systems.

Identify loss reduction and loss control methods and resources to develop mitigation strategies that reduce risks from threats and hazards to personnel, assets, and operations. Maintain continuity of government and/or continuity of operations/business continuity.

Become familiar with public information and warning systems, share information with friends and neighbors, build skills to enhance behavioral health resilience, plan ahead, and promote mitigation efforts within communities.

Conduct and fund outreach and education to effectively community successful practices, local mitigation priorities, and event-specific warnings and information in ways that are clear, consistent, accessible, and culturally and linguistically appropriate. Plan ahead and incorporate the needs of those with disabilities and others with access and functional needs.

From the Stafford Act:

Sec. 322 Mitigation Planning (42 U.S.C. 5165)

- (a) Requirement of Mitigation Plan As a condition of receipt of an increased Federal share for hazard mitigation measures under subsection (e) of this section, a State, local, or tribal government shall develop and submit for approval to the President a mitigation plan that outlines processes for identifying the natural hazards, risks, and vulnerabilities of the area under the jurisdiction of the government.
- (b) Local and Tribal Plans Each mitigation plan developed by a local or tribal government shall
 - (1) describe actions to mitigation hazards, risks and vulnerabilities identified under the plan; and
 - (2) establish a strategy to implement those actions.

From the Disaster Mitigation Act of 2000:

42 USC 5133 Sec. 101 FINDINGS AND PURPOSE.

- (a) Findings.—Congress finds that—
 - (1) Natural disasters, including earthquakes, tsunamis, tornadoes, hurricanes, flooding, and wildfires, pose great danger to human life and to property throughout the United States;
 - (2) Greater emphasis needs to be placed on—
 - (A) Identifying and assessing the risks to States and local governments (including Indian tribes) from natural disasters:
 - (B) Implementing adequate measures to reduce losses from natural disasters; and
 - (C) Ensuring that the critical services and facilities of communities will continue to function after a natural disaster;
 - (3) Expenditures for postdisaster assistance are increasing without commensurate reductions in the likelihood of future losses from natural disasters;
 - (4) In the expenditure of Federal funds under the Robert T. Stafford Disaster Relief and Emergency Assistance Act (42 U.S.C. 5121 et seq.), high priority should be given to mitigation of hazards at the local level; and
 - (5) With a unified effort of economic incentives, awareness and education, technical assistance, and demonstrated Federal support, States and local governments (including Indian tribes) will be able to—
 - (A) Form effective community-based partnerships for hazard mitigation purposes;
 - (B) Implement effective hazard mitigation measures that reduce the potential damage from natural disasters;
 - (C) Ensure continued functionality of critical services;
 - (D) Leverage additional non-Federal resources in meeting natural disaster resistance goals; and
 - (E) Make commitments to long-term hazard mitigation efforts to be applied to new and existing structures.

APPENDIX B: INTERVIEW INVITATION LETTER

NOTE: THIS INVITATION WILL BE SENT BY EMAIL, IT WILL LOOK AS FOLLOWS:

From: North Dakota State University Center for Disaster Studies and Emergency Management Dept. 2351 P.O. Box 6050 Fargo, ND 58108-6050 (701) 231-5595

Dear [Potential Participant Name],

I am writing to request your input for an exploratory study on the role of emergency managers in disaster mitigation process at the county level. Disaster mitigation can be understood as the phase in which sustained action is taken to reduce or eliminate risk to people and property from hazards and their effects.

I am exploring this issue because not much is known about the county emergency manager's role in mitigation, even while there is reason to believe there are unique challenges and opportunities for counties during mitigation.

I am eager to learn about your mitigation experience in your community. If you would be willing to participate in this project, please contact me to schedule a convenient time for a short phone interview. The interview should take approximately one hour.

Please take a look at the attached document with information about the project. Afterwards, should you have any questions, feel free to contact me by phone at (339) 225-2281 or email at amanda.savitt@ndsu.edu. You may also contact Dr. Sarah Kirkpatrick, who is assisting with this project, by phone at (859) 539-0537 or by email at sarah.kirkpatrick@ndsu.edu.

I thank you in advance for your participation in this research project and look forward to speaking with you about your experiences.

Sincerely,

Amanda Savitt

APPENDIX C: INFORMATION SHEET

NDSU

North Dakota State University
Department of Emergency Management
Center for Disaster Studies and Emergency
Management
Department 2351
P.O. Box 6050
Fargo, ND 58108-6050
(701) 231-5595

"The County Emergency Manager's Role in Disaster Mitigation"

INFORMATION SHEET

Research Study:

You are being invited to participate in an interview for a research project entitled "The County Emergency Manager's Role in Disaster Mitigation." This study is being conducted by Amanda Savitt from North Dakota State University, Department of Emergency Management.

Purpose of Study:

The purpose of this research is to explore the role of the county emergency manager during the disaster mitigation process. Disaster mitigation can be understood as the phase in which sustained action is taken to reduce or eliminate risk to people and property from hazards and their effects.

Basis for Participant Selection:

You are being invited to participate in this research project because of your role as a county emergency manager in FEMA Region III, V or X.

Explanation of Procedures:

Should you choose to participate, we will arrange a time of your choice between November 7, 2018 and February 1, 2019 for an interview. The interview will take approximately one hour unless you have more time and information to share.

The interviews will be conducted over the telephone and will be recorded using a digital recorder to assure that I accurately use the information you provide.

Potential Risks and Discomforts:

There should be no potential discomfort or physical, social, psychological, legal, or economic risk to you due to your participation in this study.

Potential Benefits:

Not much is known about the county emergency manager's role in disaster mitigation, even though there is reason to believe there are unique challenges and opportunities for county emergency managers during mitigation.

Your participation in this project will increase the information available to educate students and faculty in emergency management higher education as well as practicing emergency managers, other local elected officials, and members of local government about the realities associated with disaster mitigation.

Assurance of Confidentiality:

There are several important considerations that will be given to those who participate. First, anything you share in an interview will not be shared with any other interview participants.

Second, the interviews will be digitally recorded. Digitally recorded interviews will be uploaded to the interviewer's personal computer. The sound file will then be transcribed and codes assigned for identifying personal and geographic characteristics. The researchers for this project will be the only people in possession of the interviews, paper listing the codes, and their link to participant information. Once the recordings, transcriptions, and codes are no longer relevant to this research, they will be destroyed.

In interview transcriptions, researcher notes, and the final product, codes rather than identifying characteristics (personal or geographic) will be used. Your personal information will be kept confidential. Your name and jurisdiction will not be used in any reports. Aliases will be substituted instead (e.g. John Smith in County A).

Voluntary Participation and Withdrawal from the Study:

Your participation is voluntary and you may quit at any time. Your decision whether or not to participate will not affect your present or future relationship with North Dakota State University or any other benefits to which you are otherwise entitled. If you decide to participate, you are free to withdraw your consent and to discontinue participation at any time.

Offer to Answer Questions:

You should feel free to ask questions now or at any time. If you have any questions, you can contact me, Amanda Savitt, at amanda.savitt@ndsu.edu or by phone at (339) 225-2281, or my dissertation advisor, Dr. Sarah Kirkpatrick, at sarah.kirkpatrick@ndsu.edu. If you have any questions about the rights of human research participants, or which to report a research-related problem or injury, contact the NDSU Institutional Review Board (IRB) Office at (701) 231-8908 or ndsu.irb@ndsu.edu.

APPENDIX D: INTERVIEW GUIDE

County Emergency Managers in Disaster Mitigation

Interview Guide

Introduction Script: Before we begin, I want to make sure that you are comfortable with a few things. Are you comfortable with the fact that you have been selected for participation in this research due to your role as a county emergency manager; that your participation in this project is voluntary; that you can let me know if you want to stop participating anytime; that while your confidentiality is not guaranteed, your name and your county's name will not be used in the final write-up of the data collected for this research; and that our conversation is going to be digitally recorded? Do you have any questions before we begin?

- 1. Please tell me about your background and experience as a county emergency manager. Information sought:
 - -Explanatory factors for understanding of role in mitigation
 - -Disaster experience
- 2. Please tell me a little bit about your county.

Information sought:

- -Contextual information, explanatory factors
- -Structure of government and distribution of authority
- -Hazards, vulnerabilities, and risks
- 3. How would you define 'mitigation'?

Information sought:

- -Consistency/accuracy of EM conceptual understanding of mitigation
- 4. What has your county done to mitigate hazard risk? Information sought:
 - -Mitigation activities that have been undertaken by the county
 - -Consistency/accuracy of EM conceptual understanding of mitigation
- 5. What role did or do you play in mitigation efforts in your community? Information sought:
 - -Perceived role of EM in mitigation

County Represented			Interview Date:		
State Represented:			Time Started	•	Time Finished:
Gender:	Education:	Experier	nce:	Backg	round:

APPENDIX E: LIST OF POTENTIAL PROBING QUESTIONS

- 1. What did you do before you became [COUNTY NAME'S] emergency manager?
- 2. What do you view as your primary responsibility during disaster mitigation?
- 3. What is your vision for mitigation in your community?
- 4. What sources of funding have you used to finance mitigation?
- 5. What projects would you like to do if you had additional resources?
- 6. Which people or organizations have you partnered with on mitigation projects?
- 7. Do you feel you have regional or state support in your mitigation initiatives?
- 8. What has been your experience with the mitigation planning process?
- 9. Compared to other things you do day to day, how much time do you spend doing mitigation?
- 10. Are there any other groups in your community working on mitigation-related tasks or projects?
- 11. [If they have experienced a disaster] Have you tried to implement mitigation as part of the recovery process?
- 12. Do you feel the members of your community value disaster mitigation?
- 13. What sources of information have you used during mitigation? How did you use those information sources?

APPENDIX F: PARTICIPANT COUNTIES AND SELECT DEMOGRAPHIC INFORMATION

I	Government Type	Population	Density (pop/mi²)	Jurisdiction Type	2019 County Budget	Poverty Level	EM Office Location
1 0	Council-administrator	37,000	169	Large fringe metro	\$115 million	9	Independent
7	Council-elected exec.	201,000	96	Small metro	\$222 million	51	Sheriff's Office
n	Commission	12,000	18	Noncore	\$20 million	16	Independent
4	Commission	105,000	167	Large fringe metro	\$99 million	18	Independent
S	Council-elected exec.	102,000	141	Small metro	\$163 million	6	Independent
9	Commission	8,000	26	Micropolitan	\$18 million	14	Independent
7	Commission	40,000	26	Noncore	\$21 million	20	Independent
∞	Council-elected exec.	31,000	12	Micropolitan	\$335 million	7	Independent
6	Commission	238,000	620	Large fringe metro	\$285 million	5	Sheriff's Office
10	Commission	23,000	92	Noncore	\$7 million	17	Independent
11	Council-administrator	316,000	747	Large fringe metro	\$712 million	7	Independent
12	Council-administrator	435,000	929	Large fringe metro	\$509 million	10	Emergency Svcs
13	Council-elected exec.	2,000	0	Noncore	\$13 million	16	Independent
14	Council-administrator	321,000	360	Medium metro	\$306 million	15	Independent
15	Commission	11,000	43	Noncore	\$50 million	12	Emergency Svcs
16	Commission	28,000	75	Medium metro	\$19 million	11	Independent
17	Commission	18,000	88	Large fringe metro	\$43 million	5	Fire-Rescue
18	Commission	352,000	77	Medium metro	\$357 million	19	Independent
19	Council-administrator	500,000	999	Large fringe metro	\$476 million	7	Emergency Svcs
20	Council-administrator	65,000	101	Large fringe metro	\$331 million	9	Fire-Rescue
21	Commission	20,000	22	Noncore	\$41 million	11	Independent
22	Council-administrator	000,66	155	Small metro	\$114 million	15	Planning
23	Commission	000,6	5	Noncore	\$21 million	8	Independent
24	Commission	25,000	23	Noncore	\$27 million	16	Sheriff's Office
25	Commission	40,000	<i>L</i> 9	Noncore	\$18 million	18	Public Safety
26	Commission	83,000	75	Small metro	\$66 million	18	Sheriff's Office
27	Commission	67,000	470	Large fringe metro	\$206 million	8	Fire Department
28	Commission	28,000	89	Noncore	\$32 million	6	Independent
29	Commission	172,000	409	Large fringe metro	\$43 million	9	Independent
30	Council-elected exec.	1,930,000	913	Large central metro	\$5.8 billion	10	Independent

1	Government Type	Population	Density (pop/mi²)	Jurisdiction Type	2019 County Poverty Budget Level	Poverty Level	EM Office Location
31	Council-elected exec.	000,86	13	Small metro	\$173 million	8	Independent
32	Commission	37,000	35	Micropolitan	\$21 million	23	Independent
33	Council-elected exec.	948,000	3926	Large central metro	\$1.2 billion	21	Independent
34	Commission	11,000	98	Large fringe metro	\$10 million	15	Independent
35	Commission	26,000	51	Noncore	\$10 million	20	Independent
36	Commission	35,000	29	Noncore	\$68 million	13	Independent
37	Council-elected exec.	713,000	342	Large fringe metro	\$995 million	6	Public Safety
38	Commission	4,000		Noncore	\$12 million	19	Independent
39	Commission	200,000	32	Medium metro	\$341 million	15	Sheriff's Office
40	Council-elected exec.	166,000	612	Large fringe metro	\$262 million	15	Sheriff's Office
41	Council-administrator	197,000	211	Medium metro	\$178 million	12	Emergency Svcs
42	Commission	000,69	166	Micropolitan	\$27 million	15	Emergency Svcs

APPENDIX G: INSTITUTIONAL REVIEW BOARD APPROVAL

NDSU NORTH DAKOTA STATE UNIVERSITY

December 6, 2018

Dr. Sarah Kirkpatrick Emergency Management

Re: IRB Determination of Exempt Human Subjects Research:

Protocol #HS19112, "The Role of the County Emergency Manager in Disaster Mitigation"

Co-investigator(s) and research team: Amanda Savitt

Date of Exempt Determination: 12/6/2018 Expiration Date: 12/5/2021

Study site(s): varied

Kristy Shinley

Sponsor: n/a

The above referenced human subjects research project has been determined exempt (category #2b) in accordance with federal regulations (Code of Federal Regulations, Title 45, Part 46, Protection of Human Subjects). This determination is based on the original protocol submission (received 11/30/2018) with updated information sheet (received 12/5/2018).

Please also note the following:

- If you wish to continue the research after the expiration, submit a request for recertification several weeks prior to the expiration.
- The study must be conducted as described in the approved protocol. Changes to this protocol must be approved prior to initiating, unless the changes are necessary to eliminate an immediate hazard to subjects.
- Notify the IRB promptly of any adverse events, complaints, or unanticipated problems involving risks to subjects or others related to this project.
- Report any significant new findings that may affect the risks and benefits to the participants and the IRB.

Research records may be subject to a random or directed audit at any time to verify compliance with IRB standard operating procedures.

Thank you for your cooperation with NDSU IRB procedures. Best wishes for a successful study. Sincerely,

Kristy Shirley, CIP, Research Compliance Administrator

For more information regarding IRB Office submissions and guidelines, please consult http://www.ndsu.edu/research/integrity_compliance/irb/. This Institution has an approved FederalWide Assurance with the Department of Health and Human Services: FWA00002439.

INSTITUTIONAL REVIEW BOARD

NDSU Dept 4000 | PO Box 6050 | Fargo ND 58108-6050 | 701.231.8995 | Fax 701.231.8098 | ndsu.edu/irb

Shipping address: Research 1, 1735 NDSU Research Park Drive, Fargo ND 58102

NDSU is an EO/AA university.

APPENDIX H: PARTICIPANTS' DEFINITIONS OF MITIGATION

I could quote a textbook but I don't want to do that. I'd say that, to help adapt to the environment. (Participant 1)

Mitigation is just lessening the impact of a disaster. (Participant 3)

I would say mitigation, the programs, it tries to mitigate the problem in the future. (Participant 4)

We always look at what we could do to make things better and not have the impact on a disaster that would come. (Participant 5)

Well, for me, mitigation is basically, kind of lessen the severity of any incident we have. (Participant 6)

An action taken to lessen the impact of an emergency (Participant 9)

Mitigation is reducing your vulnerabilities, to either reduce the impacts, or to completely remove it, so it doesn't happen again the next time. (Participant 11)

I define mitigation as, it's anything we can do to prepare for the risks we face. (Participant 13)

Mitigation, to me, it starts from prevention to providing a service during a disaster and continuing to provide that service after that disaster, and to plan, to have a plan, implement the plans, and once you get through with the disaster, reevaluate, after action report, see what you can do better, see what you need to improve on, what other resources we may need, identify those and make improvements in case we have another disaster. (Participant 15)

Well, it's working along other agencies and being able to get out there and try to inform, teach, and kind of like schooling people on, we need to do this in case this happens, or we need to do that in case that happens. Anything we can do for the county to help reduce the loss that might come out of any kind of a natural disaster, such as a tornado or straight line winds or anything like that. (Participant 16)

Mitigation essentially is actions necessary to defend against known hazards. And that could be things we do, things we build, things we change. (Participant 17)

I would define it as preparing for natural and manmade disasters, but it's more than preparing, it's almost preventing, putting in measures to prevent the loss or lessen the impact of an event. (Participant 18)

We're really looking at how we can find out the risk of – an ideal goal would be to reduce the impacts of the threat so we don't have to deal with them anymore, but we realize that things

like weather, that's not always a reality. So a lot of it is just being strategic for us, more so with funding, so we can reduce the impacts of the hazards we face here. (Participant 20)

Anytime we have a disaster, what can we do to limit the impacts of it. Now that we've focused on it, kind of one of those fields, that it still kind of goes by the wayside, no one really thinks of well, if we do this obviously it will save us money in the long run. (Participant 21)

It's actions you can take to reduce the amount of risk to your community. (Participant 22)

I guess mitigation, how I would define it, is to lessen impacts of future disasters, or doing projects to lessen impacts. (Participant 23)

Well, I look at it as a way to prevent damages and loss of life or injuries. So doing things like setting up programs, or ensuring that people aren't building and/or moving vulnerable-type facilities into hazardous areas. (Participant 25)

Mitigation to me, for a definition, is being able to reduce or eliminate any type of risk or hazard from occurring, or if it does minimizing that issue. But then also, there's two phases to mitigation, you have pre- and post-mitigation when a disaster occurs. So you can have the mitigation that occurs before the disaster, which the definition I just shared pertains to that, then you also have post-mitigation, after a disaster occurs, what do you do to eliminate that from occurring again? (Participant 26)

To me, mitigation is lessening the effects of a disaster. So, if you can take steps beforehand in order to prevent or lessen water or earthquake or damages caused by a hazard, that would be mitigation. (Participant 27)

So, fundamentally, mitigation is risk reduction. It's anything that strengthens things to reduce risk to that hazard. (Participant 30)

Mitigation is what emergency management should be primarily focused on. Just, reducing the risk to people. Property or structures by, it could be either adjustments to infrastructure, physically moving people, education. It's just anything to make the potential impact of a hazard less. (Participant 31)

I look at it as anything that reduces the severity towards life, property, and I add in the financial component, I can't get a business up and running, we're not going to make money in the county, so that's kind of how I look at it, literally in that order. But it's just trying to take the risk and severity and lessen it, is really how I look at it. (Participant 32)

We really consider mitigation steps that we can take to lessen the impact of future disasters. (Participant 33)

Mitigation, we know bad things are going to happen, we try to lessen the impact of those bad things. That mitigation, to me, in a nutshell. (Participant 34)

I look at mitigation as the efforts and steps we can take to minimize the impact of disaster events. (Participant 36)

Certainly we view it in the traditional sense of FEMA's definition of hazard mitigation, you know, reducing risk to people and property ahead of disasters. There's kind of the overarching FEMA definition hat is kind of the underpinning of most of what we do, but there are nuances that we look at in terms of how we're adapting to things like a changing climate and how we're actually addressing things that are manmade, that aren't naturally, or part of the natural ecosystem. (Participant 37)

Mitigation would be the planning and process of preventing disasters, planning for the disasters, and then going ahead and preparing to plan for the disasters and doing the mitigation work prior to, so it's a bridge. (Participant 38)

I would think of it as things we can do before the bad thing happens so we make it not so bad. (Participant 39)

Well mitigation, in my mind, is something that can be done to help prevent or alleviate any further hazards from happening, and that could be all hazards, for that matter. It can be something simple as a physical, something physically done to prevent something from happening, to having the right systems in place to better notifications or to prevent something from happening. (Participant 41)

You know, it's the preplanning. So, having plans in place for worst case scenario is something that would come to the top of my brain, whenever you ask me about mitigation. You want to always protect life and property, so to always think of what worst case scenario could happen and what plans or what training do we have to reduce that? It's something we work on constantly. (Participant 42)